



THE CAUSES AND COSTS OF MODIFICATIONS TO MILITAR, CONSTRUCTION CONTRACTS

A thesis procented to the Faculty of the U.S. Army Command and General Staff College in partial fulfiliment of the requirement for the degree

MASTER OF MILITARY ART AND SCIENCE

bу

ERIC T. MOGREN, MAJ, USA B.S., State University of New York College of Environmental Science and Forestry, 1974 M.S.E., University pf Texas at Austin, 1984

> Fort Leavenworth, kansas 1980



APPROVED YOU I MINI DE SASE

DETUBUTION LALER. A.

Approved for public release; distribution is unlimited.

86-3471

FILE COPY

310

86 10 8 287

DISCLAIMER NOTICE

THIS DOCUMENT IS BEST QUALITY PRACTICABLE. THE COPY FURNISHED TO DTIC CONTAINED A SIGNIFICANT NUMBER OF PAGES WHICH DO NOT REPRODUCE LEGIBLY. Unclassification of this PAGE

ADHIDA 833

			REPORT DOCUM	MENTATION	PAGE		_
1a. REPORT SE Unclass	CURITY CLASSI	FICATION		1b. RESTRICTIVE	MARKINGS		
2a. SECURITY CLASSIFICATION AUTHORITY			3. DISTRIBUTION	AVAILABILITY O	FREPORT		
				Approved f	or public r	elease;	distribution
20. DECLASSIFI		NGRADING SCHEDU	Lt	is unlimit	ed. Leontr	hucu on	reveres ;
4. PERFORMIN	G ORGANIZATI	ON REPORT NUMBE	R(S)	5. MONITORING	ORGANIZATION R	EPORT NUME	3ER(S)
6a. NAME OF	PERFORMING C		6b. OFFICE SYMBOL (If applicable)	7a. NAME OF M	ONITORING ORGA	NIZATION	
Staff Col	llege	and General	ATZL-ŚWD-GD				
6c. ADDRESS (City, State, and	I ZIP Code)		7b. ADDRESS (CH	ty, State, and ZIP	Code)	
ATTN: ATZ	ZL-SWD-GD						
Fort Leav	venworth,	Kansas 66027	/-6900				
8a. NAME OF ORGANIZA	FUNDING/SPOI	NSORING	8b. OFFICE SYMBOL (If applicable)	9. PROCUREMEN	T INSTRUMENT ID	ENTIFICATION	NUMBER
8c. ADDRESS (City, State, and	ZIP Code)	· · · · · · · · · · · · · · · · · · ·	10. SOURCE OF	FUNDING NUMBER	RS	
		••••		PROGRAM ELEMENT NO.	PROJECT NO.	TASK NO.	WORK UNIT ACCESSION NO.
11. TITLE (Incl	ude Security Cl	assification)		L	1	<u>i </u>	
The Caus	res and Co	osts of Vocif	lications to "i	litery Cons	truction Co	ntrata	(11)
12. PERSONAL	. AUTHOR(S)	Brie W. Morre	n. Kaior H.S.	Army Corns	of Frainco	~ 0	
123 TYPE OF	REPORT		OVERED	A DATE OF PER	BT Mar Month	(B) (Dav) 15 P	
Unster's	s Thesis	FROM <u>8-1</u>	<u>.985</u> TO <u>6-1986</u>	7936		Juy / 13. (1	199
16. SUPPLEME	NTARY NOTAT	ION					
17	COSATI	CODES	18 SUBJECT TERMS (Continue on reven	e if necessary an	d identify by	block number)
FIELD	GROUP	SUB-GROUP	Sodifications	s, Cost Gro	wth, Constr	uctibili	ty Review.
			Technical Dec	cign Seview	, Chance Cr	ders, Us	ing Dervice
			Review				
19 ABSTRACT	(Continue on i	reverse if necessary	and identify by block n	iumber)			
This stud	ly identif	ies the cous	to stace bra as	C changes to	o military	construc	tion contracts
and discu	isses nov. Lihility.	excessive pr	'OJECT COST Grow	ath Can be : an	reduced	thorough	using service
cont v. set	or streey g	and technice	II GESIMA PEVIEN	NE .			
25 milita	ry constr	ruction proje	ects, administer	red by the (Corps of En	cineers.	were
identifie	ed for stu	dy. Project	; files were rev	viewed to de	etermine th	e reason	s for and
costs of	778 items	of change of	ontained in 268	8 modificat	ions. Proj	ect cost	growth was
correlate	ed by revi	eus conducte	d, project size	e, project ·	type, and d	esign af	anov. The
study fou	ind the ma	ijor causes o	i mode to re de	esian defic:	iencies, us	er reque	sted changes,
ano unkno architect	own site C tural corr	conditions.	. ost design dei m. followed by	the mechanic	ere found	to occur	in al docima
disciplin	ner. Con	itinued on bo	ack)	the mechan.	icar, then	electric	ar debien
20. DISTRIBU	TION / AVAILAB	ILITY OF ABSTRACT		21. ABSTRACT S	ECURITY CLASSIFIC	CATION	
	SSIFIED/UNLIMIT	ED SAME AS	RPT. DTIC USERS	Unclasci	fied		
ZZa. NAME O	JF RESPONSIBLE			ZZD. TELEPHONÉ	(include Area Cod	e) ZZC. OFFI	CE SYMBOL
	473. 94 MAD		PR edition may be used in	I			
	· → / J į 04 MIAK	03 A	All other editions are o	bsolete.	SECURITY	CLASSIFICAT	ION OF THIS PAGE

Unclassified

Unclassified SECURITY CLASSIFICATION OF THIS PAGE

Other requests for this focument must be referred U.S. Arm collers, ATTU: ATZL-SEP-GD, Fort Leavenworth, Cansar (1927-

19. Abstract (Continued)

The study further found that projects receiving technical, constructibility, and using service reviews had simificantly lower overall cost growth than projects not receiving all three reviews. However, the lack of any one review could lead to cost growth increases regardless of the detail in which the remaining reviews were conducted. Less expensive OLL and AFH projects tended to have higher cost growth rates than more costly UCA projects, apparently due to the higher priority given to design and review of large projects. In-house designs were found to have lower cost growth than designs by contracted private L/E firms largely due to creater problems with unknown site conditions experienced by A/E designed projects.

The study also examined the design review system. Information on reviewer training and experience, time available, and review procedures was collected by substitute aire. It was found that the primary reason for poor suality reviews was lack of time brought about by competing work requirements.

The findings of this study were compared with the findings of 16 related works in order to evaluate the external validity of the results. A strong correlation between most of the findings was found to exist, particularly in the area of modification causes and the effect of good reviews on cost growth reduction.

The study concludes that the pajor causes of modifications to military construction contracts are desire deficiencies, user requested changes, and unknown site conditions. Thorough design reviews can decrease the rate and cost of modifications in these areas. Teviews must be properly menaged, however, to insure they receive the priority needed to be effective.

Unclassified

SECURITY CLASSIFICATION OF THIS PAGE

THE CAUSES AND COSTS OF MODIFICATIONS TO MILITARY CONSTRUCTION CONTRACTS

A thesis presented to the Faculty of the U.S. Army Command and General Staff College in partial fulfillment of the requirement for the degree

MASTER OF MILITARY ART AND SCIENCE

bу

ERIC T. MOGREN, MAJ, USA B.S., State University of New York College of Environmental Science and Forestry, 1974 M.S.E., University of Texas at Austin, 1984

> Fort Leavenworth, Kansas 1986

Approved for public release; distribution is unlimited.

86-3471

ł

MASTER OF MILITARY ART AND SCIENCE

THESIS APPROVAL PAGE

Name of candidate: Eric_I. Mogren. Major. Corps_of_Engineers

Title of thesis: <u>The Causes and Costs of Modifications to</u> <u>Military Construction Contracts</u>

Approved by: , Thesis Committee Chairman ц., M.S., M.S.E. Much JER Major Clayton H. Dethleften, M.B.A. (Alt Force) Committee Member ..., Member, Consulting Faculty LTC Milfred L. Dellva, P.H.D.

Accepted this 9th day of May 1980 by:

F

Aulys J. Brooker Parector, Graduate Degree

The opinions and conclusions expressed therein are those of the student author and do not necessarily represent the views of the U.S. Army Command and General College or any other governmental agency. (References to this study should include the folgoing stalement).

ABSTRACT

THE CAUSES AND COSTS OF MODIFICATIONS TO MILITARY CONSTRUCTION CONTRACTS, by Major Eric T. Mogren, USA, 199 pages.

This study identifies the causes and costs of changes to military construction contracts and discusses how excessive project cost growth can be reduced by thorough constructibility, using service and technical design reviews.

25 military construction projects, administered by the Corps of Engineers, were identified for study. Project files were reviewed to determine the reasons for and costs of 778 items of change contained in 268 modifications. Project cost growth was correlated by reviews conducted, project size, project type, and design agency. The study found the major causes of mods to be design deficiencies, user requested changes, and unknown site conditions. Most design deficiencies were found to occur in architectural aspects of design, followed by the mechanical, then electrical design disciplines.

The further study found that projects receiving technical, constructibility, and using service reviews had significantly lower overall cost growth than projects not receiving all three reviews. However, the lack of any one review could lead to cost growth increases regardless of the detail in which the remaining reviews were conducted. Less expensive OMA and AFH projects tended to have higher cost growth rates than more costly MCA projects, apparently due to the higher priority given to design and review of large projects. In-house designs were found to have lower cost growth than designs by contracted private A/E firms largely due to greater problems with unknown site conditions experienced by A/E designed projects.

The study also examined the District's design review system. Information on reviewer training and experience, time available, and review procedures were collected by questionnaire. It was found that the primary reason for poor quality reviews was lack of time, brought about by competing work requirements.

The findings of this study were compared with the findings of 16 related works in order to evaluate the external validity of the results. A strong correlation between most of the findings was found to exist, particularly in the areas of modification causes and the effect of good reviews on cost growth reduction.

The study concludes that the major causes of modifications to military construction projects are design deficiencies, user requested changes, and unknown site conditions. Thorough design reviews can decrease the rate and cost of modifications in these areas. Reviews must be properly managed, however, to insure they receive the priority needed to be effective.

111

Distribution Statement A is correct for this report. Per Ms. Bess Knapke, AC&GSC/ATZL-SWS-L

1. .

Dist

ACKNOWLEDGEMENTS

This study was, in many ways, a team effort. I would like to take this opportunity to extend my appreciation to those people who contributed their time and energy to enable this work to be completed.

First and foremost is Miss Christine Hendzlik. Christine has been involved with the "mod study" since its inception. For over a year, she studied project files, visited project sites, interviewed personnel, and coordinated preliminary results between the District and installation. Her thoughtful suggestions and common sense contributed immeasurably to the study's findings and conclusions. This study would have been much less complete without her perception, dedication, and professionalism.

Special thanks is also due Major Clay Dethlefsen, Corps of Engineers, who originally suggested this study be done. In addition to his help in sorting through the mountains of data Christine and I collected, Clay's encouragement went a long way in Keeping up my own motivation. His experience in the Corps and knowledge of the District were essential to keeping the study oriented in the proper direction.

I would also like to acknowledge the excellent cooperation and support provided by the resident office: Elgin Saathof, Captain Jim McCarthy, Dale Wagner, John Morrison, and Ray Meis. Each was always willing to drop what they were doing to help find some misplaced file or answer questions about specific projects.

I would like to extend my sincerest appreciation to Colonel

Robert Amrine, Paul Barber, Glen Davis, Bill Morris, Merle Braden, Gus Spallo, Sally Christie, and the other District employees who contributed so much in the way of comments, suggestions and insight.

Additionally, I would like to recognize the support provided by installation personnel. Dennis Grady, of the District project office, provided invaluable administrative support. Lieutenant Colonel Robert Lowry and his staff at the Directorate of Engineering and Housing were always available and willing to answer questions, provide information, and give a customer's perspective on the data collected.

Special thanks is due my thesis committee - Major Stan Murphy, Doctor Willfred Dellva, and Clay Dethlefsen - for the hours they spent proofreading the various drafts this paper went through before reaching its final form.

Finally, I want to thank my wife, Robby, and daughter, Chrissy for their love and support. Their tolerance of missed meals, late nights, and worked weekends went far beyond what I had any right to ask for or expect. In that sense they contributed more than anyone else to the success of this project.

> Eric T. Mogren Major, Corps of Engineers

Port Leavenworth, Kansas 9 Mai 1986

TABLE OF CONTENTS

		Page
List of Tat	bles	5
List of Fig	jureg	X 1 1
Chapter 1.	Introduction	1
	Fackground	1
	Research Objectives and Problem Statement	5
	Aseumptions	7
	Constraints and Delimitations	5
	Significance of Study	11
	Endnotes	13
(tanter 1.	Survey of Existing Publications	14
	Gereral	14
	Government Puplications	15
	Government Studies	18
	Studies from Academic Institutions	22
	Articles from Professional Journals	
	and Magazines	25
	Summary	29
	Endnotes	30
Chapter 3.	Methodology	32
	General	
	Methodology for Determining the Causes and Costs of Modifications	34

Table of Contents (Continued)

Chapter 3.	(Continued)	Page
	Methodology for Evaluating the Design Review System	49
	Methodology for Comparing Findings with Other Studies	53
	Summary	54
Chapter 4.	Causes and Costs of Contract Modifications	55
	General	55
	Modification Causes	56
	Causes of Design Deficiencies	58
	Modification Costs	59
	Lost Correlation by Design Review	65
	Cost Correlation by Project Type	59
	Cost Correlation by Project Size	72
	Lost Correlation by Designer	27
	Summery	8 0
Chapter 5.	The Preconstruction Design Review System	81
	General	81
	Personnel Qualifications	83
	Time Available for Reviews	90
	Review Processing Procedures	96
	5ummarγ	110
Chapter 6.	Comparison of Findings with Other Studies	112
	General	112

Table of Contents (Continued)

Chapter 6.	(Continued)	Fage
	Corroborative Findings - Causes and Costs of Mods	. 112
	Corroborative Findings - Design Review System	. 117
	Contradictory Findings	. 126
	Findings Not Addressed in Other Studies	. 129
	Summary	. 130
	Endnotes	. 131
Chapter 7.	Conclusions and Recommendations	. 134
	General	. 134
	Causes of Modifications and Cost Growth	134
	Cost of Modifications	135
	Impact of Reviews on Modifications and Cost Growth	. 137
	Impact of Project Type on Modifications and Cost Growth	139
	Impact of Project Size on Modifications and Cost Growth	140
	Impact of Design Agency on Modifications and Cost Growth	140
	Qualifications of Review Personnel	141
	Review Time Available	143
	Review Procedures	144
	Recommendations for Further Study	145
	Summary	146
	Endnotes	148

Table of Contents (Continued)

Page

Appendic	es:		
	1.	Operational Definitions	
	2.	Project Summary Sheets 153	
	3.	Causes of Design Deficiencies	
	4.	Modification Cost Correlation Tables	
	5.	Indirect Costs for Engineering, Design, and Overhead	
	6.	Review System Questionnaire	
Bibliogr	aphy	177	
Initial	Dist	ribution List	

іx

List of Tables

Table	4	page
1.	The Trend in Military Construction Spending	1
2.	Cost Growth on Federally Funded Projects	2
3.	Time Spent on Modifications (Example)	48
4.	Review System Components of Evaluation	51
5.	Summary of Survey Population Sample	51
6.	Representation from Engineering Division	52
7.	Questionnaire Respondent Profile	53
8.	Comparison of Modification Sources - Cost versus Items of Change	58
9.	Disciplines in which Design Deficiencies Occur	59
10.	Potential Savings from Control of Cost Growth	62
11.	Causes of Modifications Correlated by Review	68
12.	Causes of Modifications Correlated by Project Type	72
13.	Correlation Between Project Type and Adjusted Contract Amount	75
14.	Causes of Modifications Correlated by Project Size	76
15.	Dollar Amount of Cost Growth by Project Size	7 7
16.	Causes of Modifications Correlated by Design Agency	79
17.	Professional Experience of Sample Population	86
18.	Review Time Needed versus Time Available	91
19.	Frequency of Site Visits by Type of Review	105
20.	Frequency of Site Visits by Response Group	106
21.	Emphasis on Reviews	110
22.	Comparison of Findings - Causes and Costs of Modificationsx	113

List of Tables (Continued)

Table	F	page
23.	Comparison of Findings - Mod Causes	114
24.	Benefit/Cost Analysis of Constructibility Review (Finley's Study)	115
25.	Comparison of Findings - Design Review Process	118
26.	Comparison of Findings - Problems in the Review Process	120
27.	Comparison of Findings - Cost Growth as Function of Project Size	127
28.	Comparison of Findings - Cost Growth as Function of Designer	129
29.	Project Summary by Cost of Modifications	154
30.	Project Summary by Items of Change	156
31.	Design Deficiencies - Summary by Cost	159
32.	Design Deficiencies - Summary by Items of Change	160
33.	Correlation by Review	162
34.	Correlation by Project Type	163
35.	Correlation by Project Size	164
36.	Correlation by Design Agency	165
37.	Indirect Costs for Engineering, Design, and Overhead by Project	167

хi

. .

List of Figures

Figure	P	age
1.	Sample ENG Form 3938	36
2.	Sample ENG Form 3938-B	37
3.	Sample Standard Form 30	3 8
4.	Sample Contractor's Proposal	42
5.	Sources of Modifications by Cost	57
6.	Sources of Modifications by Items of Change	57
7.	Cost Growth Frequency Distribution	61
8.	Frequency Distribution of Engineering, Design, and Overhead - Cost per Item of Change	64
9.	Frequency Distribution of Engineering, Design, and Overhead - Cost per Modification	65
10.	Cost Growth of Projects not Receiving all 3 Reviews	67
11.	Cost Growth of Projects Receiving all 3 Reviews	6 7
12.	Cost Growth by Project Type - AFH Projects	70
13.	Cost Growth by Project Type - OMA Projects	71
14.	Cost Growth by Project Type - MCA Projects	71
15.	Cost Growth by Project Size - Projects Less than \$500,000	74
16.	Cost Growth by Project Size - Projects Between \$500,000 and \$1,000,000	74
17.	Cost Growth by Project Size - Projects Over \$1,000,000	7 5
18.	Cost Growth by Design Agency - Contract A/E Firm	78
19.	Cost Growth by Design Agency - In-House Government Designers	79
20.	Professional Registration	84

List of Figures (Continued)

ſ

1

Figure	F	age
21.	Ability to Conduct Reviews	88
22.	Review Qualifications versus Workload in Each Design Discipline	88
23.	Opinions on how Often Reviews are Needed	89
24.	Opinions on how Often Reviews save the Government Money	89
25.	Frequency of Sufficient Review Time	92
26.	Reasons Review Time Not Available	93
27.	Review Priority - "As Is" versus "Should Be"	94
28.	Checklist Availability versus Use	98
29.	Frequency of Reviews at Each Design Stage	99
30.	Frequency at which the Same Individual Conducts Reviews at Each Design Stage	100
31.	Availability of Previous Review Comments	101
32.	Frequency of Review Comments Being Incorporated Into Final Design	103
33.	Frequency of Feedback Given to Reviewers	103
34.	Percentage of Reviews with Site Visits	105
35.	Reasons for Not Conducting Site Visits	108
36.	Emphasis on Reviews by "The System"	109

xiii

CHAPTER 1

INTRODUCTION

Background:

The Department of Defense annually spends billions of dollars on military construction. Much of this is managed by the United States Army Corps of Engineers. The Government's trend over the last few years has been to increase the amount budgeted for construction, as shown in Table 1. This trend is likely to be reversed, however, because the spiraling federal deficit has resulted in increasing pressure to reduce overall defense spending. Consequently, it will become increasingly important to spend each defense dollar more efficiently. The Corps of Engineers can assist in this effort by reducing contract modifications and accompanying project cost increases.

TABLE 1

The Trend in DOD Military Construction Spending (1)

Fiscal Year (\$ Millions ~ FY 86 Dollars)

<u>1976 1980 1982 1983 1984 1985 1986 (Projected)</u> 4,326 2,969 5,756 5,076 4,872 5,724 7,057 The potential for savings in construction contracting is substantial. Corps of Engineers contract audits are credited with saving \$20.1 million in FY 1984 as part of the Inspector General's efforts to curb waste, fraud, and abuse (2). There is much more, however, that can still be done. Table 2 summarizes findings from three previous studies on construction contract cost growth. All projects shown were federally funded, although not all of them were administered by the Corps of Engineers. Table 2 is simplistic; it does not take into account project size, project type, scope of work, period over which the data was collected, or other factors which may influence cost growth. When the percentages in Table 2 are applied to the budget amounts shown in Table 1, it is obvious that savings of millions of dollars are possible if cost growth is reduced by only a few percentage points per project.

TABLE 2 Cost Growth on Federally Funded Projects

			Summed				
	Number	r of	Original Cost	Percen	itage Cost	Growth	
Study	Projes	<u>sts</u>	of_all_Projects	<u>Minimum</u>	<u>Maximum</u>	<u>Average</u>	
Rosmond (3)	300	\$333,354,000	-	-	8.3%	
Diekmann	(4)	22	\$103,900,000	-4.5%	72.5%	4.8%	
Rowland (5)	20	\$ 81,053,000	0.4%	24.0%	7.1%	

NOTE: Numbers in parenthesis in column 1 are keyed to chapter endnotes. Note that Rosmond's study only reported average cost growth; the maximum and minimum figures were not given.

Military construction contracts are normally awarded by the Corps of Engineers on a lump sum, competitive bid basis. The purpose of this system is to insure the government obtains a project built to the specifications and schedule desired at the lowest possible price. This system is also designed to protect the contractor by providing him all known project requirements prior to bid submission. Theoretically, at the time the contract is awarded, both parties know what the contract requires, when it is to be done, and what it will cost.

A contract modification is defined as "any written alteration in the specification, delivery point, rate of delivery, contract period, price, quantity, or other contract provisions of an existing contract, whether accomplished by unilateral action in accordance with a contract provision, or by mutual action of the parties to the contract." (6) The Corps of Engineers divides modifications (or "mods") into two categories. The first category is "change orders". Change orders are changes to the contract that fall within the contract's general scope of work, and are governed by the "changes" clause of the standard construction contract.(7) The second category of mod is "supplemental agreements". Supplemental agreements are changes that fall outside the original scope of the contract. Α supplemental agreement "is therefore a new negotiated contract which must be founded upon offer, acceptance, and consideration."(8) The supplemental agreement is added as a supplement to the existing contract for administrative convenience.

Once a construction contract is awarded, any changes to that contract are processed as modifications. Any changes outside the original contract scope or beyond the contractor's control (such as excessive periods of bad weather) may entitle the contractor to

additional compensation, time, or both. Unlike the original contract, which was competitively bid, modifications are most often "sole source" procurement actions. The net result is that contract modifications place the government in a less than ideal bargaining position and usually involve an increase in construction cost. As Lloyd Finley stated in his study of constructibility reviews:

"Any such required adjustment (or change order) becomes, in effect, a separate, sole source agreement, lacking in the competitive atmosphere present during contract award and significantly impacting the ability of the government to assume the best (or lowest) cost for the work." (9)

1

The cost of modifications can be measured in ways other than just direct construction costs. Indirect costs are incurred for processing, negotiating, and coordinating contract changes. These actions also consume a great deal of time that could be better spent elsewhere. Often modifications result in time extensions that push back project completion dates. These delays, in turn, upset the customer's scheduled use of the facility. Finally, projects with high cost overruns and excessive delays negatively impact on the Corps professional reputation. Therefore, reducing the number of "mods" can save taxpayer dollars, allow the Corps greater productivity, and provide a more professional and timely product to the supported installation.

Data collection for this study was conducted between February 1985 and August 1986 from 25 military construction projects administered by the Corps on an Army installation. These included projects funded under the Military Construction, Army (MCA), Operations and Maintenance, Army (OMA), and Army Family Housing (AFH) programs.

The supporting District was located about 40 miles from the installation upon which construction took place.

The goal of this study was to determine the causes and costs of contract mods and to determine why design reviews were failing to identify problems prior to construction. An additional goal was to determine how applicable the findings from this relatively limited sample of projects were to military construction in general by comparison with findings from other studies. This thesis summarizes the methodology, data, and findings of that effort.

Research Objectives and Problem Statement:

The research objectives of this thesis were to:

- identify the level of cost growth of sample MCA, OMA, and AFH construction projects.
- identify the reasons for modifications to military construction contracts.
- determine the administrative overhead costs of modification processing.
- 5. determine if the design review system used by the supporting District was adequately catching potential problems prior to construction, thereby precluding the need for all or most later modifications.
- determine if the situation observed was unique to the installation under study or if it was consistent with situations described in other studies.

This study addressed those objectives by asking the following

research questions:

 Did project cost growth at the installation exceed acceptable levels?

- How much did modification administration and overhead cost the District in terms of time, money, and productivity?
- 3. What were the causes of contract modifications?
- 4 Did the current design review system adequately identify and correct problems prior to contract award? If not, why?
- 5. Do the findings of this study track with findings of modification studies conducted elsewhere? Are the findings of this study applicable to Corps military construction in general?

The paper was organized to examine each question in the following sequence. Chapter 2 presents an overview of related studies and writings on the subject of construction contract modifications. Chapter 3 is a detailed description of the methodology used to collect data from the projects selected for study.

Chapter 4 presents data on modification costs and causes by addressing both direct and indirect costs. Data for this Chapter were collected from primary source contract documents. Mod nost and causes were examined by project type, project size, design agency, and by whether preconstruction reviews were conducted on project design prior to start of construction. The purpose of this cross-correlation was to determine the influence, if any, each of these categories has on contract mods.

Chapter 5 discusses the preconstruction review system and examines why reviews seemingly failed to identify potential problems prior to project start. Most of the data for Chapter 5 came from

questionnaires completed by District personnel involved with the review process.

Chapter 6 compares the findings and conclusions of this study with those findings and conclusions stated in the other papers discussed in Chapter 2.

Chapter 7 summarizes this study's findings and conclusions, and identifies areas requiring further study. Operational definitions, data tables, and a summary of questionnaire responses are included in appendices.

Assumptions:

The following five assumptions were made at the beginning of the study:

1. The impact of contract modifications can be guantified by:

a. Direct and indirect cost (dollars). Direct cost is the money paid to the contractor for labor and materials. Indirect cost is the cost for administrative overhead to the District.

b. Project cost growth (percentage).

c. Total number of mods per project.

d. Total number of changes per project. (Note that each modification may include more than one item of change.)

e. Time spent on mod negotiation, administration, and processing (manhours).

f. Impact cost. These are costs derived from the impact changes to the original design and schedule may have on elements not directly related to the modification. Impact cost, however, were not addressed in this study. 2. Most military construction projects will have some changes made during the course of construction. The Corps of Engineers does not formally define what these levels should be. Consequently, it is necessary to establish what an acceptable level of cost growth might be. Cost growth percentages were determined based on conversations with construction experienced personnel within the District and installation DEH. The percentages shown below represent a consensus of the opinions expressed during those interviews. These levels were determined to be:

a. 3% for high cost, "from the ground up" new construction.

b. 5% for new work in existing buildings and non-complex structural and/or architectural renovation work.

c. 10% for renovation work involving complicated electrical or mechanical work, particularly where that work involves going inside the walls of older buildings.

These figures may fluctuate somewhat with the size of the project. For example, a low cost, complex renovation project may sustain a higher cost growth since the price of each change will consume a larger percentage of the original contract amount.

3. The quality of a preconstruction review is primarily dependent on three factors. Therefore, an evaluation of these factors provides an indicator of the quality of the reviews being performed. These factors are:

> a. The professional qualifications, education, training, and experience of the individual doing the review.

b. The time available for the conduct of the review.

c. The organizational and procedural framework within which reviews are conducted. This includes the system by which review results are analyzed and appropriate action taken.

These review factors described above can be evaluated by the use of questionnaires filled out by those responsible for the conduct of reviews.

4. There are three types of preconstruction reviews that have a direct impact on modifications during construction. These are using service reviews, technical reviews, and constructibility reviews. The failure to thoroughly conduct any one of these may result in extensive cost growth regardless of the detail in which the other two are conducted.

5. Three of the 25 projects selected for study were not yet complete when final data were collected. These were projects #4 (96% complete), #19 (99% complete), and #22 (B1% complete). It was assumed that these projects were substantially complete enough so that any additional mods incurred by them would not significantly alter the findings presented.

Constraints and Delimitations:

The following constraints and delimitations were identified during the course of the study:

1. Data were collected from Corps of Engineers, military construction projects only. No data was collected on civil works projects. Consequently, the findings and conclusions may not directly apply to civil works or work in the private sector. 2. The questionnaire used to evaluate the review system was distributed only to District employees involved with military construction.

3. In general, the search for previous studies and literature was limited to those completed since 1970. Older studies were used only if their methodology or findings had a direct and significant relationship to this study.

4. This study did not address the issue of contractor's who deliberately submit low bids in the hope of increasing their profits by submitting requests for modifications or claims after construction begins.

5. This study did not address the issue of A/E liability for design deficiencies, errors, and omissions.

6. This study did not address biddability and operability reviews since internal District procedures usually resulted in these being done in conjunction with the technical and constructibility reviews. Furthermore, only District conducted reviews were examined. Reviews conducted by the installation Director of Engineering and Housing (DEH), Division Headquarters, or private Architectural/Engineering firms for the Corps were not addressed.

7. Data on indirect costs was limited due to District accounting procedures. The only indirect costs accounted for by individual project were engineering, design, and overhead. Costs associated with construction activities, such as mod negotiation and contract administration, were accounted for under an overall

construction account and not credited to individual projects. Hence, the data on indirect costs is not complete.

8. Initially, it was decided to not include disputed claims requiring litigation or arbitration. This turned out not to be a problem, however, since only one such claim was found in all 25 projects for a relatively small amount (\$100). With this one exception, all mode studied were agreed to by both parties and resolved by project completion.

Significance_of_Study:

16 studies, reports, journal articles, and other papers were identified which addressed the issues raised by the objectives of this study. No one of these, however, could be considered conclusive as each drew its conclusions from a necessarily limited data base. No study was found which attempted to analyze and correlate the findings of these diverse studies. Therefore, the value of this effort is twofold. First, it adds to the general modification knowledge base by adding to the pool of data previously collected. Second, it systematically compares the findings of the identified studies with the findings of this study to determine general trends appearing in military construction projects. Minor variations in these trends can be expected between individual locations. But in general they should provide the Corps as a whole with a general model of what causes modifications, how much they can be expected to cost, and what efforts should be made in the area of design review to preclude excessive cost

growth. By identifying the major causes of project cost growth, Districts should then be in a better position to more efficiently focus their review efforts to reduce overall construction costs.

Endnotes, Chapter 1

1. U.S. Department of Defense, <u>Annual Report to the Congress: Report</u> of the Secretary of Defense on the FY 1986 Budget, FY 1987 <u>Authorization Reguest and FY 1986-90</u> Defense Programs, (Washington, D.C.: U.S. Government Printing Office, 4 February 1985), p. 293.

2. DOD, Annual Report to the Congress, p. 90.

3. James R. Rosmond, "Analysis of Low Bidding and Change Order Rates for Navy Facilities Construction Contracts", (Master's thesis, Naval Postgraduate School, Monterey, June 1984), p. 39 and p. 51.

4. James E. Diekmann and Mark C. Nelson, "Construction Claims: Frequency and Severity", <u>Journal of Construction Engineering and</u> <u>Management</u>, (Vol. 111, No. 1, March 1985), calculated from data on p. 76.

5. Henry J. Rowland, "The Causes and Effects of Change Orders on the Construction Process", (Master's thesis, Georgia Institute of Technology, School of Civil Engineering, November, 1981), figures calculated from data presented in Appendix B.

6. <u>Construction Contract</u><u>Negotiating Guide</u>, (Office of the Chief of Engineers: Washington D.C., 1977), page 2-1.

7. Construction Contract Negotiating Guide, page 2-2.

8. Construction Contract Negotiating Guide, page 2-3.

9. Lloyd S. Finley, "Examination of the Constructability [sic] Review in Government Contracting", (Master's Thesis, Purdue University, School of Civil Engineering, 3 August 1984), page 4.

CHAPTER 2

SURVEY OF EXISTING PUBLICATIONS

General:

This chapter summarizes the findings of previous research projects and writings on construction contract modifications. This survey identifies 20 related studies, articles, and reports and provides a framework within which the relevance of the findings of this study can be evaluated. The ultimate objective is to determine if the findings of this study are consistent with the findings of these other works. A comparison between findings is made in Chapter 7.

In general, only publications or studies completed between 1970 and 1985 were reviewed. This 15 year period was arbitrarily selected in order to delimit the amount of material to be reviewed. Exceptions were made only if a particular work made a direct and significant contribution to the study of contract modifications. The publications reviewed are grouped under four categories:

- Government publications. Includes regulations, pamphlets, and other documents establishing procedural guidance for military construction.
- Official government studies. Includes studies carried out by government agencies or by civilian agencies under contract to the government.
- 3. Studies completed at academic institutions. Includes individual master's theses.
- 4. Articles from professional journals and magazines.

Publications were selected based on topic relevance and availabilitv. Studies and theses were obtained through the Defense Technical Information Service (DTIC). Abstracts of professional magazines and journals, dating back to 1970, were reviewed for pertinent articles. Several reports and studies were obtained through the Corps' District administering the projects under study. Many of the papers surveyed did not directly deal with the topics covered in this study. In some cases, only a few pages (or at best a chapter) was dedicated to modifications. In other cases, the author collected the same types of data but used them to address different problems. This survey discusses only those items contained in existing works that relate to the two main issues of this paper: the sources and costs of modifications and the relationship of design reviews to modifications.

Government_Publications:

The U.S. government publishes a number of documents regulating military construction. These begin with legislation enacted by the U.S. Congress, such as the Annual Military Construction Appropriation Act. Subsequent guidance is issued by the Department of Defense (DOD), Department of the Army (DA), the Corps of Engineers (COE), and Engineer Divisions and Districts responsible for project execution. The regulations written at each level are intended to establish the policies and procedures for implementation of guidance from higher authority.

Normally, guidance issued by the higher authorities tends to be more general in nature than that issued at the lower levels. Consequently, laws and regulations enacted at levels above the Chief of Engineers have little impact on day-to-day contract administration other than to provide a legal framework within which regulations issued at lower levels must comply. The regulations impacting most on military construction include ERs and District regulations listed in the bibliography. While they prescribe the procedures by which mods to military construction contracts are to be administered, they will not be discussed individually. Instead, the next few paragraphs center on three documents which summarize all these other regulations. These are Engineer Pamphlet (EP) 415-1-2 "Modifications and Claims Guide", the "Construction Contract Negotiation Guide", and a locally published "Resident Engineers Manual" used by the supporting District. These provide a good overview of how the procedure is to work.

EP 415-1-2, the "Modifications and Claims Guide", is issued as a guide only. It is not intended to supersede any regulation or contract requirement. The pamphlet is not directive in nature. Its purpose is to "provide a single source definition of the modifications and claims process which references all applicable regulations and relates their requirements in a coordinated manner."(1) It provides an excellent general reference document which clearly summarizes the essential requirements of applicable regulations. The pamphlet orients on those contract clauses dealing with mods and claims. It covers in detail initiation and processing procedures. The appendices include an

excellent glossary of terms, completed sample forms, and a regulation reference index. This pamphlet is an outstanding general reference on mod and claim procedures for construction contracts administered by the Corps of Engineers.

An excellent supplement to EP 415-1-2 is the "Construction Contract Negotiating Guide". (2) It is prepared by the Chief of Engineers' office as a training manual for personnel involved in contract and mod negotiations. As such, it also is not directive or regulatory in nature. There is some overlap between the material covered in these two documents. For example, both generically discuss the sources and types of modifications. The main difference is in orientation. Where EP 415-1-2 focussed on administrative procedures, the Negotiation Guide concentrates on topics related to negotiation of fixed-price contracts and mods. It includes discussion on pricing objectives, cost and profit analysis, negotiation procedures, and contract controversies. Like EP 415-1-2, this book is an outstanding reference for general background material.

The third government publication is District Regulation 415-2-1, the "Resident Engineers Manual". It provides an example of how higher level regulations are incorporated into operating instructions at the District level. Unlike the preceding two documents, this manual is directive in nature. Its purpose is

"to present and prescribe, in one publication, the basic duties, responsibilities, policies, procedures, and essential information for guiding Resident Engineers and their staffs in administering and inspecting construction contracts under their jurisdiction". (3)

The manual's applicability is limited to those projects under. District
control. Researchers studying the issue of modifications within other Districts may want to determine if a similar local policy document exists. If so, it can be a valuable source of information on local procedures.

Government_Studies:

ł

Three general sources of government studies were found. The first was the U.S. Army's Construction Engineering Research Laboratory (CERL) which has published numerous papers related to contract modifications, two of which were used in this study. The second source was the Office of the Chief of Engineers, which initiated two panel studies to evaluate the areas of installation support and construction quality. Lastly, two studies were found which the Corps had contracted from the Texas A&M Research Foundation. These dealt with the application of operations research techniques to construction operations. Each study will be discussed, with the emphasis on those areas relating directly to the issues of this paper.

The findings of the most recent CERL study were released in USA-CERL Technical Report P-85/11, "Ways to Improve Construction Contract Modification Processing: USAFEA Korea Case Study". (4) This case study was completed by the U.S. Army Facilities Engineer Activity (USAFEA) in Korea and summarizes actions taken by that agency to improve mod processing within its area of responsibility. Broad application of the findings may be limited since many of the problems identified are unique to the Korean environment, such as high personnel turnover, regulations governing materials, the language barrier, and a

lack of computer automation. Furthermore, most of the projects in Korea consist of maintenance and repair work to five-year "temporary" facilities still in use after 25 years. But many of the recommendations can be applied in a general sense. These include the discussion on policies concerning user requested changes, destructive testing to verify site conditions, the need to perform oversight and follow-through on design review comments, the need for formal management controls on mod processing, the need for automated mod tracking systems, and the need for increased use of electronic communications.

Another CERL document was published in May 1985. This was the preliminary draft of the "Automated Review Management" System (ARMS)".(5) This paper summarized the findings of a Corps wide survey conducted in 1983 and the results of numerous field visits, workshops, and interviews in 1984 concerning design reviews. The paper discusses the link between good reviews and reduced modifications. Those findings which pertain to this study include statements that 56% of modifications are due to design deficiencies, that too little time is being devoted to design reviews, and that current review procedures are not being centrally managed. ARMS was designed to provide a menu driven computerized management system to manage review suspenses, review comments, and develop a design review data base. It appears to be an excellent management tool with broad application.

The Chief of Engineers published the "Report of the Blue Ribbon Panel on Management of Construction Quality in the U.S. Army Corps of Engineers" in March, 1983. (6) The panel s mission was to find methods

to improve the quality of Corps construction, effectiveness of contractor quality control, and the level of Corps quality assurance administration. It did this by conducting workshops, interviews, questionnaires, site visits, and by reviewing previous studies, such as the Business Roundtable Reports. Modifications per se were not specifically addressed. Instead, the study focussed on general construction quality, and addressed issues such 25 review effectiveness, cross training between Design and Construction personnel, and the need to reduce the number of mods and change orders.

The "Report of the Green Ribbon Panel on U.S. Army Corps of Engineers Support to Army Installation Commanders" was prepared by the Chief of Engineers in March, 1985. (7) The purpose of the report was to recommend initiatives and improvements in Corps support to military installations. The report does not directly address modifications or their impact on installation support. Much of the report echoes the findings of the Blue Ribbon Panel completed two years earlier. The Green Ribbon Panel report emphasized the need for electronic automation to assist in installation support. It highlighted the need for rapid funding procedures to fix deficiencies. It cited the need for better review quality and the importance of review comment feedback. All of these issues were applicable to the situation observed during this study.

The Corps contracted for two studies from the Texas A&M Research Foundation in the late 1960's. "A Systems Approach to Design and Construction for the Corps of Engineers" was published in May 1968. (8) The purpose of the study was to determine the method of operation

the Corps uses to accomplish military construction and to structure this operation as a system. The study did an analysis of the military design and construction mission using systems engineering techniques. The study identified different sources of mods. Furthermore, it prescribed specific solutions to the problem of modification reduction. Among these solutions were more complete reviews, relief from statutory limitations on the amount of money set aside for reviews, and the delegation of greater authority to Resident Engineers. Though somewhat dated, this study provides excellent background material and recommendations that still apply.

The second study, a "Systems Analysis of Corps Design Engineering", was prepared in June 1969. (9) This study found eight problem areas in the Corps' design process, two of which relate to the subject of this paper. These are the areas of design data and design review, both of which will be discussed in subsequent chapters. The major strength of this study is its practical recommendations for solving the problems identified. Specific findings of interest here include the recommended use of checklists to assist in reviews, the need for and means of providing adequate project scope definition prior to start of design, and the lack of a rational, analytic review policy in the Corps.

In September, 1985, the Corps of Engineers South Atlantic Division sponsored a forum on design quality. The meeting was attended by members of the Division as well as representatives from numerous private A/E firms routinely involved in the design of Corps projects. The forum was called in response to the perception that design quality

within the Savannah District had diminished. The purpose was to obtain A/E opinions as to the significant things that adversely affect the quality of the product they provide to the Savannah District. The results of the forum were published in minutes dated 3 September 1985. (10) The views expressed by the A/E participants in many ways parallel observations made during data collection for this study. Specific problems included inadequately defined project scopes, lack of a first rate effort on design reviews, lack of A/E involvement in field work, failure on the part of the District to coordinate and edit review comments before turning them over to designers, and a lack of continuity in review when different individuals review projects at each design stage. Numerous other issues were discussed as well. These minutes provide insight as to the perception of the Corps held by the private engineering community and how that perception is affected by Corps practices. Although the forum only involved the South Atlantic Division, it is not unrealistic to expect A/E firms contracted by other districts to hold similar opinions under similar circumstances. Hence, the results of that forum have Corps wide applicability.

Studies_from_Academic_Institutions:

Four master's theses were found that addressed the issue of contract modifications in federal construction. The first was written by Henry J. Turowski at the Naval Postgraduate School in Monterrey and titled "Contractor Quality Control". (11) The thesis examined the government's contractor quality control (CQC) program. Its purpose was to research the attitudes of individuals directly involved in the CQC process. Only the chapter on alternatives to the CQC program directly related to the issues of modifications and design reviews. Turowski discussed using good design reviews as a means of improving construction quality. The study recommends separate design review teams, a conclusion which corresponds to that of this paper. The study also calls for increased designer responsibility and involvement in the construction process once a project begins. Turowski's paper only devotes a few pages to topics related to this study. However, the information in those pages relating to design reviews are in keeping with the findings of this study and will be discussed further in Chapter 6.

Henry J. Rowland completed his thesis on "The Causes and Effects of Change Orders on the Construction Process" while a student at the Georgia Institute of Technology in November 1981. (12) The objective of his study was to highlight complications which relate to legal disputes in construction. A secondary objective was to quantify the causes and effects of contract change orders. The data for this second objective was collected from projects administered by the Southern Division of the Naval Facilities Engineering Command. Ιn several ways, however, the study is disappointing. The causes for each of the types of modifications are not identified. The author deliberately selected projects for study which had many modifications; consequently, it is not possible to calculate a meaningful "average" number of mods per project. Cost growth is not discussed at all. Finally, it is the opinion of this author that the paper failed to meet the objective implied in the title. Instead of defining the causes and

effects of change orders, he develops general "rules of thumb" for construction projects. But, since his data base consists of only "problem" projects, the general applicability of these rules is suspect. Regardless of these shortcomings, much of the raw data Rowland collected on construction and modification costs was able to be processed using the methodology of this study. This, in turn, allowed for an "apples-to-apples" comparison between the two studies. The purpose of James R. Rosmond's work on "Analysis of Low Bidding and Change Order Rates for Navy Facilities Construction Contracts" is to evaluate what effects bidders of fixed price contracts (government construction) have on contract prices when the level of bidding intensity increases. (13) His basic thesis is that in environments of intense competition, contractors are willing to assume more risk in preparing estimates in order to be awarded contracts. The contractor may then attempt to improve his financial position by the use of change orders after construction begins. Rosmond draws his data from a sample of projects administered by the Western Division of the U.S. Naval Facilities Engineering Command. He provides a good discussion on the sources of modifications and on the use of "change order rates" (cost growth) to assess contractor performance.

The last study was an "Examination of the Constructibility Review in Government Contracting" by Lloyd S. Finley, Purdue University, August 1984. (14) The purpose of his study was to examine the constructibility review process in construction projects administered by the Navy. Finley presents data collected from two constructibility reviews, calculates the direct and indirect costs of

the reviews, and then determines the benefits derived by calculating what the items identified during the review would have cost had they been left in the contract and later modified. Finley makes an excellent case for the cost effectiveness of constructibility reviews. For one project, the benefit-cost ratio for the review was 2.8 to 1. For the other, it was 28.4 to 1. He further makes the point that reducing mods by review improves the overall contract management effort by improving the productivity of the contract administrators. He also supports the use of review checklists to provide consistency. His methodology is excellent, and could be used to evaluate the effectiveness of constructibility reviews anywhere. A major shortcoming, however, is in the limited data base. He uses the data from two projects to draw rather sweeping conclusions about the value of constructibility reviews.

Articles from Professional Journals and Magazines:

Journal and magazine abstracts were reviewed for the period 1970~1985 for articles related to the subject of construction contract modifications. One was found in <u>The Military Engineer</u>, one in <u>Engineer</u> <u>News Record</u>, one in <u>Civil Engineering</u>, and two in journals of the proceedings of the American Society of Civil Engineers (ASCE). An additional article was found on a standardized review checklist in a reprint of an article first appearing in the <u>DPIC Communique</u>.

The article in <u>The Military Engineer</u> was titled "Productivity Improvement" by Dennis L. Ballou. (15) The author stated that one of the detractors from good productivity is contract disputes. He claimed reducing time to settle disputes and avoiding litigation is essential to increasing productivity. Sources of disputes were listed as imprecise specification language, ambiguous drawings, unknown site conditions, and owner~initiated changes. The author did not address eliminating the sources of disputes. Although not dealing directly with modifications, much of what the author says about the relationship between disputes and productivity is applicable to modification processing. The article provides good background material, even though not directly related to the subject of this paper.

Engineer_News_Record ran a very short article titled "Design Changes: The Largest Cause of Overruns" in March 1975. (16) The article cited an unnamed Government Accounting Office report which studied 269 federal projects, most of them construction, that experienced cost overruns. The information provided on the study was very sketchy. Of interest to this study was the fact that 41% of cost overruns were found to be caused by design changes. 59 projects had cost overruns of 100% or more; 47 of these were administered by the Corps of Engineers. Since the study upon which the article was based was not identified and therefore could not be reviewed in detail, the applicability of the limited information contained in the article was limited.

The "Forum" section of <u>Civil Engineering</u> magazine published the comments Walter P. Moore delivered at a keynote address to the 1984 annual meeting of the American Concrete Institute. (17) Titled "Structural Safety: The Profession at a Grossroads", the article provided an excellent discussion as to why civilian engineer firms no longer have time to perform thorough reviews or visit project sites. The author claims the prime reason for this is the increased "fast tracking" of projects to save time and the reluctance of owners to budget the necessary time and money for reviews. He further blames the shift of supervisory responsibility from the architect to the project owner. The applicability of this article to federal projects lies in the fact that much federal design is done by civilian Architect/Engineer (A/E) firms that are operating under similar time and budget constraints. Although not directly related to this study, Moore's comments provide insight into some of the problems within the private engineering community. They provide an interesting supplement to the views expressed during the Design Quality Forum.

Harvey Kagan's article "How Designers Can Avoid Construction Claims", Journal of Professional Issues in Engineering of the ASCE provides more information of direct interest to the study of mods.(18) Kagan divides his article into a discussion of contract documents and shop drawings. In his discussion of contract documents, he notes that mods are often the result of standard contracts (such as those use by the Corps). He claims that designers often do not read the clauses of these "off-the-shelf" contracts. The result is contracts containing conflicting clauses which leads to later claims. He emphasizes the need for good reviews to resolve these problems early in the design cycle.

"Construction Claims: Frequency and Severity" was written by James E. Diekmann and Mark C. Nelson and published the results of a study they completed at the University of Colorado, Boulder. (19) Data on mod causes, sources, costs, and cost growth of 22 federally administered projects involving 427 claims and modifications was presented. The Diekmann-Nelson study found design deficiencies to be the most frequent cause of construction claims. They further studied the impact of project size and design agency on modifications. Their approach was very similar to the methodology employed here and provided an excellent reference for comparing their results with those of this study.

One of the issues raised during this study was the use of review checklists to assist in the review process. Checklists can be an aid to design reviewers in conducting their reviews. There is a problem, however, in designing such a checklist that is flexible enough to cover the myriad projects the Corps administers yet is detailed One such checklist developed to meet these enough to be of any use. criteria is "Redicheck". In an article written for the Design Professionals Insurance Company Communique April 1984 issue, William Nigro described what "Redicheck" was and what it could do.(20) He describes "Redicheck" as "a structured review system which provides a logical and orderly approach to checking construction drawings." The system is designed to be used in the later stages of design to assist in coordinating drawings with written specifications and comparing drawings dealing with one design discipline with those of another. It is not a technical review, and is not intended for use as a document review within any one discipline. The author goes on to provide quidance on how the system should be used. "Redicheck" was favorably referred to in two of the other papers surveyed for this study. Hence,

it may be a system the Corps should investigate for use on their construction projects.

SUMMARY:

This chapter provided an overview of existing papers, articles, studies, and reports completed over the past 15 years and relating to the research objectives of this study. Whereas numerous authors have addressed some aspect of construction contracting, relatively few have specifically targeted the issues examined in this study. Consequently, this study has the potential to make a significant contribution by not only adding to the existing data base on mod causes and costs but also by tying together the findings of these various other studies.

Endnotes, Chapter 2

1. U.S. Army Engineer Pamphlet (EP) 415-1-2, <u>Modifications and Claims</u> <u>Guide</u>, (Washington D.C.: Office of the Chief of Engineers, 1 October 1976), page 1-1.

2. Office of the Chief of Engineers, <u>Construction Contract Negotiating</u> <u>Guide</u>, (Washington D.C.: Office of the Chief of Engineers, FY 77 Edition).

3. District Regulation 415-2-1, <u>Resident Engineers Manual</u>, (1 August 1978), page 1-1.

4. USA-CERL, "Ways to Improve Construction Contract Modification Processing: USAFEA Korea Case Study", (CERL Technical Report P-85/11, May 1985).

5. USA-CERL, "Automated Review Management System", Users Manual, 3 May 1985.

6. Corps of Engineers Report of the Blue Ribbon Panel on "Management of Construction Quality in the U.S. Army Corps of Engineers", (Office of the Chief of Engineers, March 1983).

7. Corps of Engineers Report of the Green Ribbon Panel on "U.S. Army Corps of Engineers Support to Army Installation Commanders", (Office of the Chief of Engineers, March 1985).

8. Texas A&M Research Foundation, "A Systems Approach to Design and Construction for the Corps of Engineers", (TR 68-041, College Station, Texas, May 1968).

9. Texas A & M Research Foundation, "Systems Analysis of Corps A/E Design Engineering", (College Station, Texas, June 1969).

10. Corps of Engineers South Atlantic Division, "Minutes of the Design Quality Forum", (Savannah District, 3 September 1985).

 Henry J. Turowski, "Contractor Quality Control", (Master's thesis, Naval Postgraduate School, Monterrey, December 1980).

12. Henry J. Rowland, "The Causes and Effects of Change Orders on the Construction Process", (Master's thesis, Georgia Institute of Technology, School of Civil Engineering, November 1981).

13. James R. Rosmond, "Analysis of Low Bidding and Change Order Rates for Navy Facilities Construction Contracts", (Master's thesis, Naval Postgraduate School, Monterey, June 1984).

14. Lloyd S. Finley, "Examination of the Constructability [sic] Review in Government Contracting", (Master's thesis, Purdue University, School of Civil Engineering, 3 August 1984).

.

15. Dennis L. Ballou, "Productivity Improvement", <u>The Military</u> Engineer, (Vol. 77, Sept-Oct 1985), page 484. Ballou is the Assistant General Manager for Transit System Development at the Metropolitan Atlantic Rapid Transit Authority.

16. "Design Changes: The Largest Cause of Overruns", <u>Engineer News</u> <u>Record</u>, (6 March 1975), page 10.

17. Walter P. Moore, "Structural Safety: The Profession at a Crossroads", <u>Civil Engineering</u>, (July 1985), pages 7-8. Excerpted from ACI's <u>Concrete International</u>, October 1984. Moore is president of the Houston based structural, civil, and traffic engineering firm of Walter P. Moore and Associates.

18. Harvey A. Kagan, "How Designers Can Avoid Construction Claims", Journal of Professional Issues in Engineering, Proceedings of the American Society of Civil Engineers, (Vol. 111, No. 3, July 1985), page 100. Kagan is a professor of Civil Engineering at Rutgers University.

19. James E. Diekmann and Mark C. Nelson, "Construction Claims: Frequency and Severity", <u>Journal of Construction Engineering and</u> <u>Management, Proceedings of the American Society of Civil Engineers</u>, (Vol. 111, No. 1, March 1985), page 74.

20. William T. Nigro, "Redicheck: A System of Interdisciplinary Coordination", <u>DPIC__Communique</u>, (Design Professionals Insurance Company, April 1984), pages 1-4.

CHAPTER 3

METHODOLOGY

General:

This chapter details the methods and procedures used during the conduct of the study. All project data was obtained from primary sources, such as contract documents, project correspondence, standard administrative forms, questionnaires, and discussions with individuals involved with modification processing. Secondary sources were used in the review of previous studies and writings as discussed in Chapter 2.

Foth completed and ongoing projects were selected for study. Frojects were selected from an arbitrarily chosen window of 18 months running from 1 January 1984 through 30 June 1985. Specific criteria for project selection were:

- 1. All projects were to be located on the supported installation.
- Completed projects must have been finished after 1 January 1984.
- 3. Ongoing projects may have started at any time prior to 30 June 1985 providing a minimum of six months (as of 30 June) passed since issuance of the notice to proceed <u>OR</u> the project was scheduled for a minimum of 50% completion as of 30 June 1985.
- Plans, specifications, and project files for all projects had to be available at the Resident Engineer Office.

The purpose of these criteria was to insure all projects selected would be substantially complete by the time this paper was to be completed and to insure project files would be readily available. A total of 25 projects were selected based on this criteria. These projects are identified throughout this paper by a project number (1 through 25). Each project number was assigned based on project start date. The earlier the project was started, the lower the number assigned.

These 25 projects represented a mix of five MCA, 15 DMA, and five AFH funded projects and consisted of a total of 268 modifications. These mods included 778 individual items of change. Project award amounts ranged from \$71,000 to \$8,000,000. All but three projects were completed as of the date of this paper. As stated in the assumptions in Chapter 1, these three projects were considered substantially complete enough to be included in the study.

There were three other projects, however, which contain modification costs of such magnitude that they distorted the findings. These projects were:

Project 5: Project 5 contained a \$260,194 mod to remove asbestos insulation.

Project 15: This contract was issued with a poorly defined scope in an hurried attempt to take advantage of year-end funds. The project underwent three scope revisions, with associated modifications, after work began. Project cost growth totaled \$477,408 (89%). 40% of the increase was due to user requested changes; 39% due to site conditions.

Project 22: Project 22 replaced central heating and air conditioning in family housing units. After the contract was awarded,

a design criteria change was issued concerning ductwork installation. This change cost \$319,250.

Combined, these account for over 44% of the modification cost for all 25 projects and have a disproportionate impact on the overall results. Consequently, unless otherwise noted, the high cost mods and Project 15 were deleted from calculations in the remainder of this study.

The study was organized into three phases, each of which is the basis for one of the three following chapters. The titles of these phases are:

- 1. The Causes and Costs of Modifications.
- 2. The Preconstruction Design Review System.
- 3. Comparison of Findings with Previous Studies.

Each involved a different research methodology. Mod causes and costs were determined by studying information contained in project files and contract documents. Data on the review system was collected by a questionnaire distributed to District personnel involved with military construction. The comparison of findings used a descriptive methodology to contrast results of related studies to the findings of this study. Each methodology is described in detail in the sections which follow.

Methodology for Determining the Causes and Costs of Modifications:

Once negotiated and agreed to by the parties concerned, modifications become part of the construction contract. Consequently, the terms of each mod were usually well documented. Most of the data collected in this segment of the study came from reviewing project files. Primary information sources were ENG Forms 3938 and 3938-B (Contract Modification Proposal and Acceptance), Standard Form 30 (Ammendment of Solicitation/Modification of Contract), memorandums for record, contractor proposals, shop drawings, and other documents or correspondence relating to each project studied.

ENG Form 3938 (Figure 1) was used for modifications within the approval authority of the resident contracting officer. On this form were recorded the details of the changes to be made, the total cost of the changes, and any changes in contract time negotiated with the contractor. This form was designed to be a bilateral contract agreement and is signed by both the contractor and the resident contracting officer. It then becomes part of the contract. ENG Form 3938-B (Figure 2) was used to supplement the information contained in Form 3938. The 3938-B was used to record the reason(s) why the change was necessary and provided a brief summary of the negotiations between the contractor and the government. Standard Form (SF) 30 (Figure 3) was used for modifications outside the approval authority of the resident contracting officer. The information contained was similar to that of Form 3938. The SF 30, however, was signed by the contracting officer instead of the resident contracting officer. These forms were maintained for each modification to each+project. Consequently, excellent primary sources were available to identify individual changes and their corresponding cost.

	CUNTRACT MOE (Me	odification of the 1	f less than \$50,000) form, and ER 1180-1-1.	SX002
ISSUING OFFICE			2. CONTRACT NO.	3. MODIFICATION NO.
	من شده معند المعند المعند مستقدم المعند			P00001
TO (Contractor)			5. PROJECT LOCAT	ION AND DESCRIPTION
	ويعاد والمتحد الأستر الرواي و	, Carteria de la	Inc.	
رتش مستعد شدار ومن الارجاد بردار	مین نفریس			
A proposal is requeste atterment hereto, Su suse of mis contract en	d (or making the hereinejter des bmit vour propose in spece ind titled. Modification Proposes	ended change licated on page - Price Breakd	n eccordance with speci 2. attach detailed bro own") DO NOT start w	fluetion and drawing revisions cited herein or lie abdown of prime and subcontract cotts. See s ork under into proposed change witil you receiv
pv signed by the Contro	icting Officer or a directive to pi	roceed.		
(inte	Tuesd home	A Treta	. <u> </u>	S. and then
				21840 itere
Reference Com	work necessary to accomplish in tract Drawing Sheet:	e foilowng de: s C5 and	C7: Replace e	xisting Manhole 22-95 with a
TCA LATERATC'				
<u>Time</u> : It is a performance for	inderstood and agree	ed that, d by this	pursuant to th Contract shal	e above, the time of 1 remain unchanged.
Payment: Due the amount of	to the changes desc \$1,588.00 increase	eribed he and is d	rein, the Cont esignated for	ract price will be adjusted payment purposes as follows:
Item_No.	Description	Unit	Unit Price	Amount Increase
P00001	Replace Manhole	3ор	Lump Sum	\$1,588.00
This is consident any and all effects	iered full and just Meets of the modif:	consider ication c	ation for the in the Contract	described modification and f as a whole.
Punds: This: PolOO \$23026 }	mdification is fund MA1060100182004.	ded under	Appropriation	No. 2112050 108-9725
				In Kachemark and in full force and affer
cept as hereby Madified	t, all terms and conditions of sur	d commect as h	erecolore modified reme	the state of the s
cept as hereby Modified	I, all terms and conditions of star	d comment as h	UNITED STATES	OF AMERICA
cept as hereby Modifie a torapone modificatio INTRACTOR	I. all terms and conditions of sau n is hereby accepted:	d comment as h	UNITED STATES	OF AMERICA
copt as hereby Modifie is tenessing modificatio INTRACTOR	d, all terms and conditions of say n is hereby accepted:	d comment as h	UNITED STATES	OF AMERICA
cept as hereby Madifie a terepaine modificatio INTRACTOR	d, all terms and conditions of six		UNITED STATES	OF AMERICA
copr as hereby Modifies a terespone mostlaste INTRACTOR	d, all terms and conditions of sale n is hereby accepted:		UNITED STATES	OF AMERICA
crpr as herre'y Modifie a Tangaing modifies to DITRACTOR	d. eli terms end conditione o/ sai n n heraby escapital:		UNITED STATES	OF AMERICA
CTPT AS NOTES Modified	d, all terms and conditions of sau n is hereby accepted : Signature		UNITED STATES	OF AMERICA
copt as hereby Modifie a torapping modification INTRACTOR	d, all terms and conditions of sal n is hereby accepted : 		UNITED STATES	OF AMERICA
-19-84	d, all terms and conditions of sale in a hereby accested: Signature Typed Name and Title		UNITED STATES	OF AMERICA
COPT as hereby Modifie a torogoing modifies to DNTRACTOR -19-84 Dere	d, all terms and conditions of sal in is hereby accested: Signature Typed Name and Title		UNITED STATES	OF AMERICA Synamice esident Contracting Officer Typed Neme and Tule
G FORM 3938,	d. eli terms end conditione o/ sui in in hermov escastal: Sieneture Typed Name and Title Jul 81 ED		UNITED STATES UNITED STATES 34 Jun 04 R Derr	OF AMERICA Symmute esident Contracting Officer Typed Name and Tule CP PAGE POP 2 PAGE

ONTRACT M	ODIFIC TION PROPOSAL AND ACCEP		CONTRACTING OFFICERS CO
2 HOBCT (الورومين الموغيل أخدين وطعهم أن 10 منه. الروم معمل أمرية في معرفتهم الماليني		
PUNDS PROGRAM	WED TOP THIS CHANGE IN THE AMOUNT OF		
\$1,588.0	0 Increase		
Necessity Hon -	HANGE AND REASON FOR DWISSION FROM MANS AN	o sachanons le is dilapidate	d.
Reason f	or Ordssion from Flans and St	ecifications: P	hysical condition of manhols
Vas not	known until after construction	n commenced.	
RESUME OF NEG	OTIATIONS OR LECOMMENDATIONS (Gui rement repres	widtit()	<u></u>
Jegotiat	ing Representatives: Governm Contrac	nt: Chinaland tor: Chinaland	
The cond Contract	ition of the manhole was foun or was verbally requested to .	d during the fir-	st part of May 1984. The 1 to replace it.
The Cont no days	ractor submitted his proposal requirement for a Time Satens	on May 18 1984 : ion.	in the amount of \$3,246.00 with
Since th was in a write br	e Contractor's price proposal ufficient detail to disclose estrious to determine an equit	was less than the element of ca the element of ca	25,000 and the price breakdown ost, it was decided to use the as permitted by 301-1-372(a).
The pric that the	• breakdown was reviewed by Contractor was pricing the r	epinement of the	May 1984. The review revealed o (2) manholes. On 23 May 1984
(1) manh 31 Hay 1	ole. The other manhole has d 984 the Contractor accepted to	r and he was did steriorated but : he offer.	is considered to be usable. Or
In my ju Extensio	igement the negotiated price : n in Contract time is not war	is considered equivalent of the second se	uitable and reasonable. It of this modification.
	TYPEC NAME AND TITLE OF SOVERNMENT REPRESENTA	fi ve	SIGN
4 Jun 34	Besident Contracting Office	ut .	
1G +0m 3938	-3		2 U.S. GOVERNMENT PRINTING OFFICE 1982-38

FIGURE 2 Sample ENG Form 3938-B

	3. EFFECTIVE DATE	4. REQUISITION/PUR	CHASE REQ. NO. S. PROJECT NO. III APPlicab
+00014	See Block 16c	110000000000000000000000000000000000000	
155020 84			CODE
.S. Army Corps of Engineer	r9		
District		Renovate	
NAME AND ADDRESS OF CONTRACTO	R INO . street, county, State and	ZIP Code)	T A TAA. AMENDMENT OF SOLICITATION NO.
		no 1to	
	- Final K	ushuman.	
	1 cm		98. DATED ISEE ITEN 11)
•	ECOO RECON	CONTRACT FUE	104. MODIFICATION OF CONTRACT/ORC
			X
B/SA/P/C			
OF.	FACE ITY CODE		- 83 Sep 30
11 Terr	SITEM ONLY APPLIES TO	AMENDMENTS OF S	GL CITATIONS
The apove humbered solicitation is amen	ded as set forth in item 14. The h	Our and date specified for	
		and the second of the	
fers must acknowledge receipt of this amen	dment prior to the hour and date	specified in the solicitatio	in ar as amended by one of the following methods
By completing items 8 and 15 and returns smitted or ict By separate letter or telegri	ng	ment (b) By acknowledge the solicitation and amend	ing receipt of this amendment on each cody of the c iment numbers FAILURE OF YOUF ACKNOWLE
NT TO BE PECEIVED AT THE PLACE O RELECTION OF YOUR OFFER IN by VI	ESIGNATED FOR THE RECEIP intue of this amendment you desire	PT OF OFFERS PRIOR TO re to change an offer airea	O THE HOUR AND DATE SPECIFIED MAY RESI dy submitted, such change may be made by relegra-
er provided wach telegram or letter makes	reference to the solicitation and t	his amendment, and is rec	eived prior to the opening hour and date specified
2152050 508-9725 P6700-	3200 S23028 W/O 85	5-11 RJ50035011	182004 \$42.462.00
ti THISITE	M APPLIES ONLY TO MOD		
IT MODI	FIES THE CONTRACT/ORD	ER NO AS DESCRIB	ED IN ITEM 14 PRETICUSII
TRACT OPOER NO IN ITEM 104	PURSUANT TO Specify author	THE CHANGES SET	FORTH IN ITEM 14 ARE MADE IN THE COM
Clause 3, "Changes," o	f the Contract Gener	ral Provisions	
B THE ABOVE NUMBERED CONTRA appropriation date, etc.) SET FORTH	CT/ORDER IS MODIFIED TO R	HE AUTHORITY OF FAI	TRATIVE CHANGES (such as changes in Paying of) R 43.103(d).
C. THIS SUPPLEMENTAL AGREEMEN	IT IS ENTERED INTO PURSUAR	NT TO AUTHORITY OF	
O DINER Specify type of must freepor			
O JTHER Specify type of mudification			
D STRER Specify type of mudification	Tequested		CODY
MPORTANT Contractor so	Tequested	is document and return	one copy
MPORTANT Contractor son DESCRIPTION OF AMENDMENTIMODI	K requested	is document and return chon needings, including (n_one_copy solution control to the issuing office policitation/control to beet matter where feasible /
MPORTANT Contractor 500 Second Second Secon	Requested SAMAXAN Sign hi FICATION (Organized by UCF and labor, material and labor and sof	is document and return choo area inglimeter equipment to pe	<u>one</u> <u>copy</u> <u>solciano</u> , to the issuing office solciano, contract where matter where feelbe, rform the specific items of product b Shorto Labora
MPORTANT Contractor som obschiption of Amenomentimoon a. <u>Change</u> : Provide all 3 k, in the basement and oth ed and revised by Request	Requested SAMANAS organity FICATION (Organized by UCF and labor, material and ler areas of Contact for Proposal SXC20	is document and return chan needings, including equipment to pe in accordan iated 30 May 19	<u>one</u> <u>copy</u> <u>solcianon/contract whether what familie</u> , rform the specific items of nce with Sheets 1 thru 7 pro- 85, as hereafter listed:
MPORTANT Contractor son DESCRIPTION OF AMENDMENT/MODI a. <u>Change</u> : Provide all k, in the basement and oth ed and revised by Request	A prequested SECATION (Organized by UCF and labor, material and her areas of the proposal SXC20	is document and return chon breaking, including equipment to per fin accorda lated 30 May 19	<u>one</u> <u>Copy</u> <u>solcies</u> to the issuing office solcies non-icontract in West matter where feasible, rform the specific items of nce with Sheets 1 thru 7 pro- 85, as hereafter listed:
IMPORTANT Convertor some modification important Convertor some Description of Amendmentimdon a. <u>Change</u> : Provide all 1 k, in the basement and oth ed and revised by Request (1) Demolition; Remove	K spectral of the second state of the second s	is document and return chon hrading, including of equipment to per financial in accordan lated 30 May 190 ing in the based	<u>one</u> <u>Copy</u> <u>solcies</u> no the issuing office <u>solcies</u> non-indext matter where featble, rform the specific items of nce with Sheets 1 thru 7 pro- 85, as hereafter listed: ment not suitable for reuse,
IMPORTANT Convertor some obscaliption of amendmentimoduli a. <u>Change</u> : Provide all 1 k, in the basement and oth ed and revised by Request (1) Demolition; Remove directed by my field repre	K requested K ATION COMMENTS of the FICATION COMMENTS by UCF with labor, material and her areas of Comments for Proposal SXC20 e existing wood fram esentative.	is document and return chon hradings, including of equipment to per in accordan lated 30 May 190 ing in the based	<u>one</u> <u>copy</u> <u>solcies</u> to the issuing office <u>solcies</u> non-indext matter where familie, rform the specific items of nce with Sheets 1 thru 7 pro- 85, as hereafter listed: ment not suitable for reuse,
IMPORTANT Convertor 500 Secondary of modification a. <u>Change</u> : Provide all 1 k, in the basement and oth ed and revised by Request (1) Demolition; Remove directed by my field repre- (2) Complete all roug;	Requested (X) (Very and Stright in the FICATION (Organized by UCF with labor, material and ler areas of (C) for Proposal SXC20 e existing wood fram esentative. 	is document and return chon headings, including (equipment to per in accordan lated 30 May 19) ing in the baser itional partitic	<u>one</u> <u>copy</u> <u>solcies</u> to the issuing office <u>solcies</u> non-inset in <u>west</u> matter where familie, rform the specific items of nce with Sheets 1 thru 7 pro- 85, as hereafter listed: ment not suitable for reuse, poss in the basement as shown
IMPORTANT Convertor 500 Secondary of modification a. <u>Change</u> : Provide all 1 k, in the basement and oth ed and revised by Request (1) Demolition; Remove firected by my field repro- (2) Complete all roug; Sheet 5 referenced above.	REATION COMMENTS of the fillen and the fillen areas of t	is document and return chon needings, including (equipment to per in accordan lated 30 May 19) ing in the baser itional partitic	<u>one</u> <u>copy</u> <u>solution</u> to the issuing office <u>solution</u> the specific items of nce with Sheets 1 thru 7 pro- 85, as hereafter listed: ment not suitable for reuse, ons in the basement as shown
IMPORTANT Convertor 500 State Specify type of modification IMPORTANT Convertor 500 DESCRIPTION OF AMENDMENT/MODI a. <u>Change</u> : Provide all 1 k, in the basement and oth ed and revised by Request (1) Demolition; Remove (1) Demolition; Remove firected by my field repre- (2) Complete all roug; Sheet 5 referenced above.	Requested (X) (Very and Stright in FICATION (Organized by UCF with labor, material and ler areas of for for Proposal SXC20 e existing wood fram esentative. n-in framing for add	is document and return chan headings. Including (equipment to per in accordan lated 30 May 19) ing in the based itional partitio	<u>one</u> <u>copy</u> <u>solution</u> to the issuing office <u>solution</u> the specific items of nce with Sheets 1 thru 7 pro- 85, as hereafter listed: ment not suitable for reuse, ons in the basement as shown (continued)
IMPORTANT Convertor son IMPORTANT Convertor son DESCRIPTION OF AMENOMENT/MOOT a. <u>Change</u> : Provide all : k, in the basement and oth ed and revised by Request (1) Demolition; Remove lirected by my field repre- (2) Complete all roug! Sheet 5 referenced above.	Resentative.	is document and return chan headings, including (equipment to per in accordan lated 30 May 19) ing in the baser itional partition	<u>one</u> <u>Copy</u> <u>solcianovicentract where matter where familie</u> , rform the specific items of nce with Sheets 1 thru 7 pro- 85, as hereafter listed: ment not suitable for reuse, ons in the basement as shown (continued) stofore changed, remains wechanged and in full force
IMPORTANT Contractor son IMPORTANT Contractor son DESCRIPTION OF AMENDMENT/MOOT a. <u>Change</u> : Provide all : k, in the basement and oth ed and revised by Request (1) Demolition; Remove directed by my field repre- (2) Complete all roug! Sheet 5 referenced above.	Tequested , X (requested forgon in FICATION (organized by UCF with labor, material and ter areas of for for Proposal SXC20 e existing wood fram esentative. then framing for add there of the accument (elevenced organit)	is document and return chan needings, including (equipment to per in accordan lated 30 May 19) ing in the based itional partition in them 9A or 19A, as Der [16A, NAME AND YITLE	<u>one</u> <u>copy</u> <u>solcianovices to the issuing office</u> solcianovices to beer matter where <i>leader</i> rform the specific items of nce with Sheets 1 thru 7 pro- 85, as hereafter listed: ment not suitable for reuse, ons in the basement as shown (continued) stolore changed, remains unchanged and in full force E OF CONTRACTING OFFICER (Type or print)
IMPORTANT Contractor son DESCRIPTION OF AMENDMENTIMOOT a. <u>Change</u> : Provide all : k, in the basement and oth ed and revised by Request (1) Demolition; Remove directed by my field repre- (2) Complete all roug! Sheet 5 referenced above.	Tequested Tequested FICATION (organized by UCF with labor, material and ter areas of for Proposal SXC20 e existing wood fram esentative. the framing for add there of the document (elevenced argunt)	is document and return chan neadings, including equipment to per in accordan lated 30 May 199 ing in the based itional partition in them 9A or 19A, as her [16A: NAME AND YITLE	<u>one</u> <u>copy</u> <u>solution</u> to the issuing office solution the specific items of nce with Sheets 1 thru 7 pro- 85, as hereafter listed: ment not suitable for reuse, ons in the basement as shown (continued) stofore changed, ismain unchanged and in full force E OF CONTRACTING OFFICER (Type or punct)
IMPORTANT Contractor son DESCRIPTION OF AMENDMENTIMOOT a. <u>Change</u> : Provide all : k, in the basement and off ed and revised by Request (1) Demolition; Remove directed by my field repre- (2) Complete all roug! Sheet 5 referenced above.	Tequested Tequested FICATION (organized by UCF with labor, material and her areas of for Proposal SXC20 e existing wood fram esentative. h-in framing for add those of the document (sterenced or prot)	is document and return chan preduces, including i equipment to pe initian accordan iated 30 May 191 ing in the baser itional partition in tem 9A or 19A, as her liss. NAME AND TITLE Major, Coops	<u>one</u> <u>copy</u> <u>social</u> to the issuing office <u>social</u> to the issuing office <u>social</u> the specific items of nce with Sheets 1 thru 7 pro- 85, as hereafter listed: ment not suitable for reuse, <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(continued)</u> <u>(co</u>
IMPORTANT Contractor sono DESCRIPTION OF AMENDMENTIMOOT a. <u>Change</u> : Provide all is k, in the basement and oth ed and revised by Request (1) Demolition; Remove directed by my field repre- (2) Complete all roug: Sheet 5 referenced above. contactor, off ERGA	REALION COMMENTS of the second stands of the social and the second stands of the second stands of the second stands of the second stands of the social stand	is document and return chan headings, including i equipment to pe initiated 30 May 191 ing in the based itional partition in item 9A or 19A, as her 16A, NAME AND TITLE Major, Comps 16B, international	<u>one</u> <u>copy</u> <u>social</u> to the issuing office social social to be example to the issuing office social social to be example to be example to social to be example to be example to social to be example to be example to (continued) social to be example to be example to (continued) social to be example to be example to (continued) social to be example to be example to constant to be example to be example to be example to constant to be example to be example to be example to constant to be example to be example to be example to constant to be example to be example to be example to constant to be example to be example to be example to constant to be example to be example to be example to be example to constant to be example t
IMPORTANT Contractor of modification IMPORTANT Contractor of some DESCRIPTION OF AMENDMENTIMOOT a. <u>Change</u> : Provide all if k, in the basement and oth ed and revised by Request (1) Demolition; Remove directed by my field repro- (2) Complete all roug: Sheet 5 referenced above. (2) Complete all roug: Sheet 5 referenced above. (2) Complete all roug: Sheet 5 referenced above. (3) Signature of person exthemated to a	Isc DATE SIGNED	is document and return chan headings, including i equipment to per international for a cordar iated 30 May 191 ing in the based itional partition in item 9A or 19A, as her 16A. NAME AND TITLE Major, Comps 16B. united States ave.	n <u>one</u> <u>copy</u> solicitation/contract in Sect matter where families reform the specific items of nce with Sheets 1 thru 7 pro- 85, as hereafter listed: ment not suitable for reuse, ons in the basement as shown (continued) stofore changed, remains unchanged and in full force E OF CONTRACTING OFFICER (Type or print) Stofore changed, remains unchanged and in full force E OF CONTRACTING OFFICER (Type or print) Stofore changed, remains unchanged and in full force E OF CONTRACTING OFFICER (Type or print) Stofore changed, remains unchanged and in full force E OF CONTRACTING OFFICER (Type or print) Stofore changed, remains unchanged and in full force E OF CONTRACTING OFFICER (Type or print) Stofore changed, remains unchanged and in full force of Engineers Stofore Contracting Officer)
IMPORTANT Contractor soo DESCRIPTION OF AMENDMENT/MOOT a. <u>Change</u> : Provide all : k, in the basement and oth ed and revised by Request (1) Demolition; Remove directed by my field repro (2) Complete all roug: Sheet 5 referenced above.	requested r	is document and return chan neadings, including i equipment to per international part of the international part of the international part of the Major, Comps issues of the same issues of the Major, Comps issues of the international part of the	<u>one</u> <u>Copy</u> <u>social</u> to the issuing office <u>social</u> to be example to the issuing office <u>social</u> to be example to be example to be <u>social</u> to be example to be example to be example to be <u>social</u> to be example to be example to be example to be <u>social</u> to be example

Six causes of mods were identified. Each item of change was classified under one of these six categories based on a subjective evaluation of the nature of each change item using the criteria defined below. These groupings evolved as each mod was reviewed: they were not arbitrarily identified beforehand. The six sources of mods identified in this study were:

- 1. Design deficiency.
- 2. Unknown site conditions.
- 3. User requested changes.
- 4. "No fault" design changes.
- 5. Value engineering.
- 6. Other.

Items were classified as "design deficiencies" if it were clear the change was due to faulty design or other errors/omissions on the part of the designer. For those instances where it was not clear if the change was due to a failure on the part of the designer, the benefit of the doubt went to the designer and the item was placed under Consequently, this category was one of the other categories. deliberately designed to err on the side of conservatism. "Unknown site conditions" was used for changes resulting from site conditions encountered during construction different than those expected. This included underground conditions such as unrecorded utility lines, soil conditions, or bedrock. It also included situations where improvements to existing buildings had been made over the years and not properly documented. This was a particular problem for the projects of this study since many involved renovation of existing buildings, some of

which were almost 100 years old. "User requested changes" were those changes initiated by the individual or agency that would ultimately use the facility. "No fault design changes" included changes to building codes, changes to military regulations which impacted on design, or other situations where events outside the control of the District, contractor, or user. "Value engineering" changes were those resulting from the Corps of Engineers value engineering program. Under this program, a contractor can share in the savings the government realizes if the contractor discovers a less expensive method of completing the work designed. The "other" category was used for time extensions due to poor weather, administrative changes to contract wording, problems with site accessibility, problems caused by user interference with work crews, or other items occurring too rarely to qualify for a category of their own and not readily classified under one of the other headings.

"Design deficiencies" were further analyzed to determine in which engineering design discipline deficiencies occurred. All changes due to design deficiencies were classified as being "architectural", "civil", "mechanical", "electrical", "structural", or "administrative". Thus, a basis for further examination of the problem of design deficiencies was established. This technique also provides a basis for the focus of design review effort.

Analyzing the causes of change was done in two ways. The first analysis was by "items of change". The purpose of this "item of change" analysis was to identify the total number of changes and their causes. Every mod may consist of one or more individual changes to the contract. Multiple changes are often combined under one contract

4 ŷ

modification to expedite administrative requirements. Changes within the same mod may have different sources. For example, a modification containing seven changes may have three due to design deficiencies, two due to site conditions, and two due to user requests. All change items were treated equally, regardless of cost. For example, a design deficiency costing \$10,000 was considered one item of change; a no-cost administrative change in contract wording was also considered one item of change. This analysis resulted in an accurate assessment of the causes of contract changes to the projects studied.

The second analysis was by cost. The purpose of this analysis was to determine which modification sources cost the government the greatest amount of money and to determine the overall direct costs associated with mods. Only direct costs, as charged by the contractor, were considered in this analysis. Recall that ENG Form 3938 only contains the total mod cost: it does not record the cost of each item of change. Consequently, the cost per change for mods containing multiple items of change had to be obtained from contractor proposals (Figure 4). These proposals were submitted by the contractor when the need for a change was first identified. This proposal was then negotiated bilaterally with the government's representative and a final mod price established. The corrected estimate was filed with the contract documents. This combination of cost and "item of change" analyses provided a comprehensive evaluation of contract mod causes.

ISSUING OFFICE		9. CONTRACT	MUDIFICATION NO
			SX001
CONTRACT	OR'S PROPOSAL	CHANGE IN CONTRACT PRIC	E
NOTE: SIGN ANI	D RETURN ORIGINAL ANI	COPIES: RETAIN ONE COPY FOR YOUR	1112
1 226 761 78	NET DECREASE	CALENDAR D	268
PROPOSED CHANGE BLDG	66	PROPOSED CHANGE BI	DG. 67
Material:		Material:	
Sheet Metal	\$ 586.74	Sheet Metal	\$ 551.36*
Insulation	304.59	Insulation	228.38*
Venting	19.59	Venting	240.44
Relocate Flue	19.43	Register - Grilles	216.00
Registers - Grilles	277.00	Electrical	29.55
Electrical	9.69	87% Furnace	60.00
87% Furnace	60.00	Water Piping	8.04
2 x 4 Bracing	18.00	Material Total	\$1,333.77
Water Piping	8.04		
Material Total	\$1,363.08		
Labor:		Labor:	
Sheet Metal Worker		Sheet Metal Worker	
127 Hrs. @ \$11.00/Hr.	\$1,452.00	59 Hrs. @ \$11.00/H	lr. \$ 649.00
Insulation Installer		Insulation Installer	
62 Hrs. 3 \$14.00/Hr.	868.00	44 Hrs. @ \$14.00/H	lr. 616.00
Pipe Fitter		Pipe Fitter	
4 Hrs. @ \$12.11/Hr.	48.44	1 Hr. @ \$12.11/Hr.	12.11
Electrician		Electrician	
5 Hrs. @ \$9.06/Hr.	45.30	2 Hrs. @ \$9.06/Hr.	18.12
Painter		Painter	
2 Hrs. @ \$13.54/Hr.	27.18	2 Hrs. @ \$13.59/Hr	. 27.18
Carpenter		Labor Total	\$1,322.41
3 Hrs. @ \$11.53/Hr.	34.59		
Labor Total	\$2,420.51		
avroll & Benefits Burden:		Payroll & Benefits Burd	en:
38.7%	\$ 936.73	38.7%	<u>\$ 511.77</u>
SUBTOTAL	\$4,720.32	SUBTOTAL	\$2,661.29
Overhead:		Overhead:	
15.26%	\$ 720.32	15.26%	\$ 406.11
SUBTOTAL	\$5,440.64	SUBTOTAL	\$3,067.40
Profit:		Profit:	
102	\$ 544,06	10%	\$ 306.74
TOTAL COST PER UNIT	\$6-, 330-81	TOTAL COST PER UN	IT \$3,374.14
TOTAL COST 22 UNITS	\$131,633.40	TOTAL COST 30 UNITS	\$101,224.20
Forderstein Taban	97 11- \$1	3 75/₩+ \$1 196 75	
Dayroll Burdon		۲., ۲۶, ۱۲۰ ۸۵۶ ۸۸	
rayioli butuen	C1/%	ATOTAL SI 659 19	-
Overband	15 26 7	253 10	
Overnead	TO'	TAL \$1,912.38	
IE ITPED NAME AND TITLE		SIGNATURE	
5/8/85	V.P. Project Ma	inager	<u></u>
		-	

ENG 10M 3938

PAGE 2 OF 2 PAGES

FIGURE 4 Sample Contractor's Proposal

The main statistic used to measure mod cost in this study was project cost growth. There is, however, no single method of calculating cost growth within the Corps of Engineers. For the purposes of this study, cost growth was calculated from direct mod costs. Cost growth was expressed as a percentage and reflected how much a project exceeded the original contract amount. This was calculated by first adjusting the contract award amount in the following manner:

Adjusted Contract Amount = (Award Amount) + (Supplemental Agreements) - (Work Deletions or Terminations)

Supplemental agreements, work deletions, and terminations all reflect changes to project scope. In several cases, these represented extremely large values. Consequently, failure to adjust the original contract amount to allow for scope changes resulted in an inaccurate cost growth figure. After calculating the adjusted contract amount, cost growth was found in the following manner:

Cost Growth = (Net Mod Cost) / (Adjusted Contract Amount)

The use of cost growth figures provided a convenient, standard basis of comparison between different projects.

Projects were also analyzed on the basis of project size, project type, agency responsible for design, and predesign reviews to

determine if any correlation existed between these factors and mod sources or costs. Project sizes were classified based on adjusted contract amount. The classifications used were "less than \$500,000", *\$500,000-\$1,000,000", and "over \$1,000,000". These size intervals were selected to match the intervals used in other studies to facilitate comparison of findings. Classification by project type involved grouping projects by funding source: Operations and Maintenance, Army (OMA), Army Family Housing (AFH), and Military Construction, Army (MCA). Classification by designer involved identifying which projects were designed by contracted, private A/E firms and which were designed by in~house, government employed engineers. These government engineers worked for either the District or the installation's Directorate of Engineering and Housing (DEH). Each of these agencies had the option of designing projects using their own in-house resources or contracting the design out to civilian architecture or engineering (A/E) firms. Each of these three items was very objective in nature and allowed each project to be easily classified.

Classification under the fourth factor, predesign review, was more subjective. The Corps recognizes five types of predesign reviews. These are:

> Constructibility Reviews. Conducted to insure a project can be built as designed.

Technical Reviews. Conducted to insure design meets all applicable codes and technical specifications.

Using Service Reviews. Conducted by the user in the early stages of design to insure the design meets the needs of those who are to use the finished facility. 44

Biddability Reviews. Conducted to insure sufficient information is provided to the contractor to allow a bid to be submitted.

Operability Reviews. Conducted to insure that, once the project is completed, the design allows for ease in maintainability and operability.

)

Of these, only the first two were considered in detail for this study for the following reasons. Using service reviews were not evaluated because those reviews are normally conducted by the installation DEH. Since the District has minimum control over these reviews, they were not included in this analysis. Second, the supporting District did not conduct separate biddability and operability reviews at the time this study was conducted. Those items pertaining to operability and biddability were checked as part of either the constructibility or technical review. Consequently, the only reviews evaluated were technical and constructibility reviews conducted within the District.

One study objective was to determine if failure to conduct any or all of the first three reviews listed above resulted in a noticeable change in project cost. Since records on reviews were not routinely kept once a project was complete, this was accomplished by interviews with individuals familiar with each project in the District or DEH. There were often differing perspectives on what reviews took place and on the detail with which it was conducted. For the purposes of this study, if anyone in any of the contacted agencies attested to a review having taken place for a particular project, that project was classified as having been given that review. Reviews were considered on a "yes" or "no" basis only; no attempt was made at this point to establish the quality or depth of the review conducted. Evaluation of the review process was done in a separate phase of the study and will be discussed in the following section.

Modification costs may be measured by ways other than just direct cost. Negotiation, design, administration, and processing activities all consume manhours the government must pay for. This study was to quantify the dollar amount of this overhead and to determine how the effort expended on mods impacted on overall productivity. In this District, the personnel responsible for mods were the same people responsible for new project design, construction supervision, and quality assurance. Consequently, any time spent on modifications is time not spent on these other tasks. Therefore, if the time spent on mods could be reduced, overall District productivity would be increased.

The District maintained a cost account of engineering, design, and overhead for modifications to each project beginning at the time the contract was awarded. Cost amounts were available from periodic reports prepared and maintained in the contract administration branch within Construction Division. Costs for engineering and design were maintained separately from overhead. However, these amounts related only to costs incurred by Engineering Division. They did not reflect the costs incurred by Construction Division in negotiations, site investigations, administrative processing, etc. Such costs were accounted for under a general "construction" account which combined all project construction related costs together. Individual modification costs were included in this cummulative total and could not therefore

be established. Consequently, the amounts presented for overhead are conservative in that they reflect only Engineering Division's administrative costs. They do not include costs credited to the construction account.

Within the District, engineering, design, and overhead costs were accounted for by project, not by individual modification. The amounts were analyzed by taking project overhead figures and determining the average cost per mod and the average cost per item of change. These figures were then statistically analyzed to determine the mean, mode, and range on both a "per mod" and "per item of change" basis.

The impact of mod processing on productivity was determined by a one of the questions in the questionnaire prepared to evaluate the desgin review system. The questionnaire is discussed in detail in the following section. This particular question listed the activities, to include modifications, that each individual could be expected to be involved in on any given day. The question asked the respondent to identify how much time during a typical week was spent on each activity. Responses were set on a multiple choice format, with each choice representing a range of percentages. (See question #9, Appendix 6.) The intervals were arbitrarily selected at 0%, 1%-5%, 6%-10%, 11%-20%, 21%-30%, 31%-50%, and over 50\%. The resulting information was analyzed by taking all responses from a particular office and converting the individual responses to hours out of a 40 hour week. For example, if one individual answered that he spent 6%-10% of his time on mods, that was converted to a window of 2.4 - 4.0 hours per 40

hour week. The lower and upper figures of this window were summed for all respondents from the same office. These summations were converted back into percentages by dividing each by the total number of available manhours. The following example illustrates how this was done for a hypothetical office with four people providing responses:

TABLE 3 Time Spent on Modifications (Example)

	Answer Given	Range i	n Hours	Total Hours
Respondent	on_Questionnaire_	Lower	Upper	<u>in Week</u>
#1	6%-10%	2.4	4.0	4 Ŏ
#2	11%-20%	4.4	8.0	40
#3	1%- 5%	0.4	2.0	40
#4	6%-10%	2.4	4.0	40
	Sums:	9.6	18.0	160

By dividing the summed lower (9.6) and upper (18.0) ranges of hours each by the total available manhours (160), it is estimated that this office spends between 6% and 11% of its available manhours on modifications. This calculation was performed individually for each office involved with mod processing or design and gave an indication of the total time modifications consumed within the District.

All computations were performed on a Texas Instruments Professional Computer using Lotus 1-2-3 spreadsheet software. Tables of data are inclosed in the appendices. Lotus 1-2-3 was also used to develop graphs and charts for data presentation. These figures are presented, where appropriate, in Chapters 4 and 5.

In summary, data on mod causes, costs, and sources of design deficiencies were collected from primary source documents, including

construction files, cost reports, a questionnaire, and individual discussions. Mod costs were analyzed from the perspectives of direct dollar cost, indirect cost, and the diversion of manhours away from other essential tasks. This methodology, consisting as it did of analysis by both cost and items of change, provided a comprehensive assessment of the cost and causes of contract changes.

Methodology for Evaluating the Design Review System:

The data collected under the foregoing methodology led to the conclusion that the primary causes of modifications on the projects studied were design deficiencies, user requested changes, and unknown site conditions. The discovery of this fact led to an analysis of the design review system used by the supporting District. Detailed, thorough design reviews are a major factor in insuring design problems are identified and corrected before construction begins. The fact that so many design deficiencies were identified indicated possible problems with this system. Consequently, it was decided to evaluate the review system to determine where those problems might lie.

Information for the evaluation was collected by questionnaire. The purpose of the questionnaire was to obtain data from which to evaluate the preconstruction review process within the District. The objective was to solicit the opinions of those personnel involved in the review system, both at the supervisory and "worker" levels.

The evaluation was based on an analysis of three primary components of the review system: personnel qualifications, time available, and system organization and procedures. Specific items of interest were identified within each of these components. Opinions on each item were obtained by individual questions. A total of 43 questions were asked. Most items were covered by multiple questions asking for the same information but in a slightly different manner. This redundancy was used to insure a reliable representation of the respondents opinion on each item was obtained. Redundant questions were separated within the questionnaire. A summary of the items within each component and the specific questions addressing each item is shown in Table 4. A copy of the questionnaire, with a summary of responses, is at Appendix 6.

The questionnaire was validated by distributing six copies to District employees to insure questions were clearly worded and understood by recipients. The questionnaire was then modified and distributed to a representative sample selected from those offices within the District involved with the conduct of military construction reviews. 71 questionnaires were originally distributed. Of these, 15 the respondents were subsequently found to have no direct οf involvement with reviews. Their questionnaires were discarded. Representation from Engineering and Construction Divisions as contained in the remaining 56 questionnaires was as shown in Table 5. Table c presents a further breakdown of the respondents from Engineering Population statistics were taken from the District's Division. organization chart. Population figures included all authorized engineers, architects, and other professional personnel (such as geologists) and engineer technicians. Temporary hire, draftsmen, and clerical personnel were not included.

				Specific Que	stions
Campan	ont	Items of	Interest	Item of Int	erest
Personne	2				
Qualit	ications	Educatio	n	#4 0	
		Training	/Expertise	#4,5,22	
		Practica	1 Experience	# 35 ,36,3 7	,38,39
		Professi	onal Licensing	#41,42,43	
		Attitude	Toward Review	s #16,17,26	
Time Ava	nlable	Priority	of Work	#10,15,27	
		Time Req	uired vs		
		Time	Available	#6,7,8,14	
System (rganization				
and Pr	ocedures	Site Vis	its	#11,12,28	,29
		Incorpor	ation of		
		Сомм	ents into Desi	gn #18,19,31	
		Review C	hecklists	#21,33	
		Review Q	uality	#34	
		Continui	ty, Feedback		
		and	Flow of Info	#25,30,32	
		Frequenc	y of Revie⊭s	# 23 ,24	
		Estimate	Accuracy	#20 (Not	e)
		Familiar	ity with		
		Regu	lations	#13 (Not	e)
Admin Da	ta	Branch a	nd Uffice	#1	
Note: A 1	nalysis showe 3 and 20, Co	ed problem onsequentl TA	s in the inter y, the respons BLE 5	pretation of q es were not ev	uestions aluated.
Sum	mary of Surve	ey Populat	ion Sample - B	ath Divisions	
Eng	ineering Div	15100		Construction	Division
	f	roject	Constr		
Ē	ngr_DivN	lanagers	Rees	<u>Rsdnt_Off</u>	
Sample:	29	10	6	5	6
Population:	88	23	10	9	7
% of Pop:	27% (42%)	43%	60%	56%	86%

TABLE 4 Review System Components of Evaluation

Branch and Section	Gample	Population	Sample as % of Population
SECCION	-34#67E		
Design Branch:			
Mech/Elec	7	15	47%
Gen Struct	4	10	40%
Arch	7	9	78%
Est/Specs	6	19	32%
Res Struct	3	11	27%
Fndtns/Mtrls Branch	:		
Local Protection	1	12	8%
<u>Geology</u>			<u>8%</u>
Totals:	29	88	27%
Less Fndns/Mtrl B	r: 27	64	42%

TABLE 6 Representation From Engineering Division

Responses were analyzed by response group. The purpose of classifying responses by response group was to determine differences in perceptions of the various issues being examined. Five response groups were identified; two from Engineering Division and three from Construction Division. Engineering groups consisted of project managers and engineers in other branches involved in reviews. Construction Division groups included construction field representatives working on installation projects, personnel from quality assurance branch, and office engineers and technicians from the resident office.

A respondent profile by job area is presented in Table 7. Slightly over 50% came from Engineering Division, the agency responsible for design and technical review. 30% were construction representatives, resident office personnel, and personnel from quality assurance branch. These individuals all fall under the Construction Division, and share responsibility for constructibility reviews. Project managers comprised the remaining 18%.

> TABLE 7 Respondent Profile (Question 1)

Office,	Branch	Percentage
or_Job_Des	cription	Respondents

Engineering Division	51.8%
Project Managers	17.9%
Quality Assurance	10.7%
Construction Representatives	10.7%
Resident Office	8.9%

Responses were tabulated manually. The tabulated data was then entered into a TI professional computer (PC) using Lotus 1-2-3 spreadsheet software. Tables, bar graphs, and pie charts were used to summarize the findings. These rindings are discussed in detail in Chapter 5.

Methodology for Comparing Findings with Other Studies.

The purpose of identifying other studies was to determine if the findings of this study were unique to the installation or District from which collected or universally applicable to the Corps of Engineers as a whole. Chapter 2 discussed the secondary sources of previously completed papers reviewed for the purpose of making that comparison. The comparison was made by identifying the key issues of this study and comparing the findings pertaining to those issues with the findings of the other studies. General trends were thereby established where possible for the three other segments of the study.
This analysis, by its nature, was more descriptive/qualitative than quantitative. From this analysis, it could be determined which findings appeared to be unique and which appeared to be part of the overall nature of military construction. These comparisons are discussed in detail in Chapter 7.

Summary.

This study consisted of three distinct phases, each of which required a different methodology. The first part, Causes and Costs of Modifications, required examination of project source documents to obtain data on individual items of change, reasons for each change, and the cost of each change. Additional data was obtained from management reports on indirect overhead costs. The second part, Evaluation of the Design Review System, involved collecting by questionnaire and analyzing opinions on specific issues of those individuals involved with the review system. Lastly, a descriptive method was used to compare the findings of this study to those of previous studies. The use of all of these methods provided a detailed inquiry into the overall modification picture as it applies to military construction.

CHAPTER 4

CAUSES AND COSTS OF CONTRACT MODIFICATIONS

General:

This chapter identifies the causes and costs of modifications to the 25 projects studied. It addresses the following research objectives:

- identify the causes of modifications to military construction projects.
- determine direct modification costs for the projects studied.
- determine the indirect costs associated with modification processing.
- identify the level of cost growth for DNA, MCA, and AFH projects.

Modifications were classified as being caused by design deficiencies, unknown site conditions, user requests, "no-fault" design changes, value engineering proposals, or "other". The "other" category included no cost administrative changes to contract wording, time extensions due to bad weather, occupant interference with contractors, and additional items not covered under the first five classifications. Items of change within each modification were analyzed and classified individually. The cost per item of change was determined from Eng Form 3938 and written contractor proposals. This approach allowed a thorough analysis of causes due to change items as well as the cost of those changes.

The remainder of this chapter is organized in the following manner. The causes of modifications for projects studied are presented in the next section, along with a discussion of design disciplines contributing most to design deficiencies. Modification cost and project cost growth are discussed next. Subsequent sections correlate cost growth with preconstruction reviews, project type, project size, and responsible design agency. The final section is a chapter summary.

Modification Causes.

Appendix 2 contains tables showing the causes of modifications by cost and by item of change, per project. Figure 5 summarizes the data from the cost tables. The primary sources of mod cost was found to be design deficiencies (36.3%), user requested changes (22.3%), and unknown site conditions (21.8%).

Figure 6 summarizes the results of the analysis by items of change. Design deficiencies were found to make up 33.2% of all items of change, followed by unknown site conditions (27.4%) and user requested changes (21.7%). Recall from the previous chapter that the percentages shown in both figures for design deficiencies are conservative.

A comparison of mod causes by cost and change item is shown in Table 8. Design deficiencies, unknown site conditions, and user requested changes are the three main causes of mods in both categories; together they make up 80% of mod costs and 83% of all items of change. The significance of this is that these mod causes are those which should be most influenced by thoro&oh preconstruction reviews.



Table B Comparison of Modification Sources Cost versus Items of Change

	Percent	Percent
Cause of	by Item	of Total
<u>Modification</u>	<u>of Change</u>	Mod_Cost
Design Deficiency	33.2%	36.3%
User Requested Changes	21.7%	22.3%
Unknown Site Conditions	27.4%	21.8%
Design Changes	8.4%	6.2%
Value Engineering	1.0%	0.5%
Other	8.2%	12.8%

Causes of Design Deficiencies:

From the analysis done to determine the design disciplines in which design deficiencies occur, architectural items were found to have made up the largest share of design deficiencies. These findings are tabulated at Appendix 3 and summarized in Table 9. This analysis was also done on the basis of both items of change and cost. The figures show architectural items accounting for 41.7% of design deficiency items of change and 57.5% of design deficiency costs. Typical architectural modification items included location of non-load bearing walls, errors in door and window schedules, changes to finishes. improperly specified hardware, casements, cabinets, and improperly sized door frames. Deficiencies in mechanical design comprise the next largest portion, accounting for 23.6% of cost and 25.4% of change items. Typical mechanical problems included conflicts with ductwork routing and structural elements, HVAC design, and improperly designed/routed interior plumbing systems. Electrical deficiencies

make up 12.5% of change items but only 5.9% of cost. Structural, civil, and administrative changes together account for the remaining 20.4% of change items and 13.0% of cost. It is concluded from this that had reviews placed greater emphasis on the architectural and mechanical aspects of design significant cost growth savings may have been realized.

Design	Percentage by	Percentage by
Discipline	Cost	<u>Items of Change</u>
Architectural	57.5%	41.7%
Mechanical	23.6%	25.4%
Electrical	5.9%	12.5%
Structural	5.9%	10.4%
Civil	7.1%	5.0%
Admin Changes	.	5.0%

Table 9 Disciplines in Which Design Deficiencies Occur

Modification_Costs.

Costs were measured in three ways. The first was direct cost. This is the amount the government pays to the contractor as compensation for work performed. It includes labor, materials, profit, and impact items. The second measurement was indirect costs incurred by the district, which include overhead for modification design, processing, negotiations, and other administrative tasks consume employee time accounted for under various cost codes. The third method evaluated the effort expended in mod processing because time spent on modifications is time not available for construction inspection, design, reviews, and other tasks. Thus, modification processing has a direct effect on district productivity.

Using these three cost measurement techniques provided a comprehensive assessment of overall modification cost. There are, however, two additional, intangible costs. These are the loss of customer satisfaction and damage to the Corps' professional reputation. These nonquantifiable costs will be discussed further in Chapter 6.

The 25 projects studied represent total contract amounts of almost 26 million dollars. Total direct modification costs amounted to almost 2.4 million dollars, for an overall cost growth of 9.4%.

A frequency distribution of the number of individual projects falling within various cost growth brackets is shown in Figure 7. Recall that in Chapter 1, "acceptable" levels of cost growth were assumed depending on project scope. These levels were 3% for new, "from the ground up" projects, 5% for new work in existing buildings or non-complex renovation work, and 10% for renovation work involving complex mechanical or electrical work in older buildings. Figure 7 shows that 12 out of 24 projects (50%) exceeded these acceptable cost growth levels. This represents more than an expense to the taxpayer. Excessive cost growth reduces customer satisfaction, degrades the Corps' professional reputation, and decreases district productivity by diverting manhours from design, reviews, inspections, and other activities.

Controlling direct cost can result in substantial savings. Table 10 illustrates how much may have been saved if cost growth on

those projects with excessive costs had been held to acceptable levels. In several cases, even though the savings represents a reduction of only two or three percentage points, the dollar cost savings is significant. The total savings amounts to approximately \$380,000. If the high cost items previously deleted from calculation were included. the total savings rises to over \$1,250,000.



Figure 7 Cost Growth Frequency Distribution

District indirect cost for engineering, design, and overhead amounted to \$129,000 #117,800 of this wis spent on modification design. If an hourly rate of \$20 is assumpt, this amounts to 5690 manhours (2.9 man-years) of a GS-11 engineer's time. This represents a significant drain on overall design productivity. This figure, however, does not represent total indirect cost: All time spent by

personnel in the district's Construction Division were accounted to a general "construction" account and was impossible to trace to individual projects. Consequently, actual indirect costs are much higher. A summary of engineering, design, and overhead costs is at Appendix 5.

Table 10 Potential Savings From Control of Direct Cost Growth

PROJECT	ADJ CONTR	ACTUAL COST	ACCEPTABLE	POTENTIAL
NUMBER	AMOUNT	GROWIH	COST_GROWTH	SAVINGS
1	\$ 796,000	\$140,390 (17.6%)	\$ 79,600 (10%)	\$ 60,790
2	\$ 83,000	\$ 10,467 (12.6%)	\$ 4,150 (5%)	\$ 6,317
2	\$4,637,000	\$235,468 (5.1%)	\$139,110 (3%)	\$ 96,358
6	\$ 224,576	\$ 24,115 (10.7%)	\$ 11,229 (5%)	\$ 12,886
10	\$ 79,191	\$ 5,814 (7.3%)	\$ 3,960 (5%)	\$ 1,854
11	\$ 488,970	\$ 55,632 (11.4%)	\$ 24,449 (5%)	\$ 31,184
13	\$ 209,603	\$ 27,539 (13.1%)	\$ 10,480 (5%)	\$ 17,058
14	\$ 860,518	\$161,420 (18.8%)	\$ 86,052 (10%)	\$ 75,368
16	\$ 207 ,500	\$ 25,145 (12.1%)	\$ 10,375 (5%)	\$ 14,770
17	\$ 411,373	\$ 52,320 (11,8%)	\$ 20,569 (5%)	\$ 31,751
20	\$ 267,000	\$ 19,367 (7.3%)	\$ 13,350 (5%)	\$ 6,017
23	\$ 505,742	\$ 53,098 (10.5%)	\$ 25,287 (5%)	\$ 27,811
			:	
			TOTAL:	\$382,164

IF HIGH DOLLAR MODS AND PROJECT 15 ADDED:

5	\$4,666,590	\$411,844	(8.8%) 4	233,330	(5%)	\$178,515
15	\$ 536,300	\$477,408	(89.0%) 4	53,630	(10%)	\$423,778
22	\$1,247,433	\$331,873	(26.6%)	62,372	(5%)	\$269,501
					=	
				TOTAL		\$871,794
			GRANE) TOTAL	\$	1,253,958

, ---, ---

An attempt was made to correlate indirect costs with other variables such as project size, design agency, and project type. No correlation with any of these variables was found. Engineering,

design, and overhead costs per item of change averaged \$165; costs per modification averaged \$483. Figure 8 is a frequency distribution showing the number of items of change falling into each of six cost brackets. The graph shows how many projects had an average indirect cost per item of change falling between the brackets shown. The mode was for projects to have their average cost per item of change for E&D and overhead to be less than \$100. There is, however, no consistent pattern for items outside this range. Figure 9 is a similar distribution but presents the cost per modification. Again, the vertical axis indicates how many projects had an average cost per mod falling within the designated brackets. As can be seen, no clear pattern is discernible. The conclusion drawn from this is that engineering design and overhead costs appear to be a function of mod design complexity, which varies with each modification. Hence. generalizations concerning relationships between these costs and other factors cannot be made.

Modification cost may be measured in terms other than dollars. Mods consume numerous manhours of personnel (other than design) involved in mod processing. An estimate of how much time was involved was made by asking those involved in mod processing to identify what percentage of a typical week was spent on various activities, to include modification processing. Within the supporting District, the three offices showing the most effort expended on mods were the resident office (21% to 35%); design branch (6% to 13%); and quality assurance branch (12% to 20%). The time consumed by mod processing in the resident office is often cited as the reasons other essential work.

such as constructibility reviews, is not done in the detailed manner that it should be. While the ranges for design branch and quality assurance are probably not excessive considering overall work load, they do represent a significant amount of time that could more productively be spent on design or design review.







Figure 9 Frequency Distribution of Engineering, Design, and Overhead Cost per Modification

Cost Correlation by Design Review:

This section analyzes the projects on the basis of preconstruction reviews. Three types of reviews were considered: constructability, technical, and using service. Personnel familiar with each project were contacted in the supported installation Directorate of Engineering and Housing (DEH), the local Corps of Engineers resident office, and the supporting District headquarters to identify which projects received which reviews. There were often differing perspectives on what reviews took place for a given project between these agencies. For the purposes of this study, if anyone in any of the contacted agencies attested to a review having taken place for a particular project, that project was classified as having been given that review. Reviews were considered on a "yes or no" basis only. No attempt was made, at this point in the study, to establish the quality of the review conducted.

Table 33 of Appendix 4 indicates 21 projects were reviewed for constructibility, technical sufficiency, and using service requirements. Four projects (including Project 15) are shown as having one or more of these reviews missing. Projects which received all three reviews show mod costs amounting to 4.7% of their summed contract bid amounts. Projects not receiving all reviews show a cumulative cost growth of 35.6%. When adjusted by deleting Project 15, this percentage drops to 15.4%. Thus, it may be concluded that projects which are not carefully reviewed for technical sufficiency, constructability, and user serviceability tend to result in higher overall cost growth.

The histograms at Figures 10 and 11 provide a frequency distribution of cost growth for projects receiving and not receiving all three reviews. Figure 10 shows that the four projects not receiving all three reviews exceeded acceptable cost growth levels. Figure 11 presents a frequency distribution for projects having received all three reviews. Although the overall cost growth for reviewed projects was lower than the cost growth for non-reviewed projects, nine of the 21 reviewed projects still had unacceptably high levels of cost growth. This may indicate that preconstruction reviews were not conducted with the same degree of thoroughness for all projects, and did not consistently result in precluding problems before construction began.



Table 11 presents a comparison of the causes of modifications as a function of review. Projects receiving all three reviews show a significantly lower percentage of cost growth (33.7%) due to design deficiencies that do non-reviewed projects (50.2%). There is, however, only a modest decrease in problems with unknown site conditions (28.7% to 20.5%). Consequently, it may be concluded that reviews, as currently conducted, do not adequately reduce the effects of cost growth due to problems with site conditions and user requested changes. The problem with site conditions was probably due to a lack of site visits conducted as part of the design and review process, while poor project scope definition was a major contributor to user requested modifications. This appears to be most significant with loosely designed "year-end-crunch" projects (such as Project 15).

Table 11 Causes of Modifications Correlated by Review

Percentage of Mod Cost

	Projects	Projects
Modification	Receiving all	Not Receiving
Causes	Three_Reviews	all_Three_Reviews
Design Deficiency	33.7%	50.2%
User Requests	15.8%	4.5%
Site Conditions	20.5%	28.7%
Design Changes	6.5%	4.5%
Value Engineering	0.6%	0.2%
Other	13.0%	12.0%

NOTE: The quality of the review conducted was not considered at this point in the study.

Cost_Correlation_by_Project_Type:

The 25 projects studied contained three different types of projects. These were Army Family Housing (AFH); Military Construction, Army (MCA); and Operation and Maintenance, Army (OMA). Five of the projects studied were AFH, five were MCA, and 15 were OMA. One of the five NCA projects is minor MCA which, for the purposes of this study, will be treated as MCA. Table 34 of Appendix 4 contains data for projects categorized under each project type.

Cost growth averaged 9.4% for AFH projects; 4.2% for MCA projects; and 6.7% for DMA projects. Figures 12, 13, and 14 illustrate how many of each type of project which experienced unacceptable levels of cost growth. Figure 12 shows data for AFH projects; note that four out of five (80%) of the AFH projects had excessive levels of cost growth. DMA data, shown at Figure 13, is similar; eight out of 15 projects (53%) had unacceptable cost growth levels. Conversely, the data for MCA projects shows only one out of five (20%) experiencing cost growth problems. In general, MCA projects tend to experience lower levels of cost growth than do AFH and OMA projects.

Two the five AFH projects (60%) did not receive all three preconstruction reviews. Of the three which did get fully reviewed, two had unacceptably high cost growth. Two of the 15 OMA projects did not receive all three reviews, and each of these two had cost growth of over 15%. Of the 13 OMA projects which received all three reviews, six still registered unacceptable levels of cost growth. All MCA projects received all three reviews. Only one MCA project exceeded its acceptable cost growth level, but only by two percentage points

(Project 3). This indicates that MCA projects were more carefully reviewed and have correspondingly lower cost growths than do OMA and AFH projects. The fact that six of the OMA projects and two of the AFH projects which received all three preconstruction reviews still had unacceptably high cost growth indicates that reviews for these types of projects tend to be less thorough than those given MCA projects. Furthermore, OMA and AFH projects are more susceptible to end-of-the-year funding pressures. Consequently, they are more likely to be released for bid with only cursory review.



Figure 12 Cost Growth by Project Type AFH Projects



Figure 14 Cost Growth by Project Type MCA Projects



The causes of modifications for each project type is shown in Table 12. Design deficiencies, site conditions, and user requested changes were the main causes of mods for each project type, although AFH projects appeared to be more subject to criteria design changes than did OMA or MCA projects. Note that although MCA projects tended to have lower modification costs than OMA and AFH projects, and although they appeared to be more carefully reviewed, they still had large numbers of modifications due to design deficiency that were obviously not being caught by the review process.

Table 12 Causes of Modifications Correlated by Project Type

Percentage of Mod Cost

modification			
Causes	OMA	AEH	MCA
Design Deficiency	40.2%	16.9%	40.3%
User Requests	18.6%	18.2%	25.7%
Site Conditions	24.5%	20.1%	21.3%
Design Changes	0.3%	16.9%	4.5%
Value Engineering	0.0%	0.0%	0.8%
Other	16.3%	21.9%	7.5%

Cost Correlation by Project Size:

M 1 1 **P** 1

Þ

An analysis by project size was conducted to determine the correlation, if any, between cost growth and adjusted contract amount. The 25 projects studied ranged from \$71,432 to \$4,666,590. Projects were divided into three categories based on contract amount. Thirteen projects had an adjusted contract amount of less than \$500,000; six between \$500,000 and \$1,000,000; and six over \$1,000,000. Table 35 of Appendix 4 presents the data for projects under each of these classifications.

Projects whose adjusted contract amount was less than \$500,000 had an overall cost growth of 7.7%; those between \$500,000 and \$1,000,000 were at 12.8%; and those over \$1,000,000 at 3.5%. Figures 15, 16, and 17 are histograms showing the cost growth of projects within each cost category. Figure 15 shows that eight of the 13 (61.5%) least expensive projects had cost growth exceeding acceptable limits. Figure 16 shows four of the six (66.7%) projects contracted for between \$500,000 and \$1,000,000 have excessive cost growth. In contrast, only one out of six (16.7%) of projects over \$1,000,000 exceed acceptable levels, and that one only exceeds the limit by two percentage points. This may indicate that more expensive projects tend to receive greater attention in controlling cost growth than do less expensive projects.

This conclusion is further supported by the number of projects within each category receiving all three preconstruction reviews. All projects contracted for over \$1,000,000 received all three reviews. However, two projects of less than \$500,000 and two between \$500,000 and \$1,000,000 did not receive all reviews. Additionally, the lower costing projects tended to be OMA and AFH projects. These budgets were controlled by the installation and therefore more likely to be caught in the crunch for end-of-year funds and therefore possibly reviewed in a less than thorough manner. A comparison between project type and adjusted contract amount is shown in Table 13.



Figure 16 Cost Growth by Project Size Projects Between \$500,000 and \$1,000,000





Table 13 Correlation Between Project Type and Adjusted Contract Amount

Adjusted Contract Amount (\$000)	Overall <u>Cost_Growth</u>	Number Projects OMA	Number Projects AFH	Number Projects MCA	
UNDER \$500	7.7%	11	2	0	
\$500 - \$1,000	12.8%	3	2	1	
DVER \$1,000	3.5%	1	1	4	

The data in this table indicates project size may be more of a determinate in cost growth than project type. This may be due to the fact that larger projects tended to be assigned the more experienced

project managers, are more thoroughly reviewed, and in general receive more intensive management than do smaller projects.

Table 14 illustrates the causes of modifications for each size category of project. The table indicates that design deficiencies, user requests, and site conditions are the main causes of mods for each size grouping of projects. Of significance is the "other" classification for the two lesser cost categories. Two major components of "other" costs in these projects were site accessibility and occupant interference. Many of the AFH and QMA projects were constructed under conditions of joint occupancy. In other words, the contractor and user both occupied the facility during construction. This in turn often led to conflicts resulting in the contractor claiming lost time and impact costs for schedule interruptions.

Table 14 Causes of Modifications Correlated by Project Size

Percentage of Mod Cost

Modification	Less Than	\$500,000 ~	Over
Causes	\$ <u>500,000</u>	\$1,000,000	\$1,000,000
Design Deficiency	53.4%	21.5%	40.5%
User Requests	19.3%	20.3%	24.6%
Site Conditions	8.9%	29.0%	21.1%
Design Changes	1.5%	12.2%	4.6%
Value Engineering	0.0%	0.1%	0.9%
Other	16.9%	17.0%	8.3%

Although the smaller projects had higher percentages of cost growth; their total dollar amount of mod costs is relatively small (Table 15). However, these projects were often built at a much higher level of visibility to the customer than are larger projects. High cost growth rates have an impact on customer satisfaction not reflected in the dollar cost, particularly those projects funded by the installation.

Table 15 Dollar Amount of Cost Growth By Project Size

Adjust	eđ			
Contrac	:t			
Amount	Ł	Overall	Number	Cost of
<u>(\$000</u>)]	_Cost_Growth	<u>of Proj</u>	<u>Modifications</u>
UNDER \$50)0	7.7%	13	\$227,038
\$500 - \$1,	000	12.8%	5	\$467,609
OVER \$1.	000	3.5%	6	\$647,486

Cost Correlation by Designer:

Military construction projects are designed by either in-house government employed engineers and architects or by privately owned firms contracted for that purpose. In-house designs may be by either DEH personnel at the installation or by the supporting district. 20 of the 25 projects studied were designed by contracted architectural/engineer (A/E) firms. Of the five in-house designed projects, three were done by the installation and two by the district. Table 36, Appendix 4, shows the projects designed by each agency.

Projects designed by in-house personnel had an overall cost growth of 2.8%. This is less than half of the 5.9% overall cost growth experienced by A/E designed projects. Figures 18 and 19 illustrate the number of projects designed by each source of design that experienced unacceptable levels of cost growth. Figure 19 is the data for A/E designed projects. Note that 11 of 19 (58%) of A/E projects had excessive levels of cost growth. In contrast, only one of five (20%) in-house designed projects had unacceptable levels of cost growth. This indicates that for the projects on the installation under study, in-house designers tended to do a better job of project design and/or review than did contracted A/E firms.





The causes of mods within each design grouping is shown in Table 16. Of significance is that 23.5% of the mod costs to projects designed by A/E firms was due to unknown site conditions compared to only 6.7% for in-house designs. This may have been a result of in-house designers having more ready access to construction sites during the design phase and being more familiar with project location than did contracted A/E firms. It could also have been a function of the A/E failing to visit the site, either due to lack of funding or for other reasons.



Table 16 Modification Causes Correlated by Design Agency

Percentage of Mod Cost

Modification	Designed	Designed by
Causes	In-House	Contract_A/E
Design Deficiency	41.3%	35.8%
User Requests	27.7%	21.7%
Site Conditions	6.7%	23.5%
Nesign Changes	5.6%	6.3%
Value Engineering	0.0%	0.6%
Other	18.7%	12.2%

AD-A172 833	THE CAUSES	AND COSTS	OF MODIFICA	TIONS TO MILITAR	2/3
UNCLASSIFIED	SIMPE COLL			E I HVIREN US JU	1575 NL
 	 				.)}.
				· · · ·	
				· · ·	
		· .			- <u> </u>
					- <u> </u>
					· · · · · · · · · · · · · · · · · · ·



ł

Þ

Summary:

This chapter discussed modification causes, cost growth, direct and indirect costs, and the impact of mod processing on productivity. It presented a correlation of cost and causes with preconstruction reviews, project type, project size, and design agency. An examination was made on the basis of items of change to each project contract as well as the cost of each item of change. Design disciplines contributing most to design deficiencies were identified. Design deficiencies, unknown site conditions, and user requested changes were identified as the major sourc of contract modifications.

The correlation between cost growth and review leads to the conclusion that modifications, by themselves, were not the real problem but rather a symptom of a greater problem. Obviously, perfect designs would not result in changes. Equally obvious is the fact that there is no such thing as a perfect design. Therefore, a systematic and thorough design review process is essential to keeping contract changes (and corresponding cost growth) under control. The fact that 50% of the projects studied exceeded acceptable levels of cost growth indicated a major problem with the review system in use. An evaluation of that system is the subject of the next chapter.

CHAPTER 5

THE PRECONSTRUCTION DESIGN REVIEW SYSTEM

<u>General:</u>

This part of the study was designed to evaluate the preconstruction review system used by the Corps of Engineer District which built the projects examined. Of the 21 projects that had been given using service, constructibility, and technical design reviews, nine (43%) still had unacceptably high levels of cost growth. Furthermore, it was found that 80% of all cost growth was due to design deficiencies, user requested changes, and unknown site conditions. Of all the reasons identified for cost growth, these three should be the most easily influenced by thorough preconstruction reviews.

The supporting District had design responsibility for 13 (52%) of the 25 projects. The remainder were either designed in-house by the installation DEH or contracted to A/E's by the DEH. In these cases, the district could only review projects after the completed design had been forwarded for construction. But of the 13 projects the district (or its contracted A/E's) did design, seven (54%) had excessive cost growth. This implied the District's review system was not fully effective in correcting problems prior to contract award. It was suggested in the last chapter that excessive mods and cost growth are not of themselves a problem, but rather a symptom of a larger problem. The objective here is to determine where within the review process that problem might be.

The primary source of data was a questionnaire collected from 56 district employees. Respondents included all construction representatives and office engineers located at the resident office, engineers working in Quality Assurance Branch of the Construction Division, engineers working in Design Branch and Foundations and Materials Branch of Engineering Division, and project managers. The findings which follow are a compilation of the subjective responses of these individuals to those questions. A copy of the questionnaire is at Appendix 6.

The study covered only those constructibility and technical reviews conducted by the district. Reviews by Division of District in-house designs, reviews contracted out to private A/E firms, and installation using service reviews were not addressed.

This was not intended to minimize the importance of using service reviews. Functional reviews by the installation are essential to reducing user requested modifications. In general, it appeared that poor project scope definition was a major contributor to user requested mods. Projects were designed and let for bid without a firm scope definition being communicated to the designer or user. Consequently, the designer may not have been aware of what the customer wanted and the customer not aware of what was designed until construction actually began. This problem was aggravated by personnel rotations at the installation which often resulted in the ultimate user being unfamiliar with design decisions made by his predecessor. The DEH must play the major role in addressing this issue. Improved coordination between the District and DEH to include functional reviews conducted early in the

design cycle and including designers, project managers, DEH, and the using agency would be a major step in resolving this problem. This area requires further study as to specific causes of user requested mods and solutions.

The chapter is organized to individually discuss each of the three main variables the questionnaire was designed to address. These variables were:

- 1. The training, experience, and motivation of review personnel.
- 2. The time available with which to do reviews.
- The organizational and procedural framework within which reviews are conducted.

The questionnaire was designed to obtain respondent perceptions and experiences in these areas. Each is discussed in the sections which follow.

Personnel Qualifications:

)

- -----

Questions in the questionnaire concerning personnel qualifications were designed to evaluate three areas:

- Education/training/professional qualifications of personnel conducting reviews. This includes academic degrees held, professional licensing, and training in how to conduct technical/constructibility reviews.
- Experience in engineering design and construction. This includes experience both inside and outside of government service.
- 3. Attitude and motivation toward the conduct of reviews.

The percentage of reviewers holding professional degrees was fairly high. 87.5% of all respondents held at least bachelor degrees

in a professional discipline. Those not holding degrees generally were construction representatives and engineer technicians working in Construction Division whose responsibilities depend more on experience than academic background.

The proportion of professionally registered personnel was also high. 38 of the 56 (68%) respondents held professional licenses or training certificates. A breakdown of professional registration by review category is at Figure 20. The largest number of those not holding professional licenses were construction representatives. The figure includes registration as a professional engineer, architect, landscape architect, or other professional field.



Figure 20 Professional Registration (Questions 41, 42, and 43)

Respondents were asked to identify training they had received, if any, in review techniques. The majority (55.0%) stated they learned how to do reviews through "experience". 33.9% indicated training in some kind of formal course. Responses of where this training was received included an engineering graphics course, training in the Navy, courses in value engineering, the Corps of Engineers Military Design Review Course (most common response), review seminars and in-house training, and courses taught at the United States Army Engineer School at Fort Belvoir. 11.2% stated they have learned how to do reviews through on-the-job training (OJT). Although a couple of respondents clearly identified a formal OJT program, it is not entirely clear if the others intended OJT to mean a formal, structured program or if this was another way of stating they learned through experience. Consequently, the percentages for "experience" and "on-the-job training" are probably not as clear-cut as shown and should be interpreted accordingly. In any event, it is clear that a standard base of training did not exist. Reviewers approached each review from This does not lend itself to the framework of past experience. providing a consistent review product due to the varied backgrounds of the individuals concerned.

Respondents were asked to identify their experience in the areas of design, construction, construction management, and total time in the Corps (either at District or DEH). Table 17 summarizes the responses given. Only five of the 56 respondents (three construction representatives and two quality assurance personnel) had less than two years with the Corps. The majority had over five years, with 60% of

the project managers, 59% of the design engineers, and 60% of the resident office personnel having over ten years total Corps experience.

Of the 29 respondents involved with technical reviews, only 3 (10%) had less than two years experience in design. All 11 individuals claiming over 20 years design experience were individuals responsible for technical reviews. Their experience with construction, however, was limited. 69% had less than two years; 86% less than five years construction experience.

Table 17 Professional Experience (Questions 35, 37, 38, and 39)

Number of Responses (Percentage)

Experience 0-5 yrs 5-10 yrs 10-20 yrs Over 20 yrs

Ιn	Corps	10	(18%)	16	(28%)	20	(36%)	10	(18%)
Аs	Designer	27	(48%)	10	(18%)	8	(14%)	11	(20%)
Ιn	Construction	37	(66%)	3	(5%)	6	(11%)	10	(18%)
As	Constr Mngr	36	(69%)	4	(8%)	8	(15%)	4	(8%)

NOTE- Only 52 respondents answered question on Construction Management experience.

Constructibility reviews required less experience in formal design and more in actual construction. 81% of the respondents responsible for constructibility reviews claimed over ten years experience in construction. However, just as technical reviewers had relatively little construction experience, those responsible for constructibility reviews had limited design experience. Only five (29%) indicated having over five years experience in design.

60% of project managers had between ten and 20 years in the Corps of Engineers, and 30% between two and five years. This time was spent between technical design and construction management type positions. They had virtually no experience in construction; 90% reported having less than two years construction experience. Most had none.

Respondents were asked to subjectively evaluate how often they were able to conduct thorough reviews within their areas of expertise. The results are shown at Figure 21. 45 of the 51 (88%) answering this question felt their experience qualified them to do good reviews "always" or "most of the time". They were then asked to identify those disciplines in which they felt qualified to do reviews (based either on education, training, or experience), and in which disciplines they were actually required to conduct reviews. The results are shown in Figure 22. In only a couple of instances were individuals required to conduct reviews outside their areas of expertise. This indicates that, except in rare cases, the qualifications of reviewers matched the work load requirement within the various design disciplines.

Attitude toward reviews was evaluated based on opinions expressed concerning the need for reviews and the perceived value of reviews in reducing construction costs. Figures 23 and 24 summarize the answers to these questions. The overwhelming majority of respondents felt reviews were necessary either "always" or "most of the time". 95% replied that reviews are necessary regardless of project size. 89% responded that reviews are needed regardless of who the project is designed by. As seen in Figure 24, 77% of the respondents indicated that reviews will reduce cost growth either "always" or "most of the time". It can be concluded that the greater majority of those involved feel reviews are important and do save the government money.


Figure 22 Review Qualifications vs Review Workload in Each Design Discipline (Question 4 and 5)





Figure 23 Opinions on How Often Reviews are Needed (Questions 16 and 17) (Based on Size of Project and Design Agency)

Figure 24 Opinions on How Often Reviews Save the Government Money (Question 26)



In summary, the data collected indicate those doing technical reviews had the requisite education and experience in engineering design, while those doing constructibility reviews had extensive experience in construction. Project managers, whose role was the coordination and management of reviews and review comments, tended to have less overall experience than reviewers. They were found to be weak in the area of construction, due mostly to the fact that project managers tend to be drawn from the ranks of designers. However, the experience they did possess in the areas of construction management was sufficient to meet their review coordination responsibilities. Consequently it was concluded that, although there was little cross training between disciplines, reviewers in general had the required education and experience to carry out review tasks within their respective areas. The only weak areas noted were in cross training between design and construction personnel and in the lack of a standardized training base.

Question 34 asked for an unqualified evaluation of the quality of reviews conducted by the respondents. 40% replied that reviews were only "sometimes" as thorough as the project warranted. This is in significant contrast to the confidence and qualifications discussed above. Therefore, there must have been other factors other than personnel qualifications affecting review quality. These other factors will be discussed in the subsequent sections.

Time_Available_for_Reviews:

The analysis of time was done by determining the time required

to do a review, the time needed to work reviews in with other job requirements, and the time actually provided. The questionnaire asked for review time as it related to small (less than \$500,000), medium (between \$500,000 and \$1,000,000) and large (greater than \$1,000,000) projects. Additionally, the timing of submittal for 15 projects designed by DEH and forwarded to the district for construction was examined to determine if the installation was at fault in providing insufficient time to allow district reviews to be conducted.

Review time sufficiency is illustrated in Table 18 and Figure 25. The numbers shown in the table are averages of responses in each project size category. An average shortfall existed of 3.1 days for small projects, 4.7 days for medium projects, and 7.1 days for large projects. The figures indicate that the time provided to do reviews fell far short of that needed given other job requirements. These numbers are supported by the qualitative responses to question 14 as shown in the figure. 74% of the respondents stated sufficient time was available only "sometimes", "rarely", or "never".

Table 18 Review Time Needed vs Review Time Available (Questions 6, 7, and 8)

NOTE: All times shown in work days.

	Avg Time Needed for	Avg Time Needed Given	Avg Time
Project_Size	Review_Only	<u>Other Regents</u>	<u> Available</u>
Small	2.3	9.7	6.6
Medium	3.5	12.5	7.8
Large	6.2	16.4	9.3



When asked the reasons why sufficient time was not available, the responses were as shown in Figure 26. Response choices included poor information, lack of funding, poor plans and specs, other work, short suspenses, and "other". The majority listed "other work" and "short suspenses" as the main causes of insufficient review time. This data was corroborated by responses to question 10. Question 10 asked respondents to list, in priority, their daily activities. They were asked to prioritize on the basis of the way things actually were and then to reprioritize based on the way they felt things ought to be. Each priority list was then divided into thirds. The priority of reviews was determined by determining how often they fell in the top, middle, and bottom third of each individual list. The results are shown in Figure 27. 73% of those answering this question prioritized reviews with the middle or lower third of other things they had to do. More significantly, if given the opportunity to change these priorities, most would not increase the position of reviews. In fact, several placed them at a lower priority. This indicates that, regardless of how important reviews were felt to be, other activities are considered just as important or more so.

Figure 26 Reasons Review Time is Not Available (Question 15)





The issue of competing work raised the question of exactly what kind of work reviews were competing against. Much of it was other essential tasks, such as design, mod processing, and quality assurance. But it was suggested that there were also significant administrative tasks which also consumed large amounts of manhours. These included reports, meetings, correspondence, and other requirements which detracted from essential, design/construction more related responsibilities. Many of these requirements were thought to be unnecessary by the individuals required to perform them. The impact of outside detractors on design and review quality was beyond the scope of this study, but it is an issue deserving further study.

The district required a minimum of 51 calendar days prior to contract award for DEH designed projects to be processed through the district review system. This time included administrative processing, copying, distribution, review comment coordination, and other activities besides the actual reviews. 15 projects were traced to see if "end of year crunches" or other reasons resulted in DEH not providing plans and specifications to the district with enough lead time within which to conduct a proper review. These included ten small, four medium, and one large project. All 15 projects were found to have been provided with sufficient time to meet the 51 day requirement. The average time prior to bid opening for small projects was 85.4 days, the medium projects 86.5 days, and the one large project 105 days. Consequently, it was concluded that timely submission by DEH did not contribute to the shortage of review time.

Funding is cited in other studies as a main reason reviews are not fully conducted. However, in this study, lack of funds was seldom identified as a significant detractor. Consequently, this area was not fully explored. Review funding for MCA projects was not studied at all. For OMA projects, the DEH automatically authorizes 12 hours of technical review per needed discipline. DEH also provides up to \$1000 for constructibility reviews. Discussions with DEH and district personnel indicate that it is a relativaly easy matter to obtain additional funds should this time prove insufficient for the review needed. DEH is prepared to provide a fund commitment within eight

hours of a telephonic justification for the additional funds (assuming such funds are available). However, such requests were rarely made.

It was concluded from this portion of the study that reviews were not likely to receive the priority and effort they need to adequately catch design deficiencies and site condition problems unless the review mission is conducted by individuals who have review as their first priority. This in turn implies the need to either rearrange existing priorities within the various offices now conducting reviews or establish separate constructibility and technical review cells. It is essential to controlling cost growth that one action or the other be taken.

Review Processing Procedures:

F

This section presents findings as to the procedural framework within which reviews are conducted. Specific issues addressed are:

- 1. Overall district review policy.
- 2. The frequency at which reviews are conducted at each stage of design, and the frequency that the same reviewer stays with a project throughout the design period.
- The availability of previous review comments on a given project, and the frequency at which comments are checked to insure incorporation in the final design.
- 4. Site visits during design and review.
- 5. The manner in which projects are assigned to individuals for review.

The potential for review inconsistency due to lack of standardized training was discussed in a previous section. This could

be mitigated by the existence of a standard review policy to include review aids, such as checklists. However, there was no standard policy in effect at the time of the study. Although various checklists existed, they were not widely disseminated and seldom used. A comparison of checklist availability versus use is shown in Figure 28. The bars with diagonal lines indicate the frequency of availability; the cross-hatched bars the frequency of use. 27 of 51 (53%) responses indicate checklists are rarely or never available to reviewers. 25 of 47 (53%) responses indicate checklists are rarely or never used. The difference in total responses for each category was due to not all respondents answering both questions. As can be seen from the graph, checklist use generally paralleled availability. However, having a checklist on hand did not always guarantee its use, quality, or completeness. For example, four respondents claimed to always have such an aid available, but only one reported always using it. Whereas the use of checklists would not be a panacea to the overall review issue, such an aid, used as part of an overall review policy, would provide guidance on what types of items were to be checked and how reviews should be conducted.

Existing engineer regulations did not provide specific guidance on how often reviews were to be conducted. ER 1110-345-100 required at least one review upon design completion. However, the regulation stated additional reviews may be required commensurate with the complexity of the project. ER 415-1-11 required a minimum of two constructibility reviews; one at the concept stage of design and the second upon design completion within 30 days before formal advertising.

Specific policy was left up to the individual divisions and districts. The district regulation in use at the time of this study discussed reviews at the "concept", "preliminary", and "final" stages of design. DEH design reviews were supposed to be completed prior to the design being sent to district, with a final review by district upon receipt. For district in-house designs and A/E designs, reviews were supposed to be conducted at the 35%, 90%, and 100% stages of design, although this varied from project to project depending on scope of work and type of review. In some cases, reviews were also done at the 65% design stage.

Figure 29 Checklist Availability vs Use (Questions 21 and 33)



Figure 29 presents data for the frequency of reviews at each required stage for new and renovation work. Data for new work projects

is shown by the bars with diagonal lines, and renovation work by the cross-hatched bars. For new work, 55% of the respondents report reviews done at all stages either "always" or "most of the time". For renovation work, this percentage drops to 40%. Of significance in both cases is the high percentage of times when these reviews are performed only "sometimes", "rarely", or "never". The main reason for this appears to be the lack of time and/or competing work requirements as identified in the previous section.

Figure 29 Frequency of Reviews at Each Design Stage (Questions 23 and 24) New and Renovation Work



Figure 30 shows data on how often the reviews that are conducted at each stage are done by the same individual. 54% of the respondents claim that the same individual does reviews at all stages

only "sometimes" or "rarely". This, by itself, is not necessarily a major problem provided the comments of one reviewer are made available to subsequent reviewers. Furthermore, some mechanism should exist to record which items were checked and which were not by each reviewer. Without this mechanism, the potential exists for some items being needlessly reviewed multiple times while other items are overlooked altogether.



Since different individuals conducted reviews at each design stage, it was essential that comments from previous reviewers be made available. The graph at Figure 31 presents the respondents view as to

how often this occurred. In the majority of cases (59%), previous comments are available "always" or "most of the time". What is significant, however, is the large percentage (41%) of instances when comments are only "sometimes" or "rarely" available. The implication is that there often was no way a problem identifieu in an early review could be passed on to subsequent reviewers.



Figure 32 shows how often it was felt comments submitted by reviewers were incorporated into final design, as well as how often designs were checked to insure all valid comments were incorporated. In only slightly more than half (54%) the responses was it felt comments "always" or "most of the time" found their way into the final

product. Submitted comments were checked to insure incorporation even less frequently. 48% of the respondents stated comments are checked "always" or "most of the time". Part of the problem may be perception; as will be shown, feedback to as to whether a comment was incorporated was frequently not provided to the reviewer. It should also be noted that the final decision on incorporation rests with the project manager. A comment considered valid by the reviewer may not be viewed in the same manner by the project manager. Unsolicited comments written on the questionnaire provide another explanation. If there was not time to incorporate comments by contract ammendment prior to bid opening, they may be tabled for later modification or dropped altogether. In any of these cases, it is possible that the reviewer may not have been aware of what took place.

The frequency with which feedback was given reviewers is presented Figure 33. In more than half of the responses (57%), feedback as to the quality of review comments was given only "sometimes", "rarely", or "never". This trend was seen in the responses of all respondent groups; it was not a problem isolated to any one agency. The lack of feedback, having different reviewers at each design stage, lack of overall review policy, and nonstandard training together create a serious obstacle to the conduct of consistent reviews.



Figure 33 Frequency of Feedback Given to Reviewers (Question 32)



Unknown sited conditions accounted for almost 22% of project cost growth. This problem is greatest for OMA and AFH projects where the majority of work was done in existing structures. Projects which incorporated thorough site visits as part of the design and review process should have been able to catch many of these problems prior to construction.

The questionnaire asked for data on the issue of site visits only as they were or were not conducted during reviews. It did not address visits made by designers during design. Furthermore, not all projects necessarily need site visits. The data which follows must be viewed keeping this fact in mind. However, it is the opinion of the author that, had site visits been conducted during project design, the magnitude of the problem with unknown site conditions would not have been so great.

The District did not have a formal policy requiring site visits during design review. The frequency visits were conducted is graphically shown in Figure 34. 69% of the respondents report visiting sites during reviews less than 25% of the time. Table 19 breaks this data down by type of review. In general, those reviewers located at the District (quality assurance and technical design reviewers) were much less likely to visit the site than personnel located in the field. 83% of the respondents from quality assurance and 46% from engineering division reported they never visited a site as part of the review process. Since constructibility reviews are performed by field personnel as well as quality assurance personnel, these reviews are

more likely to have site visits than technical reviews. Even so, only three of six respondents involved with constructibility reviews reported visiting the site more than 75% of the time.

Figure 34 Percentage of Reviews With Site Visit (Question 11)



Table 19 Frequency of Site Visits By Type of Review (Question 11)

> Percentage of Times Site Visits Conducted as Part of Review

Type of Review	0%	1-25%	26-50%		76-100%
Technical	13	13	1	0	1
Constructibility	6	3	1	3	3

NOTE - Figures indicate number of responses in each category.

If this data is further broken down by response group, the results are as shown in Table 20.

Table 20 Frequency of Site Visits By Response Group (Question 11)

Percentage of Times Site Visits Conducted as Part of Review

Response Group	0%	1-25%	26-50%	51=75%	76-100%
Engineering Div	13	13	1	0	1
Quality Assurance	5	1	Ō	0	ō
Resident Office	1	0	0	2	1
Construction Reps	0	2	1	1	2

NOTE - Figures indicate number of responses in each category.

The groups whose respondents tended to visit the site more frequently were construction representatives and resident office personnel. However, the actual amount of time these groups had to devote to reviews was extremely limited. Responses from the questionnaires and personal interviews with the individuals concerned found that over 50% of these respondent groups placed reviews in the bottom third of their priorities. Consequently, the reviews conducted by these two groups tended to be cursory at best. The conclusions drawn from this are:

- The top priority of work by these groups was the administration and quality assurance of ongoing construction. Given existing high work loads and staffing levels, it was probably not realistic to expect thorough reviews from either the resident office or the field construction representatives.
- 2. When these groups do conduct reviews, they are more likely to visit the site as part of the review process. 106

In the case of the construction reps, this was probably due to being located on the installation and having ready access to the project site.

3. 35% of those conducting constructibility reviews visited the site more than 50% of the time. As has been previously shown, however, those conducting these site visits were those with the least amount of time to devote to reviews. Consequently, the primary burden for the thoroughness of constructibility reviews fell to quality assurance, a group which seldom appeared able to include site visits as part of their reviews. (This is not intended to imply that quality assurance personnel never leave their office. On the contrary; much of the QA respondents time was spent doing field inspections of ongoing construction.)

Figure 35 presents the reasons given for not visiting the site. 68% of the respondents claimed insufficient time existed to visit the site either due to competing work requirements or short suspense dates. 17% reported that lack of funding was a significant reason. The significance of this chart is that it indicates internal priorities and procedures may have more of an impact on the conduct of site visits than external factors, such as funding.

Write-in responses to the "other" category included:

- 1. Reviewer was not asked to visit site.
- Site visits are conducted by the on site construction staff.
- 3. Site visits are a design, not construction, function.
- Not necessary since the on-site construction staff does it.
- Project complexity may not always warrant a site visit.
- 6. Construction site was too far to visit.



Section chiefs in design branch were individually asked as to how they assigned reviews to their personnel. In most cases, it was done on a work load basis. The individual with the least amount of work at the time when the review request arrived was given the review to do. All section heads stated that an attempt was made to match the individual's expertise with the nature of the project being reviewed. An attempt was also made to give reviews of a particular project to those familiar with the project. It was admitted, however, that this was not always possible. The first priority of design branch was project design. Since it was difficult to schedule reviews with any

long range accuracy, it was likely that, given the current system, they will continue be handled on a catch-as-catch-can basis.

Respondents were asked to evaluate the overall emphasis placed by "the system" on reviews. "The system" was defined in the questionnaire as being the district, DEH, the customer, and any of the other agencies that may be involved in preconstruction reviews. The results of this question is at Figure 36. Whereas the majority (53%) feel that it is "about right", a significant portion (36%) feel that the emphasis is "not enough".

Figure 36 Emphasis on Reviews by "The System" (Question 27)



Table 21 shows how responses to the question of review emphasis varied by response group. The largest number of those who feel it is

"not enough" are those engineers in Engineering Division. An explanation for number indicating review emphasis was "too much" may lie in an unsolicited comment from the resident office. The comment stated that given the current work load required in the respondent's section, the emphasis on reviews was "too much". In other words, it was felt there just was not sufficient time to do everything and that it was not possible to meet the standard expected in properly conducted reviews.

Table 21 Emphasis on Reviews By Response Group (Question 27)

Emphasis by "The System" Percentage of Responses per Response Group

Response Group_____Too_Much____About_Right____Not_Enough

Engineering Div	11%	50%	39%
Quality Assurance	0%	100%	0%
Construction Reps	17%	33%	50%
Resident Office	20%	20%	60 %
Project Managers	10%	60%	30%

Summary:

þ

The purpose of this chapter was to determine why the review system did not do a better job of controlling cost growth in the projects studied. This was done by collecting opinions of those involved with the review process, at the district level, by questionnaire. Data was collected in the areas of review personnel qualifications, time available, and procedural systems. The findings indicate weaknesses in the areas of review training, time availability due to competing work requirements, lack of measures to insure comments are transmitted between subsequent reviewers, lack of checking to insure incorporation into final design, and the lack of an overall review policy.

ł

CHAPTER 6

COMPARISON OF FINDINGS WITH PREVIOUS STUDIES

General:

The purpose of this chapter is to compare the findings of this study with the findings of others to determine which may be considered conclusive and which require further examination. The chapter is organized in the following manner. The next two sections identify those findings from Chapters 4 and 5 corroborated by other studies. The fourth section identifies the findings of this thesis which differ from those of previous studies. The fifth section discusses those findings examined in this study not found to be previously addressed. The final section is a chapter summary.

Most of the other papers, although related, were written to subjects other than the issues specifically addressed here. The data in these other studies were collected to support different theses. Fortunately, several authors included their raw data in appendices or tables. The methodology used in this study was applied to their data in order to arrive at a common basis of comparison. Consequently, the conclusions drawn from those calculations are not necessarily those of the author of the study from which the base data was taken.

Corroborative Findings - Causes and Costs of Mods:

Table 22 identifies the issues of Chapter 4 concerning the causes and costs of modifications. It further identifies those

findings supported by the research of others (annotated "Yes"), those findings in disagreement ("No"), and those issues not addressed in other papers ("N/A"). The remainder of this section discusses only those issues fully corroborated by other studies; conclusions of this study in disagreement or not addressed by others are discussed separately.

Table 22 Comparison of Findings - Causes and Cost of Modifications

> Findings in Agreement With <u>Other Studies</u>

Mod Causes - by Cost	Yes
Indirect Costs	Yes
Impact on Productivity	Yes
Potential Savings	Yes
Correlation by Project Size	No
Correlation by Designer	No
Correlation by Project Type	N/A
Mod Causes - by Items of Change	N/A
Disciplines of Design Def	N/A

Issue_Addressed____

The most strongly supported finding was the causes of modification cost. Six studies were identified which addressed the subject. All examined federally funded projects. A comparison of findings is at Table 23. The table identifies the study, the number of projects involved (when given), and a rank ordering as to the proportion of mods caused by each source. The numbers reflect the number one most frequent cause, the number two most frequent cause, etc. The lower the number, the larger the percentage of mods found due to that cause. Two papers did not quantify the number of mods due to each source. In these cases, an "X" is used to identify causes mentioned. The "X" signifies the cause as a major contributor to mods without a specific ranking as to its impact having been established. Each study is identified by author, if known, or title. The number in parenthesis in the first column following the identification is keyed to the endnotes at the end of this chapter. ł

Table 23 Comparison of Findings ~ Mod Causes

Causes of Modifications

Article or	No.	Dsgn	Site	User	Dsgn	Val	
<u>Study</u>	Proj_	Def	Cond	<u>Reg</u>	Choge	Engr_	Remarks
Diekmann (1)	22	1	3	2	4		
Rosmond (2)		1		2			1
TX A&M (3)		1,7,9	4	10	2,10,15	12	2
USAFEA-K (4)			1				3
Rowland (5)	20	X	X	X	X		4
ENR (6)	269	1					5
Ballou (7)		X	X	X			6
CERL ARMS (8)		1					7
This Study	25	1	3	2	4	5	

REMARKS:

- Quotes 1982 audit by Defense Audit Service, which lists no. 3 cause as plans and specs being incomplete at time of bid award.
- From list of top 16 mod causes. Others were changing technology, excusable delays, user interference, lack of proper review, and administrative actions. Multiple rankings due to differences in operational definitions.
- Concentrated on OMA projects in Korea. Identified no. 2 cause as OMA work overload.
- 4. Rowland did not quantify which caused the most mods.
- 5. Quoted GAD study of federal projects. No. 2 cause was cost estimating errors.
- 6. Does not specifically list design deficiencies, but does mention imprecise specification language and ambiguous drawings, both of which are included in the operational definition of design deficiency in this study.
- 7. Only mentions design deficiencies. Claims design deficiencies account for 56% of all mods.

These findings are conclusive. Five of the eight other studies list design deficiencies as the number one cause of contract modifications; two of the remaining three list them as a major contributor to changes. User requested changes are most frequently listed as the second and unknown site conditions as the third most common causes of mod cost. Consequently, a concerted effort to control these areas should result in significant cost growth savings.

The money that could be saved by controlling modifications was quantified in a study of constructibility reviews by Lloyd Finley. Finley studied two projects awarded by the Naval Facilities Engineering Command to determine the benefit/cost ratio between the cost of constructibility reviews and cost growth savings. It should be noted that his definition of constructibility review encompasses this paper's operational definition of both constructibility and technical design review. He calculated the costs to the government of doing the reviews and compared those figures with the amount the government might have paid if the errors discovered during the review had gone to construction. His findings are shown in Table 24. His findings reinforce the conclusion of this study that it is less expensive to identify and correct problems before contract award than after. (9)

Table 24Benefit/Cost Analysis of Constructibility Review(Finley's Study)

	Benefit to	Benefit to		
	Cost Ratio	Cost Ratio		
Project No.	(10%_mark-up)	(2%_mark-up)		
1	28.4	2.8		
2	6.8	1.5		

Although not quantified, these findings are qualitatively echoed in several other papers. The Texas A&M Research Foundation found:

"...the work accomplished by change orders and modifications is always more costly than similar work in the basic contract, and requires additional effort..." (10)

The Blue Ribbon Panel on Management of Construction Quality stated:

"...just one less change order per contract, resulting from improved customer-Corps communications will produce substantial Corps-wide savings." (11)

Harvey Kagan discovered an additional way modifications may increase project costs. In an article for the American Society of Civil Engineers' Journal of Professional Issues in Engineering, he cites an argument used by contractors to justify additional compensation. (12) Kagan states that contractors expect a certain number of modifications to be "normal" for a given type project. If the number of mods in a project exceeds the norm, contractors may ask for impact costs due to the amount of time spent preparing change order paperwork and In essence, their argument is that processing excessive estimates. numbers of modifications disrupts their work schedule and causes additional expenses, for which the government is responsible. In other words, the indirect costs of modification processing incurred by the contractor are passed on as direct cost to the government. Reducing the number of mods should reduce this source of cost growth as well.

There was also consensus among the studies as to the impact of modification processing on productivity and Corps indirect costs. Although none of the other studies attempted to quantify effort spent

on mods, several offered qualitative conclusions based on their research. The Texas A&M Research Foundation concluded:

"The chief mission of the Corps is to turn out good construction. If dollars, manhours, and talent have to be diverted from the chief mission of good construction to the issuance of contract modifications,...then the chief mission of good construction must suffer." (13)

and:

"Fewer change orders would reduce administrative cost and permit assignment of more manpower to inspection and construction problems which are of real primary importance." (14)

In his study on constructibility reviews, Lloyd Finley wrote:

"It should be recognized that any time "spent" by management in resolving conflicts that could reasonably have been caught prior to award, is time that might be more efficiently and effectively utilized on other related contract activities." (15)

It may therefore be concluded that modification cost is not limited to direct construction cost, and that significant savings can be realized through thorough preconstruction reviews. In addition to direct construction cost, the Corps pays for the lack of good reviews in terms of administrative overhead, effort, and lower productivity. These costs are real, even if they cannot always be quantified.

<u>Corroborative_Findings - Design_Review_System:</u>

Preconstruction reviews were the subject of numerous other studies. Recall this study addressed technical and constructibility reviews from the issues of personnel qualifications, time available, and procedural framework. Table 25 lists the specific issues examined under each of these categories and identifies those corroborated by other studies. "Yes" indicates corroborating findings from at least one other study with no studies in disagreement; "No" indicates one or more studies in disagreement; and "N/A" indicates issues not addressed in other papers. Supporting studies are specifically discussed in the paragraphs which follow. Findings from other studies not in agreement with those of this study are discussed in the next section.

Table 25 Comparison of Findings - Design Review Process

Findings in

Agreement With Issue Addressed_____Other Studies

Uross Iraining	Yes
Competing Work Requirements	Yes
Project Scope Definition	Yes
Site Visits	Yes
Reviews at Each Stage of Design	Yes
Same Individual does all Reviews	Yes
Coordination of Comments	Yes
Incorporation of Comments	Yes
Need for Overall Policy	Yes
Use of Checklists	Yes
Need for Reviews	Yes
Destructive Testing	Yes
Reasons Sites not Visited	Yes
Need for Separate Review Cell	Yes
Feedback Given Reviewers	N/A
Formal Review Training	N/A

Two studies addressed the issue of cross-training-between design and construction personnel. The Corps of Engineers Blue Ribbon Panel report stated:

"Procedurally, there is a need to improve consultation with field engineers in the design process of major projects. Likewise, there is a need to improve design participation in the actual construction process." (16) The report goes on to say:

"There is insufficient cross-training [sic] and joint experience among engineering division and construction division professionals." (17)

The second report, prepared by the Texas A&M Research Foundation, recommended one manner in which problems with inadequate or incomplete designs could be corrected would be to "give designers field experience in the administration of construction contracts." (18).

The findings of this study reinforce the conclusions of the Blue Ribbon Panel and the Texas A&M Research Foundation. The professional capabilities of designer and constructor would be significantly enhanced if each had some experience in the field of the other.

Table 26 presents the major review problems identified in this study and identifies how often those problems appear as major causes of poor reviews in other studies. The first column identifies the study, along with a key to the endnotes at the end of the chapter. The priority given by each author to each problem as to its relative importance in weakening the review process is shown. Those problems not rated in relative importance are marked with an "X".

The four most often cited problems are lack of time due to peak work periods/competing priorities, poor scope definition, lack of site visits, and uncoordinated review comments. These problems have been cited by independent researchers, Corps of Engineer panels, and members of private A/E firms involved in government work. The problems of multiple reviewers, coordination of review comments, and incorporating comments into final design are also often mentioned other studies.

Table 26 Comparison of Findings - Problems in the Review Process

Causes of Review Problems

	Time/	Proj Scopo	Si ta	Same	Connect	
<u>Study</u>	Work	Def	Visits	Each_Stage	Coord_	Roks
TX A&M (19)	2	1	X		9	1
TX A&M (20)	Х	X				2
Dsgn Forum (2)	1)	X	X	X	X	3
Grn Ribbon (22	2)				X	4
This Study	X	X	Х	X	X	

REMARKS:

- From list of top ten review problems. Other design review problems included unnecessary reviews, poor cost estimates, and poor quality of A/E work. Study also addressed inadequate coordination with using agency and lack of proper site investigation.
- This was a subsequent A&M study. Other causes included insufficient information to A/E, unnecessary reviews, and budget cycle.
- Also cited use of checklists, unrealistic fee structure, and lack of proper as-built drawings.
- Study emphasized customer involvement in reviews and only generally discussed need for quality review.

These issues were frequently cited by participants in the Design Quality Forum as detractors from project quality (23), as well as the Green Ribbon Report (24) and Texas A&M's May 1968 study (25). It may therefore be concluded that the problems with the review system identified in this study are not unique to the sample from which the data were collected. It is further concluded that all of these areas need to be addressed if efficient reviews are to be conducted; attacking only one or two by themselves will not correct the problem. What is needed is an all encompassing review policy within each

district specifying procedures to insure the impact of these problems is minimized. The "Corps needs to encourage Districts to quantify technical criteria for projects and establish a policy of rational, analytical review." (26).

The advantages of using review checklists were cited in several studies. The most frequently referenced checklist system was the Navy's "Redicheck". "Redicheck is a structured review system which provides a logical and orderly approach to checking construction drawings at the 100% design stage." (27) Redicheck was favorably referred to by Finley (28) and participants in the South Atlantic Division's Design Quality Forum (29). Checklists, however, should not be considered a panacea. They need to be incorporated with overall review policy and procedures. The advantages of using a checklist are that it facilitates review continuity (especially when reviews are conducted by different people at different stages of design) and it provides a logical, consistent review methodology.

The importance of reviews in producing quality construction projects and reducing cost growth is acknowledged or implied in each of the studies surveyed. Finley's analysis of review cost versus project cost was previously cited. In his study of the contractor quality control system, Henry Turowski stated:

"The design review has been called by some to be the most important tool for developing inherent quality or reliability in a product." (30)

Kagan, in his article, emphasizes the need for review to coordinate drawings between different design disciplines. He warns:

"If the coordination of drawings is first performed by the contractor, the cost will be considerable more." (31)

The Corps apparently shares this philosophy, as witnessed by the regulations requiring reviews be performed prior to bid advertisement. At the functional level, however, this guidance appears to be frequently not complied with. Furthermore, this problem has been identified before, with seemingly little progress made in correcting it. Extracts from the Texas A&M Research Foundation's 1968 report state:

"An extremely high percentage of plans and specifications are published each year which are inadequate due to conflicts or errors... ...it is necessary to advertise projects for bids with insufficiently prepared plans... ...the only effective review period is during advertisement... ...comments of using service and other interested parties frequently are received too late to incorporate in the advertised project." (32)

Yet 17 years later, the Green Ribbon Panel found "Procedures to provide quality design review...need strengthening." (33) Specific weaknesses noted in other studies (most of which were completed since 1980) were previously referred to. Thus, despite publication over the years of numerous regulations, engineering pamphlets, studies, reports, and manuals, many Districts still fail to effectively manage one of the most important tools for controlling project cost growth.

In addition to direct cost, indirect cost, and lower productivity, the Corps pays for poor review procedures in two other areas as well; customer satisfaction and professional reputation. It was suggested in Chapter 4 the reason for the high percentage of user requested changes was failure to properly define project scope during design. Installation involvement during the initial stages of design is essential to insure the needs and desires of the customer are integrated into the project. Failure to involve the customer can lead to a deterioration in the working relationship between District and DEH and lower design quality. In 1983, the Blue Ribbon Panel stated:

"During construction, the Corps must employ better techniques to assist the customer/user in understanding the project's progress and how the project compares with his original expectations. The process of responding to customer/user inquiries must also be improved." (34)

}

This conclusion was echoed in the 1985 report of the Green Ribbon Panel:

"...there is ample evidence to support the perception that the DE [District Engineer] does not fully involve his customers during the design process. The most widespread comments include: lack of opportunity to participate in the design reviews, slow response to requests, poor feedback on design review comments, claims of failure to consult the customers on problems and perceived lack of concern by the DE over customer needs." (35)

Although input from the installation is essential, it should be limited to the functional aspects of the project, and not include technical design or construction techniques. The 1968 Texas A&M Research Foundation study found that unnecessary reviews by the customer, particularly when they include reviews of technical design, were one of their top ten design review problems. (36) Their 1969 report was more specific:

"The Corps of Engineers needs [to adopt] a procedure wherein only the functional aspects of the design are presented to the User [sic] for review..." (37)

The primary reason for controlling customer input to the review process
is the cost and time that may be lost reconciling their comments with those of district reviewers. This problem may be aggravated by different interpretations of design criteria, preferences in material or construction methods, or differing opinions on architectural aesthetics. Thus, it may be concluded that custumer interface is essential for improving user satisfaction, understanding the installation's expectations, and defining project scope. Once the project scope has been determined, however, decisions on technical design and construction methods to meet those requirements are best left to the supporting district.

growth, numerous It is obvious that excessive cost modifications, and failure to address user concerns will hurt the Corps' professional reputation with the customer. That reputation may also be damaged within the greater engineering professional community, particularly with the reliance placed on private A/E firms to design many Corps projects. Perceptions held of the Corps' by several A/E firms, although not specifically mentioned, can be inferred from comments made during the South Atlantic Division's Design Quality Forum. The objective of the forum was to identify the significant things that adversely affect the quality of the product private A/E firms provide the Savannah District. Frequently cited complaints included poor definition of project scope; failure to edit, coordinate, and consolidate design review comments; an inflexible fee structure which does not compensate for predesign conferences or site visits; and loss of continuity when the district has different reviewers at each stage. One participant stated it costs 30-40% more to do Corps work

than commercial work because of the tendency to stop and start work and the time it takes A/E's to respond to review comments. (38) These and other problems closely parallel the issues raised in this and other studies. Consequently, it may be inferred that the opinions held by the forum's participants would be shared by A/Es facing similar circumstances in other locations.

The final area of general agreement was the use of separate review teams as a means to address problems with the conduct of reviews. Two studies supported the position taken in this paper that separate review cells are needed to insure reviews get the priority they need to be effective. The first the 1968 Texas A&M Research Foundation report which concluded such a cell would contribute to providing more complete reviews of plans and specifications. The report states:

"It is possible that a reorganization of units in the Engineering Division, with a special group set up specifically to review plans and specifications, is a preferred sequence." (39)

In his study of contractor quality control on Navy projects, Turowski concluded many potential problems could be eliminated by setting standards for design quality and thoroughly reviewing all aspects of the plans and specifications. He further stated:

"To do this correctly, the Navy would be required to invest a considerable amount of its EFD talent into the formation and operation of design review teams. These teams, composed of experts in each aspect of the project, would review the potential contract in far more depth and detail than is presently being accomplished. A review conducted in this manner would not only clear up design problems before they became construction problems, but would also alert the [contract A/E] designers that the Navy is interested in complete, quality designs. (40)

The creation of a separate review cell does not imply completely divorcing review from design. These functions are closely intertwined. The review team should be located within Engineering Division to facilitate communications between designer and reviewer. But it seems essential that different individuals perform these tasks. It is unrealistic to expect reviews to receive the priority needed to be effective if forced to compete with other essential tasks.

Contradictory Findings:

Two findings of this study were found to be in substantial disagreement with the findings of one or more other studies. These were the correlation of cost growth by project size and the correlation of cost growth by designer.

The issue of mod cost and project size was addressed by the Diekmann-Nelson study and in Henry Rowland's masters thesis. Table 27 compares the data of their studies with the data of this study. It should be noted that the methodologies they used differed from the methodology used here. In order to make a meaningful comparison, the data from their studies had to be recomputed using the methodology of this study. Consequently, the figures shown in Table 27 do not necessarily reflect the calculations of these other authors.

The findings of this study, expressed in Chapter 4, was that smaller projects tended to register higher levels of cost growth. This was not the trend established in the other two. Diekmann's data shows no discernable trend; Rowland's indicates that cost growth rises as projects become more expensive. It is therefore not possible to draw

meaningful conclusions concerning cost growth and project size until further research on this subject is conducted.

Table 27 Comparison of Findings - Cost Growth as Function of Project Size

Diekmann-Nelson_Data_(41)

Proje	ct			Avg.			
Size		No.	No.	Cost	Cost		
(\$000)		<u>Proj</u>	Claims	Growth_	Per_Claim		
Less Than \$	1,000	7	34	3.9%	\$ 5,000		
\$1,000 - \$	5,000	9	128	6.5%	\$16,000		
More Than \$	5,000	6	151	6.0%	\$26,000		

Rowland_Data_(42)

Project		Total					
Size	No.	No.	Cost	Cost			
(\$000)	Prei	Claims	Growth	Per_Claim			
Less Than \$1,000	4	42	6.2%	\$ 3,400			
\$1,000 - \$5,000	11	187	7.7%	\$11,700			
More Than \$5,000	4 (NOTE) 87	8.9%	\$31,100			

NOTE: Rowland originally studied 20 projects. He deleted one, however, as it had too many change orders and would lead to a false interpretation of data.

<u>This_Study</u>

Project		Total		Avg. Cost Per Claim		
Size (\$000)	No. Proj	No. Claims	Cost Growth			
Less Than \$1,000	19	96	10.5%	\$ 7,600		
\$1,000 - \$5,000	6	88	3.5%	\$ 7,600		
More Than \$5,000	0	0	-	-		

Both of the other studies considered the number of modifications per project to be a significant indicator. Rowland used the average number of mods per project by size category as an indicator of complexity. The Diekmann-Nelson study used the average cost per mod for the same purposes. A comparison of the average cost per claim in each study shows a general trend toward higher cost mods in larger projects. However, it is the opinion of this author that this is not really meaningful, since the number of items of change which may be incorporated into a single mod is not taken into account. Both of the other studies address the issue of multiple change mods, but neither quantify the number of changes per mod each of their projects experienced. Had they done so, a meaningful comparison between the cost per item of change could have been made in each of the size categories. As it is, no such comparison is possible.

The Diekmann-Nelson study also addressed the issue of design by in-house versus contracted A/E firms. A comparison of findings is at Table 28. Although the cost growth for A/E designed jobs is fairly close for both studies, there is significant variation in the cost growth for projects designed by government in-house personnel. This is possibly due to differences in local policies and procedures, although more study of this issue is needed before definitive conclusions can be drawn.

The conflicting findings of these studies indicates that data collected to date on the issues of cost growth as a function of project size and design agency is insufficient to make any generalizations. Further research is needed, at differing locations, before definitive conclusions on the impact of design agency on cost growth can be reached.

Table 28 Comparison of Findings - Cost Growth as Function of Designer

Diekmann-Nelson Data (43)

				Total	
		Proj		Claim	
	No.	Amount	No.	Cost	Cost
Designer	Proj	(\$000)	<u>Claims</u>	(\$000)	<u> </u>
In-House	8	\$14,860	89	\$1,570	10.6%
Contract A/E	14	\$89,040	224	\$4,560	5.1%

This_Study

				Total	
		Proj		Claim	
	No.	Amount	No.	Cost	Cost
Designer	Proi	(\$000)	<u>Claims</u>	(\$000)	Growth
In-House	5	\$14,860	27	\$ 416	2.8%
Contract A/E	19	\$20,354	225	\$ 1,201	5.9%

Findings_Not_Addressed_in_Other_Studies:

Five issues were examined in this study that were not found in any of the others. These issues were:

- 1. Causes of modifications by items of change.
- 2. Cost growth as a function of project type.
- 3. Disciplines which contribute most to design deficiencies.
- Impact of failing to provide feedback to reviewers as to the disposition of comments made during reviews.
- 5. Impact of formal training of review personnel on review consistency.

Consequently, the conclusions drawn are made within the context of the study sample. As with those issues in disagreement with previous

findings, additional study in these areas is needed before definitive conclusions can be drawn.

Summary:

This chapter compared the findings of Chapters 4 and 5 with those of previously completed studies, reports, articles, and theses. Most of the findings of this study were corroborated by one or more other papers. This corroboration lends external validity to the methodology and findings of this study. On many of the corroborated issues, the findings may be considered conclusive. Seven issues, however, were found where findings were either in disagreement with previous studies or not addressed. Conclusions reached in these areas are therefore tentative and subject to verification by further research. The next chapter summarizes these conclusions and specifically identifies recommendations for continued study. Endnotes - Chapter 6

1. James A. Diekmann and Mark C. Nelson, "Construction Claims: Frequency and Severity", <u>Journal of Construction Engineering and</u> <u>Management</u>, (Vol. 111, No. 1, March 1985), pp 75-77.

2. James R. Rosmond, "Analysis of Low Bidding and Change Order Rates for Navy Facilities Construction Contracts", (Master's thesis, Naval Postgraduate School, Monterey, June 1984), p. 28. Rosmond quotes a 1982 audit by the Defense Audit Service; the main thrust of his thesis is the manner in which contractors attempt to increase their profits through change orders.

3. Texas A&M Research Foundation, "A Systems Approach to Design and Construction for the Corps of Engineers", (TR 68-041, College Station, Texas, May 1968), pp. 142-143.

4. USA-CERL, "Ways to Improve Construction Contract Modification Processing: USAFEA Korea Case Study", (CERL Technical Report P-85/11, May 1985), page 8.

5. Henry J. Rowland, "The Causes and Effects of Change Orders on the Construction Process", (Master's thesis, Georgia Institute of Technology, School of Civil Engineering, November 1981), pp. B-I - B-III.

6. "Design Changes: The Largest Cause of Overruns", <u>Engineer News</u> <u>Record</u>, (March 6, 1975), page 10. Short article which reports findings of a GAD study of 269 federal projects. Not all projects were managed by Corps of Engineers. Other management agencies cited included Bureau of Reclamation, Federal Highway Administration, and General Services Administration.

7. L. Dennis Ballou, "Productivity Improvement", <u>The Military</u> Engineer, (Vol.77, No. 503, Sept-Oct 1985), pages 484-487.

8. USA-CERL, "Automated Review Management System", Users Manual, 3 May 1985.

9. Lloyd S. Finley, "Examination of the Constructability [sic] Review in Government Contracting", (Master's thesis, Purdue University, School of Civil Engineering, 3 August 1984), pages A-I-11 and A-IV-13.

10. Texas A&M Research Foundation, May 1968, page 143.

11. Corps of Engineers Report of the Blue Ribbon Panel on "Management of Construction Quality in the U.S. Army Corps of Engineers", (Office of the Chief of Engineers, March 1983), page 2.

12. Harvey A. Kagan, "How Designers Can Avoid Construction Claims", <u>Journal of Professional Issues in Engineering</u>, (ASCE, Vol. 111, No. 3, July 1985), page 103.

13. Texas A&M Research Foundation, May 1968, page 112.

14. Texas A&M Research Foundation, May 1968, page 143.

15. Finley, page 5.

16. Blue Ribbon Panel, page 12.

17. Blue Ribbon Panel, page 14.

18. Texas A&M Research Foundation, May 196B, page 149.

19. Texas A&M Research Foundation, May 1968, pages 128-129 and 199.

20. Texas A&M Research Foundation, "Systems Analysis of Corps A/E Design Engineering", (College Station, Texas, June 1969), pages 41-49.

21. Corps of Engineers South Atlantic Division, "Minutes of the Design Quality Forum", (Savannah District, 3 September 1985). Forum was attended by members of the Corps of Engineers and civilian design firms contracted for work by the Savannah District. Purpose of forum was to identify the most significant things which affect the quality of the product private A/E's provide the Corps.

22. Corps of Engineers Report of the Green Ribbon Panel on "U.S. Army Corps of Engineers Support to Army Installation Commanders", (Office of the Chief of Engineers, March 1985), page 11.

23. Design Quality Forum Minutes, pages 2, 4, and 6-9.

24. Green Ribbon Panel, page 11.

25. Texas A&M Research Foundation, May 1968, page 110.

26. Texas A&M Research Foundation, June 1969, page 19.

27. William T. Nigro, "Redicheck: A System of Interdisciplinary Coordination", reproduced from <u>DPIC Communique</u>, (Design Professionals Insurance Company, April 1984), page 1 of 4.

28. Finley, page 18.

29. Design Quality Forum Minutes, pages 8 and 10.

30. Henry J. Turowski, "Contractor Quality Control", (Master's thesis, Naval Postgraduate School, Monterrey, December 1980), page 114.

- 31. Kagan, page 104.
- 32. Texas A&M Research Foundation, May 1968, pages 143-144.
- 33. Green Ribbon Panel, page 11.
- 34. Blue Ribbon Panel, page 4.
- 35. Green Ribbon Panel, page 11.
- 36. Texas A&M Research Foundation, May 1968, pages 130 and 134-135.
- 37. Texas A&M Research Foundation, June 1969, page 19.
- 38. Design Quality Forum Minutes, pages 2-9.
- 39. Diekmann and Nelson, pages 76-79.
- 40. Rowland, appendices B and C.
- 41. Diekmann and Nelson, page 79.
- 42. Texas A&M Research Foundation, May 1968, page 149.
- 43. Turowski, pages 114-115.

CHAPTER 7

CONCLUSIONS AND RECOMMENDATIONS

General:

This chapter summarizes the study's findings, conclusions, and topics recommended for further study. Those findings supported by other studies are considered conclusive unless otherwise noted. Findings which differ from those of other studies, or concern issues not previously addressed, and which will require further study before definitive conclusions can be reached are so identified.

<u>Causes of Modifications and Cost Growth:</u>

The major causes of modifications and cost growth in military construction projects are design deficiencies, user requested changes, and unknown site conditions. Whereas there may be local variances to this general conclusion (as noted in the USAFEA Korea case study (1)), almost every previous study which examined this issue cited one or more of these reasons as primary problem areas. Furthermore, the impact of these problems can be directly reduced by thorough and complete preconstruction design reviews.

The findings of this study indicate that the two design disciplines wherein most design deficiencies occur are architectural (57.5%) and mechanical (23.6%). Civil, electrical, and structural problems together account for the remaining 18.9%. Consequently, if the emphasis of review was placed on architectural and mechanical items of design, significant cost growth reduction could be achieved. This conclusion, however, is based on the findings of this study alone and therefore must be considered tentative until further research on this topic is conducted.

<u>Cost_of_Modifications:</u>

The Corps of Engineers pays for modifications in five ways: direct construction costs, indirect administrative and overheac costs, lower productivity, reduced customer satisfaction, and loss of professional reputation. Direct costs are the most visible, as they are seen in the dollars actually paid to the contractor for contract changes. Fewer modifications would result in a direct, quantifiable reduction in cost as measured in spent dollars.

The impact on indirect costs and productivity is more subtle. Districts operate with relatively fixed staffing levels. The work of modification design, negotiation, and administrative processing must be accomplished by the same people responsible for project design, contract administration, design review, quality assurance, and construction inspection. Consequently, the processing of modifications must be done at the expense of these other critical functions. This can result in a cycle (as was observed in the resident office during this study) whereby the quality of review suffers because those responsible for review are swamped by more immediate requirements to process mods and day-to-day construction functions. This lower design and review quality then results in modifications to future projects. The processing of these additional mods in turn impact on the ability to conduct future reviews, and so on. The district finds itself reacting to mode as they occur instead of acting to minimize their occurrence. Reducing the number of mode to be processed would improve productivity by freeing effort that could be concentrated on improved design and review of future projects and quality assurance of ongoing construction.

Even less quantifiable than productivity is the cost of customer dissatisfaction. The findings of the Green Ribbon Panel indicate this problem is more a function of the manner in which reviews are conducted than the number or cost of mods incurred. In the report. installations were described as feeling that districts often do not sufficiently involve installations in the review process, do not heed review comments, and are generally insensitive to customer needs. This in turn can result in increased numbers of user requested changes as construction progresses and it becomes apparent that the facility being built may not meet the use the customer intended or perceived. This situation is not always the fault of the Corps. Even a project which received a full using service review during design can have changes made during construction due to turnover of personnel within the using agency or changed requirements. But having the customer involved from the beginning can go a long way to reducing the number of mods and the potential for negative feelings that may be generated on both sides when such changes become necessary. Installation involvement should be limited to definition of project scope and functional design review, begin at the concept stage of design, and continue (at the functional level) through subsequent reviews. Comments on technical design

issues, materials, construction techniques, etc. should be left to the District.

Like customer satisfaction, the issue of the Corps' reputation within the professional engineering community seems more a function of the manner design and reviews are handled than the number of modifications produced. Comments expressed during the South Atlantic Division's Design Quality Forum cite problems with coordination of review comments, lack of funding for site visits, poor project scope definition, and lack of continuity in design review. A member of one participating A/E firm stated it costs "30-40% more to do Corps work than commercial work because of the tendency to stop and start work and the time it takes to respond to comments." (2) This cost is of course passed on to the government. Providing a smooth, efficient design-review system would help diminish these problems and enhance the Corps' overall reputation.

The conclusions drawn regarding mod impact on productivity, customer satisfaction, and professional reputation are drawn from limited sources and must therefore be considered tentative until further study is made. Future studies should attempt to quantify the impact these areas have on district overhead, design and review quality, and customer relations.

Impact of Reviews on Modifications and Cost Growth:

Technical, constructibility, and using service reviews have a direct impact on reducing modifications and cost growth. These were Projects receiving all three of these reviews were found to have

overall lower cost growth (4.7%) than projects not receiving all reviews (15.4%). Additionally, there was a lower percentage of cost growth due to design deficiencies for projects fully reviewed. However, the data indicates that review practices may not always provide a consistent quality of review for all projects. Of the 21 projects receiving all three reviews, nine (43%) still had unacceptably high levels of cost growth. Furthermore, reviews conducted on the projects studied did not adequately anticipate site condition problems and user requested changes. It is therefore concluded that:

1. In general, reviews reduce cost growth.

ł

- All three types of review are essential. Failure to conduct any or all of them can result in significant cost growth.
- In order to be effective, reviews must consistently be conducted in a systematic and coordinated manner.

An attempt was made to determine review cost effectiveness. However, accounting procedures in practice at the district from which this data was collected did not record amounts or time spent on review for each project. One study, Finley's, was found where a benefit/cost analysis was conducted on constructibility reviews on Navy projects. Although the results indicate an excellent return on investment for review time, the findings must be considered tentative as they were drawn from only two projects. Additional study, using Finley's methodology, is necessary to reach a definitive conclusion. It is recommended such a study be conducted for reviews at each stage of design, in order to determine a "break even" point between reduced cost growth and unnecessary review. Impact of Project Type on Modifications and Cost Growth:

OMA and AFH projects were found to experience higher levels of cost growth than MCA projects. OMA and AFH projects tend to be less expensive and do not appear to receive the same quality of preconstruction reviews as the higher-priced MCA projects. MCA projects, on the other hand, seem to be more thoroughly reviewed and consequently have a lower percentage of their adjusted contract amounts spent on modifications. This is somewhat understandable, since MCA projects tend to be more expensive and each percentage point of cost growth represents a higher dollar amount than a corresponding increase in growth for an OMA or AFH project.

But where OMA and AFH projects have a lower impact on dollar cost, they may have a much greater impact on customer satisfaction. Most projects ongoing at the installation studied were DMA and AFH Often they were constructed under conditions of joint funded. occupancy, where the user was still occupying the facility being worked Consequently, these projects had much greater customer visibility on. than MCA projects which tended to be isolated and under Corps control. It seems customers in general have a higher regard for the way the Corps manages MCA projects than they do OMA/AFH projects. This may be due to the fact the customer is more insulated from day to day problems on larger projects than those where facilities are shared. It is concluded that greater attention must be paid to these smaller projects in order to improve customer relations and enhance the product provided the installation by the Corps.

This finding is tentative, however, since no other studies were found which addressed the issue of cost growth as a function of project type. Additional research in this area is needed. Recommend such a study concentrate not only on cost growth but also the perceptions of the customer as to which types of project seem to be the more efficient.

Impact of Project Size on Modifications and Cost Growth:

The issue of project size is closely related to the issue of project type, since OMA and AFH projects tend to be smaller than MCA projects. The findings of this study indicate high-cost projects experience lower percentages of cost growth than do smaller projects. This finding, however, does not agree with the findings of other studies. The Diekmann-Nelson data shows no correlation between cost growth and size, whereas the Rowland data indicates cost growth increases with increasing project size. These differences could be due to varying local practices and conditions, or it could be that the data base is just too small from which to determine general trends. In either case, this is an area requiring additional study before general conclusions may be drawn. The close relation of project size to project type would indicate that perhaps one study could address both issues.

Impact of Design Agency on Modifications and Cost Growtur

Like the issue of project size, the findings of this study concerning cost growth and design agency differ from the findings of

the only other study found which addressed this area. Data from the Diekmann-Nelson study indicates cost growth for projects designed by contracted A/E firms (5.1%) is significantly lower than projects designed by in-house government engineers (10.6%). The results of this study, however, show just the opposite. The cost growth of A/E designed projects (5.9%) is higher than those of in-house designed projects (2.8%). It is interesting to note that the cost growth of A/E projects for both studies is roughly equal (5.1% to 5.9%). The differences in the figures for in-house design may be due to different local procedures and practices within the respective districts studied. It may tentatively be concluded that A/E designed projects can expect 5-6% cost growth, while cost growth for in-house designs will vary depending on local conditions. These conditions may include review policy, design policy, relationships with the supported installations, internal procedures, the proximity of the designer to the project for site visits, and work load. The issue of design agency and cost growth, however, is another area requiring further study before definitive conclusions may be reached.

Qualifications of Review Personnel:

The personnel performing technical and constructibility reviews on the 25 projects included in this study generally possessed the requisite education, training, and experience to conduct good reviews. Those responsible for technical review tended to have more education and experience in technical design. Many held professional certification within their design fields. Those responsible for constructibility reviews had less formal education, but much greater practical experience, especially in construction. Almost everyone was convinced good reviews saved the government money and should be conducted for all types and sizes of projects. Consequently, poorly qualified individuals or people with poor attitudes toward reviews can be eliminated as major causes of problems with the system used to review the projects of this study.

Two potential areas of weakness were found. The first was the lack of construction experience among those doing technical reviews and the lack of design experience among those doing constructibility reviews. It would be a distinct advantage for someone designing (or technically reviewing) a project to have some experience in how that project would be constructed. Similarly, it would be advantageous for the individual doing constructibility reviews to have some experience with technical design. This cross training was noticeably absent from the personnel reviewing the projects studied. The second was the relative lack of standardized formal training in the conduct of reviews. It has been noted that almost everyone involved with reviews had education and experience within his or her area of expertise. However, the lack of a common base of training results in most of those doing reviews approaching them from the framework of past experiences. This does not lend itself to providing a consistent product due to the varied backgrounds of the individuals concerned. Whereas these areas by themselves do not present major problems, they can aggravate the overall lack of review quality consistency caused by the other procedural problems identified.

Review_Time_Available:

The major problem with reviews in this study was found to be the lack of available time. The time provided reviewers was found to be far short of the time needed to do reviews and other competing work requirements. Those tasked with the mission of technical and constructibility review were often the same individuals doing project design, quality assurance, construction inspection, modification processing, and other contract administration functions. It is concluded that reviews are not likely to receive the priority needed to be fully effective unless:

- the review mission is conducted by individuals who have review as their first priority of effort or
- the limitations imposed on review quality by reviewers having multiple responsibilities are recognized and planned for.

This could involve either:

- a separate technical review team in Engineering Division and a separate constructibility review team within Construction Division or
- one review team containing both design and construction experienced personnel

and

3. field representatives working with the review teams to make maximum use of the construction inspectors on-site expertise without placing the full responsibility for thorough constructibility review on their shoulders.

Since the functions of review and design are so closely intertwined, the review team (or teams) should be physically and organizationally located close to those doing project design. By separating reviews in this manner, it would be possible to place a concentrated effort on reviews and thereby reduce modifications and cost growth.

Review_Procedures:

A variety of procedural problems were discovered during this study. Although no one problem was found serious enough by itself to be significant, the combined effect of all is concluded to be major a detractor from review quality. Consequently, what is needed is an overall policy within the District that provides quidance on each Such a policy should address the following items for both issue. technical and constructibility reviews:

- 1. Guidance on what to review and how to review it. The use of checklists (such as the Navy's "Redicheck" system) could be used as a working tool to provide this guidance. In addition to assisting reviewers, a completed checklist would assist managers in coordinating reviews at different stages between different reviewers. The completed checklist would provide an indicator of which items have been reviewed and which have not (for whatever reason). Managers could use this information to determine where to conceptrate future review effort.
- 2. Information flow procedures. This should include comments between reviewer, designer, and the customer. It should also include feedback from the designer to the reviewer as to the disposition of review comments. Finally, it should include methods insuring comments from previous reviews are made available to all subsequent reviewers.
- 3. Coordination of review comments. All review comments should be edited, evaluated, and consolidated prior to being turned over to the designer for action. This would preclude the designer having to decide between conflicting review comments. Techniques should be developed to insure appropriate review comments are incorporated into final design.
- 4. Site visits and use of destructive testing. Guidance should be provided on when site visits are needed, who conducts them, and what should be looked for at the site. 144

- 5. Formal review training procedures and standards.
- 6. Qualifications of review personnel.
- Role of the customer in the review process. Buidance should be provided as to the type of comments expected from the customer (limited to scope definition and functional items) and the district's role in using service reviews.
- The criteria by which reviewers are assigned to each project.

Such a policy, if properly enforced and combined with specialized review teams, could substantially improve review procedures and thereby reduce modifications and project cost growth.

Recommendations for Further Study:

The following is a summary of those topics recommended for additional study before final conclusions can be drawn:

- The causes of user requested changes and the method by which functional reviews (or other actions) by the using service and the Corps can be used to preclude those causes.
- Sources of design deficiencies and the design disciplines within which they are most likely to occur. This in turn could lead to a model for the conduct of reviews which focusses on the areas most likely to lead to modifications.
- 3. The impact of modifications, cost growth, and design/review procedures on District productivity, customer satisfaction, and the Corps professional reputation. Although often discussed, no studies were found which made a formal effort to analyze the effects of mods on these intangible elements of cost.
- 4. The cost effectiveness of reviews in relation to project cost growth. Recommend a study be conducted, using Finley's benefit cost analysis model (3), to quantify the actual benefit of technical and constructibility design review. It is further recommended that such an analysis

be conducted for reviews at each stage of design, with the intent of determining the break-even point between reduced cost growth and excessive, unproductive review.

- 5. The relationship between project type, project size, and design agency on change orders and cost growth. Although addressed in other studies, the findings to date are not in agreement and therefore not conclusive. Establishing a relationship between cost growth and these factors would help provide a focus for review effort.
- 6. The extent and impact non-essential administrative requirements, imposed on design elements within Engineering Division, detract from design and review quality. Data collected during this study indicated competing work requirements as a major reason reviews were not conducted to the level of thoroughness desired. However, the exact nature of those other tasks was not examined. Discussions with personnel within Engineering Division indicates many of them may be unnecessary. Identifying and eliminating unnecessary requirements could greatly assist in improving design and review effectiveness.
- 7. The impact of current funding rates and procedures on the quality of design and review. It is ironic that the Government places rather strict constraints on the money that can be spent on design and review and yet always has the money to pay later claims. Research in this area should concentrate on establishing realistic rates for design and review based on local conditions and project type, size, and scope.

Summary:

Excessive mods and cost growth are symptomatic of problems within the design and review process. The problems identified during this study are not new. Design deficiencies, unknown site conditions, and user requested changes have been identified in numerous studies and reports over the years as having substantially increased cost growth on military construction projects. It has been shown that these areas can be controlled by a thorough preconstruction design review process.

This process includes technical, constructibility, and using service reviews as well as improved customer involvement in the design and review process and improved review procedures. The chief mission of the Corps is good construction. Realizing that goal requires the dedication of resources long before construction actually begins. Α commitment must be made by customers, Districts, and the Corps as a whole on this issue. Failing to devote adequate resources to preconstruction design and review activities is false economy. The connection between good reviews and low cost growth has been well established by this and other studies. One way or the other, reviews are going to be paid for. They can either be paid up front, with a corresponding decrease in cost growth, or they will be paid for during construction with lower District productivity, increased indirect cost, project delays, reduced customer satisfaction, and higher construction costs.

Endnotes - Chapter 7

1. USA-CERL, "Ways to Improve Construction Contract Modification Processing: USAFEA Korea Case Study", (CERL Technical Report P-85/11, May 1985).

2. Corps of Engineers South Atlantic Division, "Minutes of the Design Quality Forum", (Savannah District, 3 September 1985), page 2.

3. Lloyd S. Finley, "Examination of the Constructability [sic] Review in Government Contracting", (Master's thesis, Purdue University, School of Civil Engineering, 3 August 1984).

APPENDIX 1

ı.

Þ

OPERATIONAL DEFINITIONS

Operational Definitions

Bilateral Modification. A change order on which the Contracting Officer and the contractor have agreed to a price and/or time adjustment.

Change Order (Constructive). An act or failure to act by the Government which has the effect of requiring the contractor to accomplish work different from that required by the existing contract documents.

Change Order (Directed). A contract modification which requires the contractor to accomplish work different from that required by the existing contract documents where the changed work requirements fall under the general scope of the contract and within the physical limits of the construction site.

Changed Work. The contract requirements altered by a contract modification.

Claim. Any written demand by a contractor for money and/or time extensions arising under the contract terms, even if the amount of money or time is unstated.

Constructibility Review. Preconstruction design review conducted to insure the project can be built as designed. The purpose of this review is to identify potential construction conflicts within the design as well as between design and existing facilities.

Contract Change. Any variation from the terms or requirements of a contract. Commonly called "change" or "modification".

Contract Clause. Any general, special, or technical provision, or part thereof, of a contract.

Contract Completion Date. Date established by the contract for completion of all or of specified portions of the work.

Contracting Officer (CO). Any person (usually the District or Division Engineer, or his designated representative) authorized to enter into and administer contracts and to make determinations and findings with respect thereto.

Design Change ("No-fault" changes). A change for which redesign effort is required. A design change materially affects the approved requirements, the basis of design, the existing scope of the contract plans and specifications, or operating capability of the facility. Examples of sources for design changes include changes to design criteria, building codes, safety codes, etc.

Design Deficiency. A design deficiency is defined as an error clearly due to faulty design or a deficiency due to the failure of the designer to properly take into account all visible aspects of site conditions. It is further defined as deficiencies due to the failure of the designer to proofread/coordinate plans and specifications. For the purpose of this study, the benefit of the doubt went to the designer. Any item of change included in a mod which does not meet the above criteria is listed under some classification other than design deficiency. <u>Consequently, the guantity of items labeled as design</u> deficiencies is probably conservative.

Differing (or Unknown) Site Conditions. Refers to (1) subsurface or latent physical conditions at the site differing materially from those indicated in the contract and (2) unknown physical conditions at the site, of an unusual nature, differing materially from those ordinarily encountered and generally recognized as inhering in work of the character provided for in the contract.

Direct Costs. Cost of the materials, supplies, equipment, and prime contractor or sub-contractor work and labor that go into and can be clearly identified with a particular phase of construction.

Impact Costs. Impact costs are those costs related to the indirect effects of a change order. These effects are felt on portions of the project related to, but not directly involved with, the specific items being changed.

Indirect Costs. Indirect costs are those costs that cannot be attributed to a single item or unit of construction work. For this study, indirect costs are limited to administrative and wage costs incurred by the District to process modifications.

Items of Change. Those elements of the original contract being changed by the modification. For this study, all "line items of change" were considered equal, regardless of cost. For example, the removal of asbestos from Project #5 (total cost \$260,000) is one "item of change" as was the administrative correction of a paragraph of a specification even though the specification correction involves no cost to the government. Mods may include one or more of these change items. Like items within a mod are counted as only one "line item". For example, if five housing units involved in the same project each require identical changes to the electrical design, that change, even though repeated five times, is counted as one "line item of change".

Modification. A formal document that alters the contract specifications, delivery point, rate of delivery, contract period, price, quantity, or other contract provisions of an existing contract. Resident Contracting Officer (RCO). The Government representative at the project site who is authorized to execute small dollar amount modifications under the Changes clause of the contract General Provisions. Under certain conditions he may be delegated greater monetary and/or authority under other clauses of the General Provisions.

Resident Engineer. The manager of the field office responsible to the district office for the immediate on-site administration of one or more construction contracts. This individual may or may not be the RCO for the contracts he administers.

Supplemental Agreement. A contract modification for work outside the scope of the contract.

Technical Review. Preconstruction design review conducted to insure appropriate codes, regulations, and design practices are adhered to. Its purpose is to identify errors in the technical aspects of design.

Unilateral Modification. A change order issued by the Contracting Officer in accordance with his determination of an equitable price and time adjustment, but which the contractor does not agree to and does not sign.

Using Service. The Government agency or other entity that will "own" and operate the facility being constructed.

Using Service Review. Preconstruction functional review conducted by the using service, usually under the supervision of DEH. At a minimum, it should be conducted at the concept review (10% design stage) and 65% completion stage of the design process to insure the design meets the requirements of those who are to use the finished product. The purpose of this review is to insure the project scope is understood by all and identify design/construction constraints imposed by the installation such as funding, time, joint occupancy, utilities, mobilization site security, etc.

1

APPENDIX 2

PROJECT SUMMARY SHEETS

IABLE	TITLE	PAGE
29	Summary by Cost of Modification	154
30	Summary by Items of Change	156

SUMMARY OF PROJECTS By Cost of Modification

Data as of 24 Jan 1985

			Size of Project:	
Type Project:	All Three Reviews:	Design Agency:	Less than \$500,000	= 14
MCA = 5	Yes = 21	IN-HOUSE = 5	\$500,000 - \$1,000,000	= 5
OMA = 15	Na = 4	A/E = 20	\$1,000,000 - \$5,000,000) = 6
AFH = 5				

					REASON	FOR NODIFIC	CATION (B	y Cost per	Item of	Change)	1	
0001	TYPE	DEUTENO		HORE	-	050100	01.TC	DECTON		ATUES	: ADJ	PROJ
PRUJ	TTPE	REVIENS		1005	USEK	UESION	3112	ULSION	VHLUE	UTHER		CUSI
NU.	PRUJ	(NO.)	DESIGN	(101)	REQUEST	DEFICIENCY	(LUND	CHANGE	ENGK NG		; ANUUN!	PROMIN
1	084		 A/F	16	******** ***	\$56.690	\$58.911	\$6.719	(\$374)	\$18.443	\$796.000	17.6%
2	ONA	3	A/F		\$11 125	\$0,070	\$0	\$0,717	(\$658)	\$0	1 \$83.000	12.62
3	HCA	3	A/F	33	\$54.477	\$116.444	\$10.941	\$9.087	(\$3,263)	\$47.782	1 \$4.637.000	5.12
4	NNCA	3	A/F	18	\$44.931	\$19 701	\$19,141	\$2.338	\$0	\$0,,, QL	\$993.600	9.62 #
5	NCA	3	A/E	39	\$69.081	\$52.778	\$287.645	\$2,400	\$0	\$0	\$4.665.590	8.81 **
6	ONA	3	I-H (I)	6	\$0	\$390	\$6.475	\$0	\$0	\$17.250	\$224.567	10.7%
7	ONA	3	A/E	6	\$0	\$1,923	\$565	\$324	\$0	\$0	\$340.000	0.8%
8	DHA	3	A/E	3	\$6.606	\$0	(\$7.736)	\$151	\$0	\$150	\$157.500	-0.5%
9	ONA	3	I-H (I)	2	\$0	\$0	\$2.924	\$0	\$0	\$0	\$1.006.260	0.32
10	ONA	3	A/E	1	\$0	\$5.814	\$0	\$0	\$0	\$0	\$79,191	7.3%
11	OMA	3	A/E	7	\$0	\$52.841	\$11,356	(\$13,070)	\$0	\$4.505	\$488,970	11.4%
12	ONA	3	I-H (D)	1 1	\$0	\$0	\$0	(\$100)	\$Ú	\$0	\$71,432	-0.1%
13	ONA	3	A/E	8 8	\$5,331	\$8,827	\$0	\$6,086	\$0	\$7,295	\$209,603	13.1%
14	AFH	3	A/E	25	\$16.199	\$24.110	\$57,540	\$4,477	\$0	\$59.096	\$860.518	18.8%
15	OMA	2	A/E	16	\$189,818	\$3,178	\$187,659	\$43,380	\$0	\$53,373	\$536,300	89.0X ++
16	AFH	0	A/E	5 ;	\$9,765	\$8,469	\$3,778	\$3,132	\$ Ŭ	\$0	\$207,500	12.17
17	AFH	2	A/E	5 1	\$0	\$44,595	\$0	\$0	\$0	\$7,725	1 \$411,373	12.72
18	NCA	3	I-H (D)	12 (\$14,114	\$56,975	\$0	\$7,886	\$0	\$7,000	\$3,000,000	2.9%
19	MCA	3	A/E	28 :	\$4,608	\$47,959	\$97,713	\$11,022	(\$2,516)	\$0	\$4,075,100	3.9% +
20	OMA	3	A/E	10 :	\$10,459	\$2,492	\$6,416	\$0	\$0	\$0	\$267,000	7.32
21	OMA	2	I-H (I)	6 3	\$24,486	\$2.16	\$0	\$0	\$0	\$1,897	\$605,443	4.42
22	AFH	3	A/E	6 1	\$20,123	(\$7,500)	\$0	\$319,250	\$0	\$0	\$1,247,433	26.67 + +
23	AFH	3	A/E	6	\$9,323	\$0	\$0	\$43,775	\$0	\$0	\$505,742	10.5%
24	OMA	3	A/E	2 ;	\$1,959	\$0	\$0	\$0	\$0	\$0	\$173,000	1.1%
25	OMA	3	A/E	1	\$0	\$0	\$0	\$0	\$0	\$2,798	\$254,500	1.1%
				+							+	

 TOTALS
 268
 \$\$492,405
 \$495,892
 \$743,328
 \$446,857
 \$6,811
 \$227,314
 \$\$25,797,622
 9.4%

 PERCENTAGES
 20.4%
 20.6%
 30.8%
 18.5%
 0.3%
 9.4%

(CONTINUED ON NEXT PAGE)

Project Summary (By Cost - Continued)

NOTES:

- 1. Project #5 terminated. Mods shown do not include those initiated because of termination.
- 2. Project #11 award amount adjusted for deleted work.
- 3. Project #14 award amount adjusted for deleted work.
- 4. Project #17 award amount adjusted to reflect user requested created of \$33,804. Credit resulted from termination of work. Bid amount adjusted by subtracting value of work not completed from original contract value.
- 5. Project #21 award amount adjusted for deleted work.
- 5. Project #22 award amount adjusted for deleted work.
- 7. Project #23 award amount increased by amount of supplemental agreement.
- 8. Sums for value engineering mods shown as absolute value of dollar amount.
- 9. Asterisk (+) indicates projects not yet complete:
 - Project # 4 at 96% as of 24 Jan 86.
 - Project #19 at 99% as of 24 Jan 86.
 - Project #22 at 81% as of 24 Jan 86.
- 10. I-H (I) indicates in-house design done by the installation.
- I-H (D) indicates in-house design done by the supporting District office.
- 11. Double asterisk (##) indicates projects that were further adjusted as follows:

There are two individual expense items which have a disproportionate impact on the results shown above. These are the asbestos removal from Project #5 (\$260,194) and criteria change in Project #22 (\$319,250). If these two items are removed, the resulting sums and percentages are as follows:

	: : :	USER REQUEST	DESIGN Deficiency	SITE	DESIGN Change	VALUE Engring	OTHER :
TOTALS	::	\$492,405	\$495,892	\$483,134	\$127,607	\$6,811	\$227,314 :
PERCENTAGES		26.97	27.11	26.47	L 7.01	0.4%	12.47;

The problems encountered in Project #15 (total mods #387,254) also have a disproportionate effect on the results. If this project is deleted, the results are:

	ł	USER	DESIGN	SITE	DESIGN	VALUE		ł
	;	REQUEST	DEFICIENCY	COND	CHANGE	ENGR NG	OTHER	ł
	+-						•••••	ŧ
TOTALS	H	302 ,58 7	\$492,714	\$295,475	\$84,227	\$6,811	\$173,941	ł
PERCENTAGES	ł	22.37	36.32	21.8%	6.2%	0.5%	12.8%	ţ

Two types of expenses make up more than 50% of the amount shown under "others". These are problems with site access (\$43,329) and slowness in reviewing shop drawings (\$51,000).

SUMMARY OF PROJECTS By Item of Change

Data as of 24 Jan 1986

			Size of Project:		
Type Froject: MEA = 5 OMA = 15 AFH = 5	All Three Reviews: Yes = 21 No = 4	Design Agency: IN-HDUSE = 5 A/E = 20	Less than \$500,000 \$500,000 - \$1,000,000 \$1,000,000 - \$5,000,000	= = =	14 5 6

					: REA	SON FOR NODI	FICATION	(By Item	of Change	2)	1 401	0001	
0001	TVOC	0041040		8085		DECIEN	CITE	DESTEN	VALUE	OTHER	PONTR	(05T	
NG	PROJ	(Nor)	DESIGN	(TOT)	PEDIEST	DEFICIENCY	CUND	CHANGE	ENGRING	B THEN	ANDUNT	GROWTH	
	=======	=========	==================	=======================================	+=======		========	222222222	==============		+======================================		=
1	0 M A	1	A/E	16	, , 0	20	7	2	1	3	\$795,000	17.6%	
2	0 H A	3	Η/E	Ь	: 3	1	0	0	1	1	ł \$8 3 ,00 0	12.6%	
3	ĦCA	3	A/E	33	: 11	69	1	9	2	8	1 \$4,637,000	5.1%	
4	HHCA	3	A/E	18	: 6	13	5	2	Û	3	\$893,600	9.67	+
5	MCA	3	A/E	39	: 59	24	8	3	0	3	1 \$4,666,590	8.8%	**
6	OMA	2	I-H (I)	6	: 0	1	4	0	0	3	\$224,567	10.7%	
7	ONA	3	A/E	6	(Ø	2	1	2	0	1	1 \$340,000	0.8%	
8	ŪH4	3	A/E	3	: 3	3	2	1	0	4	\$157,500	-0.5%	
9	OMA	3	I-H (I)	2	i ù	Û.	1	0	0	2	1 \$1,006,260	0.3%	
10	ONA	3	A/E	1	: 0	1	0	0	Û	0	\$79,191	7.32	
11	OMA	3	A/E	7	;	13	4	6	0	3	\$488,970	11.4.	
12	DNA	3	1-H (D)	1	; 0	0	Û	1	Ú	Û	\$71,432	-0.17	
13	OHA	3	A/E	8	4	2	Û	4	Û	1	\$209,603	13.12	
14	AFH	3	A/E	25	: 27	16	111	12	0	8	1 \$860,518	18.8%	
15	OMA	2	A/E	16	: 24	6	34	4	0	4	\$536,300	89.0%	++
15	AFH	Ŭ	A/E	5	; 5	6	16	2	0	3	\$207,500	12.1%	
17	AFH	2	A/E	5	(0	12	0	0	0	2	\$411,373	12.72	
18	HCA	2	1-H (D)	12	: 3	14	Û	3	0	2	\$3,000,000	2.9%	
19	MCA	3	A/E	28	: 2	22	31	2	3	4	1 \$4,075,100	3.92	ŧ
20	OMA	3	A/E	10	6	3	3	0	Û	1	\$267,000	7.3%	
21	OMA	3	1-H (1)	6	; 9	1	0	2	0	4	\$605,443	4.42	
22	AFH	3	A/E	6	6	1	0	1	0	1	1 \$1,247,433	26.6%	4 ++
23	AFH	3	A/E	٥	; 4	0	0	7	0	0	\$505,742	10.5%	
24	AHG	3	A/E	2	; 5	Û	0	1	Q	0	\$173,000	1.17	
25	OMA	3	A⊬E	1	; 0	0	0	0	Ü	i	\$254,500	1.1%	
			TOTALS	268	177	240	228	64	7	62	1\$25,797,622	9.42	
			PERCENT	AGES	22.8	30.87	29.32	8.2%	Ú .9 %	8.02	ł		

(CONTINUED ON NEXT PAGE)

سفد .

.

TOTAL ITENS OF CHANGE: 778

Project Summary (By Item of Change - Continued)

NOTES:

- 1. Project #5 terminated. Mods shown do not include those initiated because of termination.
- 2. Asterisk (*) indicates projects not yet complete:
 - Project # 4 at 96% as of 24 Jan 86.
 - Project #19 at 99% as of 24 Jan 86.
 - Project #22 at 81% as of 24 Jan 86.
- 3. I-H (I) indicates in-house design done by the installation.
- I-H (D) indicates in-house design done by the supporting District office.
- 4. Double asterisk (**) indicates projects that were further adjusted as follows:

There are two individual expense items which have a disproportionate impact on the results shown above. These are the asbestos removal from Project #5 (\$260,194) and criteria change in Project #22 (\$319,250). To maintain consistency, these two items were removed from this analysis as they were for the "cost" analysis:

	;	USER REQUEST	DESIGN DEFICIENCY	SITE Cond	Design Change	VALUE Engring	i Other i
TOTALS PERCENTAGES	-+- 	177 22.81	240 230.9%	227 29.3%	63 8.1%	7 0.9%	62 : 8.0%

The problems encountered in Project #15 (total mods #387,254) also have a disproportionate effect on the cost results. If this project is deleted, the results are:

	ï	USER	DESIGN	SITE	DESIGN	VALUE	ł	
	ł	REQUEST	DEFICIENCY	COND	Change	ENGR'NG	OTHER	
	• • •						+	
TOTALS	ł	153	234	193	59	7	58	
PERCENTAGES	ł	21.75	33.2%	27.42	8.4%	07	8.2%	

Two types of expenses make up more than 50% of the amount shown under "others". These are problems with site access (\$43,329) and slowness in reviewing shop drawings (\$51,600).

APPENDIX 3

CAUSES OF DESIGN DEFICIENCIES

IABLE	TITLE								
31	Summary by Cost	159							
32	Summary by Items of Change	160							

١

REASONS FOR DESIGN DEFICIENCIES By Cost of Modification

Data as of 24 Jan 1985

					DISC	IPLINE IN	WHICH ER	ROR OCCUR	RED (By C	ost)	I TOT	DSGN DEF
					ł						I ANOUNT	AS PCT OF
proj	TYPE	REVIEWS		MODS	1					ADMIN	DUE TO	PROJ MOD
NO.	proj	(No.)	DESIGN	(TOT); CIVIL	NECH	ELEC	STRUCT	ARCH	CHANGE	DSGN ERR	COST
1	OMA	1	A/F	:==== ادا	: \$0	\$13.532	\$11.325	•••••••••• \$0	\$31.833	\$0 \$0	\$56.690	40.47
2	ONA	3	A/F	6	\$0	\$0	\$0	\$0	\$0	\$0	\$0	0.02
3	MCA	3	A/E	33	\$0	\$9.995	\$1.173	\$16.666	\$88.610	\$0	\$116.444	49.52
4	MMCA	3	A/E	18	\$0	\$4.067	\$0	\$0	\$15.634	\$0	\$19.701	22.9%
5	MCA	3	A/E	39	\$5.110	\$24.035	\$0	\$0	\$23.633	\$0	\$52.778	12.87
6	OMA	3	I-H(1)	6	\$0	\$0	\$390	\$0	\$0	\$0	\$390	1.6%
7	OMA	3	A/E	6	: \$0	\$1,923	\$0	\$0	\$0	\$0	\$1,923	68.47
8	ONA	3	A/E	3	: \$0	\$0	\$0	\$0	\$0	\$0	1 \$0	0.02
9	OMA	3	I-H(I)	2	: \$0	\$0	\$0	\$0	\$0	\$0	\$0	0.07
10	OMA	3	A/E	1	1 \$0	\$5,814	\$0	\$0	\$0	\$0	\$5,814	100.02
11	OMA	3	A/E	7	\$0	\$52,841	\$0	\$0	\$0	\$0	\$52,841	47.9%
12	OMA	2	1-H(D)	1	\$0	\$0	\$0	\$0	\$0	\$0	1 \$0	0.07
13	OMA	3	A/E	8	\$0	\$0	\$787	\$0	\$8,040	\$0	\$8,827	32.17
14	AFH	3	A/E	25	1 \$0	\$617	\$110	\$0	\$23,383	\$0	\$24,110	14.9%
15	ONA	2	A/E	16	: \$0	\$442	\$2,336	\$0	\$400	\$0	\$3,178	0.7%
16	AFH	Û	A/E	5	\$7,209	\$0	\$0	\$0	\$1,260	\$0	\$8,469	33.7%
17	AFH	2	A/E	5	1\$13,329	\$0	\$0	\$0	\$31,266	\$0	\$44,595	85.2%
18	MCA	3	I-H(D)	12	1 \$0	\$8,500	\$0	\$12,787	\$35,688	\$0	\$56,975	66.3%
19	MCA	3	A/E	28	\$9,424	\$2,892	\$13,010	\$0	\$22,633	\$0	\$47,959	30.2%
20	OMA	2	A/E	10	\$0	\$0	\$0	\$0	\$2,492	\$0	\$2,492	12.9%
21	OMA	3	I-H(I)	6	\$0	\$0	\$0	\$0	\$206	\$0	\$206	0.82
22	AFH	3	A/E	6	\$0	(\$7,500)	\$0	\$0	\$0	\$0	(\$7,500)	-2.3%
23	AFH	3	A/E	6	\$0	\$0	\$0	\$0	\$0	\$0	\$0	0.02
24	OMA	3	A/E	2	\$0	\$0	\$0	\$Ŭ	\$0	\$0	l \$0	0.0%
25	OMA	3	A/E	1	¦ \$0	\$0	\$0	\$Q	\$ 0	\$0	\$0	0.0%
			TOTALS	268	1\$35 072	\$117 159	\$29.131	\$29 453	\$285 078	••••••	**************************************	

 TOTALS
 260
 \$\$35,072
 \$117,158
 \$29,131
 \$29,453
 \$285,078
 \$0
 \$495,892

 PERCENTAGES
 7.12
 23.62
 5.92
 5.92
 57.52
 0.021
- - -

REASONS FOR DESIGN DEFICIENCIES By Items of Change

Data as of 24 Jan 1985

					; D	ISCIPLINE	IN WHI	CH ERROP	R OCCUR	RED		DS6N EFR
P90.1	TYPE	REUTENO	RECIEN	MODC	i 1	(By	ite#s o	t Lhange	2)	1 4 10 14 14 1	AMOUNT	AS PCT OF
NO.	PROJ	(No)		(1010			51.50	PTDUCT	40.00	ADULIN	DUE TO	PRÚJ MOD
	======	, MU. /	nvin ===================================	11017			ELEG	SIKULI	HKLH	LHANGES;	DSEN ERR	COST
1	OMA	1	4/F	16	, ,	й о	11		5=2355 C	********		===========
;	OMA	3	Δ/F	10	,	0 1 0 1	10	U A	J	U i	\$36,690	40.4%
3	NCA	3	Δ/F	77	1	v 1 0 0	1	10	0 7 7	V i	\$ 9	0.0%
4	MNCA	3	Δ/E	10		v 0 0 7	1	18	رد.	21	¥116,444	49.5%
5	NCA	7	Δ/F	10	1	V 2 1 17	0	i A	۲ ۱۵	1 i	\$19,701	22.93
5	AMG	7	I-H(I)	57		1 10 0 0	1	Ų Å	10	0.1	\$52,778	12.9%
2	GMA	ž	6/E	4	•	v () A 1	1	U A		0 1	\$390	1.5%
à	OMA	3	Δ/F	ں ۲	• •	ν 1 Ο Ο	0 A	0	1	U i	\$1,925	68.4%
ç,	RMA	3	1-H(J)	3 7	, ,	0 0	V A	0	U Q	31	\$0	0.0%
10	ONA	7	Δ/F	1	1 1	0 v 0 1	0	U A	0	0 1	\$0 •= 0+4	0.0%
11	OMA	7	Δ/E	7	, ,	v 1 0 17	0	0	0	Q i	\$5,814	100.0%
12	Ω#A	7	1-4/0 1-4/0	1	ι 1	0 0	ų A	U A	0	9.1	\$52,841	47.9%
13	Ω#Δ	7	1 0 (D/ A/E	1 0	1	υ υ ά ά		Ų ()	0	0 1	\$0 	0.02
14	AFH	হ	A/E	75	1	v v A A	1	U	1	0 1	\$8,827	32.12
15	DMC	, ,		12		्य	5.	Ų •	6	1	\$24,110	14.9%
14	AEU	4		10			4	0	1	0 (\$3,178	9.7%
10	ACU	2	H/E	33		5 0	0	0	1	Ŭ (\$8,469	33.72
17	MER	4	8/E	23		0	Û	9	8	0 :	\$44,595	85.2%
18	ITLH MCA	3	1-H(D)	12	1) 3	0	6	5	0	\$56,975	66.3%
17	ITLH ITMA	5	A/E	28 3		2 4	2	0	12	2 ‡	\$47,959	30.2%
20	UTH	5	A/E	10 ;	() ()	0	0	3	0 1	\$2,492	12.9%
21	UNA	5	I-H([)	6 1	() (0	Ú	1	0	\$206	0.8%
22	AFH	్	A/E	6 1	(1 (0	Û	0	0 1	(\$7,500)	-2.32
23	AFH	3	A/E	6 ;	() ()	0	0	0	0	\$ Û	0.02
24	UMA	3	A/E	2 :	Ć	0	0	0	0	0 :	\$0	0.0X
25	UNA	2	A/E	1	(0	Û	0	Û	0 1	\$ Ŭ	0.0%
			TOTALS PERCENTAB	268 ; ES ;	12 5.0	61 7 25.42	30 12.52	25 10, 47	100 41.77	12 : 5 07 !	\$495,892	*******

TOTAL DESIGN DEF: 240

AFFENDIX 4

•

MODIFICATION COST CORRELATION TABLES

TABLE	IIILE	PAGE
33	Correlation by Review	162
34	Correlation by Project Type	163
35	Correlation by Project Size	164
36	Correlation by Design Agency	165

CORRELATION BY PRECONSTRUCTION REVIEWS (Note Table Adjusted for High-Cost Mods to Projects #5 and 22)

Projects Not Receiving all 3 Reviews:

PROJ NO,	T TPE PROJ	REVIEWS (No.)	DESIGN	MODS (TOT)		REASON USER REQUEST	FOR MODIFIC DESIGN DEFICIENCY	SITE SITE COND	y Cost p er DESIGN CHAN G E	Item of VALUE ENGR NG	Change) OTHER	: :	ADJ Contr Amdunt	PROJ Cost Growth
	OMA	1	A/E	16	:+= 	\$ 0	\$56,690	\$58,911	\$6,719	(\$374)) \$18,443	:+:	\$796,000	17.6%
15	ONA	2	A/E	16	а	189,818	\$3,178	\$187,659	\$43,380	\$ 0	\$53,373	ł	\$536,300	89.02
16	AFH	ð	A/E	5	ł	\$9,765	\$8,469	\$3,778	\$3,132	\$0	\$0	ł	\$207,500	12.12
17	AFH	2	A/E	5	;	\$0	\$44,595	\$0	\$0	\$ Ű	\$7,725	ł	\$411,373	12.7%
												-		
													\$1,951,173	35.6%
									If Project	#15 is	deleted:		\$1,414,873	15.4%

Projects Receiving all 3 Reviews:

PROJ	TYPE	REVIEWS	BEELEN	MODS	;	USER	DESIGN	SITE		VALUE	OTHER	:	ADJ Contr Amerika	PROJ COST
au.		1.040.7	DE510N		- 1	REQUES!			UMHNDE	CNOP NO		; 	880080	GKUWIN
2	ΩĦΔ		4/F	 6	- • !	\$11 125	\$ ()	\$0	\$0	(\$859)	s ù	- T !	\$83 000	12 67
3	MCA		A/F	77		\$54 477	\$116 444	\$10.941	19 087	(\$3, 263)	447 792		\$4 A37 000	5 17
Å	HNCA	1	17 E	18		444 QT1	419 701	\$19 141	\$2 738	(*2,100/ ¢A	\$17,702 \$0	1	000,100,00	0.1%
Ę	MCA	3	Δ/F	10		\$49.091	4 52 779	\$77 451	\$2,300	40 40	40 40	1	44 444 500	7.0%
с ь	DHA	7	ниц Тен (Т)		1	407,001 40	432,779 4700	*1 175	*2,700	*0	417 250	;	*704 5.7	0.24 1/1.74
ט ר		3	1-0 11	· •	,	*0 *0	#1 037	70,7/J *545	4V 4704	¥9 #0	*17,2JU *0	1	#224,307 #710 000	10.7%
5	000 Alima	,	HIL AZE	- -	1	94 101-14	\$1,723	€004 //TT T4/	#324 #154	≱ 9 ≜≏	4U 8450	1	\$340,000	0.84
8	ONA	د -	H/E	ن م	i	\$6,808	\$U	(\$/,/00)	¥101	\$ 0	\$130	i	\$157,500	-0.5%
9	UNA	<u>ن</u>	1-H (1)	2	i	\$ 0	\$ Q	\$2,924	\$ 0	\$ 0	\$ 0	1	\$1,006,260	0.32
10	OMA	3	A/E	1	ł	\$0	\$5,814	\$ 0	\$0	\$ 0	\$0	ł	\$79,191	7.32
11	ONA	3	A/E	7	ł	\$0	\$52,841	\$11,356	(\$13,070)	\$0	\$4,505	ł	\$488,970	11.4%
12	OMA	3	I-H (D)	1	ł	\$0	\$9	\$0	(\$100)	\$ Û	\$0	ł	\$71,432	-0.1%
13	OMA	3	A/E	8	ł	\$5,331	\$8,827	\$0	\$6,086	\$0	\$7,295	1	\$209.605	13.1%
14	AFH	3	A/E	25	ł	\$16,199	\$24,110	\$57,540	\$4,477	\$0	\$59.096	ł	\$860.518	15.82
18	MCA	3	I-H (D)	12	1	\$14,114	\$56,975	\$Ú	\$7,886	\$0	\$7.000	Ì	\$3.000.000	2.92
19	MCA	3	A/E	28	í	\$4.608	\$47,959	\$97.713	\$11.022	(\$2.516)	\$0	:	\$4.075.100	3.92
20	OMA	3	Á/E	10	ł	\$10.459	\$2.492	\$6.415	\$0	\$0	\$0	į	\$267.000	7 37
21	OMA	3	I-H (I)		i	\$74.486	\$206	\$0	\$0	\$Ú	\$1.897	1	\$405 443	A 47
22	AFH	3	4	6	!	\$20.125	(\$7.500)	\$0	to	\$0	\$0.	ì	41 747 433	1 07
77	ΔΕΗ	7	A/E	4	÷	40 707	40 KO	¢0	eAT "75	*v *0	40 40	;	4505 740	1.04
23		7			;	#1 0ED	#0 #0	*0	10,775 40	#0 #0	4U #0	1	\$JVJ,742	10.34
29	OHA	3	H/C	4	÷	+1,737	¥0	¥V • * *	9U 4-0	\$ 0	V4	i	\$173,000	1.14
25	UNA	3	H/E	1	í	\$ 0	\$ 0	\$0	\$ 0	\$0	\$2,798	;	\$254,500	1.1%
												;	***********	============

\$23,846,449 4.7%

CORRELATION BY PROJECT TYPE (Note Table Adjusted for High-Cost Mods to Projects #5 and 22)

AFH	Projects:	
-----	-----------	--

F

•

ŀ

•

ł

n i () ()	alect											
					REASON	FOR MODIFI	CATION (D	y Cost per	ltem of	Change)	1	
											: ADJ	PROJ
PRUJ	1116	REVIEWS	DECION	MUUS	CUSER	DESIGN	SHE	DESIGN	VALUE	DIHER		COST
NU.	PKUJ	(NO.)	DESIGN	(101)	KEROF21	DEFICIENC	Y CUND	CHANGE	ENGK NG		i ANUUNI	GRUWIN
14	AFH	3	A/F	25	\$16.199	\$24.110	\$57.540	\$4.477	\$0	\$59.096	\$860.518	19.81
16	AEH	0	A/E	5	\$9.765	\$8.469	\$3.778	\$3,132	\$0	\$0	\$207.500	12.17
17	AFH	2	A/E	5	\$0	\$44.595	\$0	\$0	\$0	\$7.725	\$411.373	12.7%
22	AFH	3	A/E	6	\$20.123	(\$7.500) \$ 0	\$0	\$0	\$0	\$1.247.433	1.02
23	AFH	3	A/E	6	\$9,323	\$0	\$0	\$43.775	\$0	\$0	\$505.742	10.5%
								, -			93171111111	================
											\$3,232,566	9.4%
MCA Pr	ojects	.:										
											: ADJ	PROJ
FROJ	TYPE	REVIEWS		MODS	USER	DESIGN	SITE	DESIGN	VALUE	OTHER	L CONTR	COST
NO.	PROJ	(No.)	DESIGN	(TOT)	REQUEST	DEFICIENCY	COND	CHANGE	ENGR 'NG		AMOUNT	GROWTH
====== ۲	MCV	=======: ر	11===11=: Δ/Γ	*******	*======================================	*116 AAA	410 0A	22222222222 20 007	187 747	**7 707	+=====================================	5 (Y
4	NHCA	3	6/E	1.0	444 071	\$110,444 \$19 701	*10,741 *10 141	47,007	(\$3,203)	34/,/02 80	1 34,03/,000	J.16 0 14
5	MCA	7	6/E	10 1	\$69.081	417 ,701 45 7,770	\$27 A51	\$2,330 \$2,400	\$0 \$0	*0	1 073,000 1 01 111 590	7.04
18	MEA	3	1-H (D)	12 1	\$14.114	456 975	*21,131	\$7 QQL	*0 \$0	47 000	1 #4,000,010	0.0% 0.0%
19	NCA	3	A/F	28 1	\$4.608	\$47.959	\$9 7 713	¢11 022	(47 514)	\$7,000	\$ \$3,000,000	7 07
• ·					• 1,000	• 17] (31	41/1/10	*11,011	(****0107	••	=======================================	J./A
											\$17.272.290	4.72
ONA Pr	ojects	:										
				ļ							l ADJ	PROJ
PROJ	TYPE	REVIEWS		MODS	USER	DESIGN	SITE	DESIGN	VALUE	OTHER	CONTR	COST
NO.	PROJ	(No.)	DESIGN	(TOT):	REQUEST	DEFICIENCY	COND	Change	ENGR NG		ANOUNT	GROWTH
====== 1		1	A/C	+=====:	*0	#54 40A	*50 011	#/ 710	======== /*77#\	#=========	+=====================================	
2	ΩMA	्र र	H/E A/F	10 1	¥U #11 195	040;0C4	*0 *0	30,/19	(\$3/4)	\$18,445	i \$/96,000	1/.6%
6		ג ז	1-4 (1)	0 i 4 i	*11,123	40 470	4U #1 475	, a u	(\$008)	3U	i \$85,000	12.6%
7	ΩMΔ	J T	Δ/F		₹0	41 017	+0,4/J #5/5	40 8704	\$U #0	\$17,230	i \$224,367	10.7%
Ŕ	ΩNΔ	उ र	Δ/F	ויט	46 LOL	#1,723 #0	1004 147 771	*151	¥Ų #0	¥V #150	1 \$340,000	V.84
9	ONA .	र २	1-H (1)	21	01000 €0	₽V #0	47 074	40 1514	₽U #0	*13V i *6 :	*137,300	-0.34
10	ONA	3	Δ/Ε	1 !	\$0 \$0	45 014	*2,724	40 40	#0 #0	*0 I	*1,000,200	7 79
11	OMA	2	A/F	7 !	₹0 \$0	\$52.841	\$11 354	/\$13 070)	\$0 \$0	\$4 505 F	4100 97 0	1.36
12	OMA	3	I-H (D)	1 !	\$0	11111111 111	\$11,000 \$A	(\$100)	40 40	с. С. С. С. С. Г. Т. Т. Г.	471 475	11.96 -0 17
13	ONA	3	A/E	8 2	\$5,331	\$8.R27	40	\$6.084	40 \$0	₹7 705 I	\$200 LAT	13 (1
15	ONA	2	A/E	16 1	189.818	\$3,178	\$187.659	\$43.380	\$0	\$53.373	\$574 300	10.1K 89.07
20	OMA		A/E	10 1	\$10,459	\$2,497	\$6.416	\$Ú	\$0	100,070 I \$0 \$	\$267.000	7.31
21	OMA	3	1-H (I)	6 1	\$24,486	\$206	\$0	\$ 0	\$ 0	\$1,897	\$605.447	4.47
24	OMA	3	A/E	2 ;	\$1,959	\$0	\$0	\$0	\$0	\$ 0	\$173.000	1.12
25	OHA	3	A/E	1 1	\$0	\$0	\$ 0	\$0	\$0	\$2.798	\$254.500	1.12
						.,	••		••			

\$5,292,766 15.1%

If Project #15 is deleted: \$4,756,466 6.7%

CORRELATION BY PROJECT SIZE (Note Table Adjusted for High-Cost Mods on Projects #5 and 22)

Projects With Adjusted Contract Amounts of Less Than \$500,000

.

PROJ NO.	type Proj	REVIEWS (No.)	DESIGN	MODS (TOT)	REASON User Request	FOR MODIFIC Design Deficiency	ATION (By SITE COND	Cost per DESIGN CHANGE	Item of VALUE ENGR'NG	Change) OTHER	: ADJ Contr Amount	PROJ Cost Growth
	 Ω#Δ		A/F		\$11 125	•••••• \$0	t 0	±0	(\$658)	\$ 0	\$83.000	12.6%
6		1	T-H (D) 6.3	\$0	\$390	\$6.475	\$0	\$0 \$0	\$17.250	\$224.567	10.7%
7	OMA	7	A/F	م	sù	\$1.923	\$565	\$324	\$0	\$0	\$340.000	0.8%
8	ONA	3	A E	3	\$6.606	3 0	(\$7,736)	\$151	\$Ú	\$150	\$157,500	-0.5%
10	OMA	3	A/E	1	\$Ú	\$5,814	\$0	\$ 0	\$0	\$0	\$79,191	7.32
11	OMA	2	A/E	1	\$0	\$52,841	\$11,356	(\$13,070)	\$0	\$4,505	\$488,970	11.4%
12	OMA	3	1-H (D	1	\$ Ú	\$0	\$0	(\$100)	\$0	\$0	\$71,432	-0.1%
13	ONA	3	A/E	8	\$5,331	\$8,827	\$0	\$6,086	\$ 0	\$7,295	\$209,603	13.12
16	AFH	0	A/E	5 5	\$9,765	\$8,469	\$3,778	\$3,132	\$0	\$0	\$207,500	12.1%
17	AFH	2	A/E	5	\$0	\$44,595	\$0	\$0	\$0	\$7,725	\$411,373	12.7%
20	OMA	3	A/E	10	\$10,459	\$2,492	\$6,416	\$ 0	\$0	\$0	\$267,000	7.32
24	ONA	2	A/E	2	\$1,959	\$0	\$0	\$Ŭ	\$0	\$0	\$173,000	1.1%
25	CHA	3	A/E	i	\$0	\$0	\$0	\$0	\$0	\$2,798	\$254,500	1.1%
											*===========	
											\$2,967,636	7.7%
Projec	te Wit	n Adjust	ed Conti	ract An	iounts Be	tween \$500,	000 and \$	1,000,000	•			
											: ADJ	proj
PROJ	TYPE	REVIEWS		MODS	USER	DESIGN	SITE	DESIGN	VALUE	OTHER	CONTR	COST
NO.	PROJ	(No.)	DESIGN	(TOT) :	REQUEST	DEFICIENCY	COND	Change	ENGRING		AMOUNT	GROWTH
	======	;312122; ,	=======	******							+===#=================================	
1	UNA	1	A/E	16 :	\$U	\$36,690	\$58,911	\$6,/19	(\$5/4)	\$18,443	\$796,000	17.6%
	ППЦА АСИ) 7	H/E	18 1	- 344,731 - 417 100	\$17,/UI	117,141 457 540	≥2,338 AA ATT	¥U #0	30 - 450 - 00/ - 1	1 1873,800 1 10/0 510	7.0%
14	HER OMA	د. -	R/E	23 1	*100 010	\$24,110 \$7 170	107,340	44 ,411	\$U #0	\$57,096 F	\$650,316	18.84
21	UNH OMA	4 7	н/с т.н./т.	10	+107,010	₹3,170 #707	\$10/ ₁ 037	¥43,380	¥0 ¢∆	*1 007	*U30,300	87.04
41 57		J -	1-11 (1)	0 1 2 1	¥24,400	₹200 ¢0	3U #0	10 11	90 40	₹1,877 i	1 1003,443	4.46
53	нгп	J	HIL	0 1	**,320	\$ 0	¥0	\$43,773	٩V	₽ 0 -	· •J0J,742	IV.J.
											el 10 7 403	22 57
								If Project	F#15.1c	dalatad.	47 141 703	17 07
Projer	4 c 1414	h Adimeti	ad Contr	ort Ca	nunte flui	ar ti 000 0	66	IT TTUJEC		deleteu.	*310011003	14.0%
110)20	(3 #10	0 NGJG351	eu cunci			*******	vv.				ΔD.1	PP0.1
220.1	TYPE	REVIEWS		H005 !	HSER	DESIGN	SITE	DESTON		OTHER	, 100 91001 -	COST
NO	PROJ	(No)	DESIGN	(TOT) !	REDIEST	DEFICIENCY	COND	CHANGE	ENGR 'NG	UTHEN		SROUTH
======	=======	==========		======	-======================================			222222223			+======================================	
3	MEA	3	A/E	33	\$54.477	\$116.444	\$10.941	\$9.087	(\$3,263)	\$47,782	\$4.637.0 00	5.1%
Š	MCA	3	A/E	39 1	\$69.081	\$52.778	\$27.451	\$2.400	\$0	\$0	\$4.666.590	3.32
9	OMA	3	1-H (I)	2	\$0	\$0	\$2.924	\$0	\$0	\$0	\$1.006.260	0.32
18	MCA	- 3	I-H (D)	12 1	\$14.114	\$56.975	\$0	\$7,886	\$0	\$7.000	\$3,000.000	2.91
19	NCA	3	A/E	28	\$4.608	\$47.959	\$97.713	\$11.022	(\$2,516)	\$0	\$4.075.100	3.92
22	AFH	3	A/E	6	\$20.123	(\$7.500)	\$0	\$0	\$0	\$0	\$1,247.433	1.01
		-	-	-	,	,	. •					

164

. .**..**.

\$18,632,383 3.5%

.....

CORRELATION BY DESIGN (IN-HOUSE VS. CONTRACTED A/E) (Note Table Adjusted for High-Cost Mods on Projects #5 and 22)

A/E Designed Projects:

	esiyneu	rruject	2.					C	14 1	Ch	,			
					; REASUN I	FOR MODIFIC	ATION (By	Lost per	ICEM OF	Lnange/	1	40.1	0003	
										01050	i	HUJ	PRUJ	
PROJ	TYPE	REVIEWS		MODS	USER	DESIGN	STIE	DESIGN	VALUE	UTHER	i	LUNIK	0051	
NC.	PROJ	(No.)	DESIGN	(101)	REQUEST	DEFICIENCY	CUND	CHANGE	ENER NG		; 	ARUUNI	6KUN(H	
===== t		*********	2222222: Δ/F	:====: 16	+======== ! \$ 0	\$56.690	\$58.911	\$6.719	(\$374)	\$18,443	• • : 	\$796.000	17.6%	
2		3	6/E		: ••• : •11 125	\$00,070	\$0	\$0	(\$658)	\$0	÷	\$83.000	12.62	
7	MCQ.	ן ז	Δ/F	77	: 454 4 77	£116 AAA	\$10.941	\$9.087	(\$3.263)	\$47.782	÷	\$4.637.000	5.17	
Ă	MMCA	ž	Δ/F	18	\$44.931	\$19.701	\$19,141	\$2.338	\$0	\$0	÷	\$893.600	9.6%	
, ç	MC A	7	6/E	79	\$69.081	\$52.778	\$27.451	\$7,400	\$0	\$ 0	Ì	\$4.666.590	3.32	
7	ONA	3	Δ/F		: \$ 0.,001	\$1 923	\$545	\$324	\$0	\$Đ	1	\$340.000	0.8%	
Ŕ	DMA	उ र	۵/F	7	\$6.606	\$0	(\$7.736)	\$151	\$0	\$150	÷	\$157.500	-0.52	
10		3	Δ/F	Í	! \$0,000	45 RIA	\$0	\$0	\$0	\$0		\$79,191	7.32	
11	ONA	-	6/5	7	. ≠0 ! \$0	\$52 8A1	\$11 354	(\$13 070)	\$ 0	\$4 505	į	\$499.970	11.42	
11		, T	Δ/Ε	, 0	•	402,041 49 877	411,030 40	4410,0101	40 40	\$7 295	÷	\$209 603	13 12	
1.3	1000 1000	7	A/E	25	· *12 100	#0,017 #24 110	457 540	44 477	40 40	450 AQL	1	4940 519	19 97	
15		د د	A/C	1.	4101177	427,110	4107 LEO	41,711	4V 40	457 777	;	4574 760	DQ 07	
14	VUH VCU	4	H/C	10	+ #107,010	*0 *10	*10/ 10J7	**0100V *7 170	*0	400,070	;	\$207 500	15 15	
17	MER	v 2	H/C A/C		ι ≱7 ₁ /QJ ι ≱Λ	\$44 505	*3,110 #0	*0,132	\$V #0	+7 705	1	4411 777	12.14	
17	HEA	4	H/E	2	i ≱V. I ≜I (AD	\$44,070	V4 717 704	¥U	140 E1()	•1,123 **	1	\$411,3/3	12.14	
14	ILA .	2	H/E	28	i ≱ 4,608	\$47,939	\$97,713	\$11,022	(\$2,516)	30	1	\$4,0/3,100	5.44	
20	OMA	3	A/E	10	\$10,459	\$2,492	\$6,416	\$0	\$ 0	\$0	ł	\$257,000	1.5%	
22	AFH	3	A/E	6	\$20,123	(\$7,500)	\$0	\$0	\$0	\$0	ł	\$1,247,433	1.02	
23	AFH	3	A/E	ò	\$9,323	\$0	\$0	\$43,775	\$0	\$0	ł	\$505,742	10.52	
24	OMA	3	A/E	2	\$1,959	\$0	\$0	\$0	\$0	\$0	ł	\$173,000	1.17	
25	OMA	3	A/E	1	\$0	\$0	\$0	\$0	\$0	\$2,798	ł	\$254,500	1.12	

\$20,889,920 8.02

If Project #15 is deleted: \$20,353,620 5.9%

Designed by In-House, Government Engineers:

PROJ ND.	i ype Proj	REVIEWS (No.)	DES	16N	NDDS (Tot	 ((USER Request	DESIGN DEFICIENCY	SITE Cond	DESIGN Change	VALUE Engr'ng	OTHER	;;;;;;;;;;;;;;;;;;;;;;;;;;;;;;;;;;;;;;;	ADJ Contr Amount	PROJ Cost Growth	_
6	OMA	3	I-H	(1)	6	1	\$0	\$390	\$6,475	\$0	\$0	\$17,250	1	\$224,567	10.7%	
9	OMA	3	I-H	(1)	2	ł	\$0	\$ Ū	\$2,924	\$0	\$0	\$0	ł	\$1,006,260	0.32	
12	ONA	3	1-H	(0)	1	ł	\$0	\$0	\$0	(\$100)	\$0	\$0	ł	\$71,432	-0.17	
19	MCA	2	I-H	(D)	12	ł	\$14,114	\$56,975	\$0	\$7,886	\$0	\$7,000	ł	\$3,000,000	2.9%	
21	OMA	3	I-H	(I)	6	ł	\$24,486	\$206	\$0	\$0	\$0	\$1,897	ł	\$605,443	4.4%	
													2		********	:
														\$4,907,702	2.8%	

NOTE: I-H (1) indicates design by installation. I-H (D) indicates design by supporting Corps District.

APPENDIX 5

.

INDIRECT COSTS FOR ENGINEERING, DESIGN, AND OVERHEAD By project

NOD COST COMPARISON District Costs for Engineering, Design, and Overhead

				DIRECT	COST	OF MODIF	ICATION	ADMIN		ADMIN		
	DIRECT	TOTAL		PROJ	ENGR, DES	SIGN, AND	OVERHEAD	COST	ł	COST	ł	ADJUSTED
FROJ	MOD	ITEMS	HODS	COST :				E PER MOD	(PE	R ITEM	ł	CONTRACT
NŪ.	COST	OF CHNG	(TOT)	GROWTH :	E&D	OVERHD	TOTAL	(E&D,OVRHD)	0	F CHNG	ł	AMOUNT
		********	*=====	=========+	********			+2==========	+==		=+=	***********
1	\$140,390	43	16	17.6%	\$4,102	\$5 0	\$4,152	\$260	ł.	\$9 7	ł	\$795,000
2	\$10,467	6	6	12.67;	\$286	\$41	\$327	; \$54	1	\$54	ł	\$8 3,000
3	\$235,468	100	33	5.1%	\$19,983	\$2,956	\$22,939	\$695	ł –	\$229	÷	\$4,637,000
4	\$86,111	29	18	9.6%;	\$4,925	\$676	\$5,601	\$311	1	\$193	ł	\$893,600
5	\$411,844	97	39	8.871	\$24,330	\$2,424	\$26,754	\$686	1	\$276	ł	\$4,666,590
6	\$24,115	8	6	10.7%	\$968	\$156	\$1,124	\$187	1	\$140	1	\$224,567
7	\$2,811	6	6	0.8%	\$1,580	\$236	\$1,816	\$303	I.	\$303	ł	\$340,000
8	(\$829)	13	3	-0.5%	\$713	\$108	\$821	\$274	1	\$63	ł	\$157,500
9	\$2,924	3	2	0.3%	\$670	\$98	\$768	\$384	1	\$256	1	\$1,006,260
10	\$5,814	1	1	7.3%	\$315	\$48	\$363	\$363	1	\$363	ł	\$79,191
11	\$55,632	26	7	11.4%	\$2,209	\$3 33	\$2,543	\$363	1	¥98	ł	\$488,9 70
12	(\$100)	1	1	-0.12;	\$437	\$71	\$509	\$509	ţ.	\$509	ł	\$71,432
13	\$27,539	11	8	13.171	\$844	\$128	\$972	\$121	;	\$88	1	\$209,603
14	\$161,420	179	25	18.7%;	\$3,509	\$503	\$4,012	\$160	1	\$22	ł	\$860,518
15	\$477,408	72	16	89.0%	\$2,363	\$370	\$2,733	\$171	ł	\$38	ł	\$536,300
16	\$25,145	31	5	12.17:	\$893	\$131	\$1,023	\$205	ł	\$33	ł	\$207,500
17	\$52,320	14	5	11.8%;	\$2,156	\$269	\$2,425	\$485	:	\$173	;	\$411,373
18	\$85,975	22	12	2.9%	\$13,189	\$2,118	\$15,308	\$1,276	1	\$696	ł	\$3,000,000
19	\$158,786	64	28	3.9%	\$17,252	\$2,806	\$20,058	\$716	ł	\$313	ł	\$4,075,100
20	\$19,367	13	10	7.37:	\$1,508	\$245	\$1,753	\$0	(\$0	ſ	\$267,000
21	\$26,589	16	6	4.4%	\$2,888	\$482	\$3,370	\$562	1	\$211	1	\$605,443
22	\$331,873	9	6	26.67:	\$5,035	\$784	\$5,819	: \$0	1	\$647	ł	\$1,247,433
23	\$53,098	11	6	10.5%	\$1,296	\$202	\$1,498	\$250	ŧ.	\$136	ł	\$505,742
24	\$1,959	6	2	1.12	\$817	\$133	\$9 50	\$475	ł.	\$158	;	\$173,000
25	\$2,798	1	ł	1.17	\$1,533	\$193	\$1,726	\$1,726	:	\$1,726	ł	\$254,500
T01:	\$2,398,924	782	268	9,421	\$113,802	\$15,561	\$129,362	: \$483 (Avg)	•• 	\$165 (Avg)	-+-	\$25,797,622

AFFENDIX 6

REVIEW SYSTEM QUESTIONNAIRE



Modification Study Interview Questionnaire

INTRODUCTION :

Yes

Date of Interview May - June 1985

This questionnaire is part of a study of modifications to military construction contracts within the **contraction** District in general and Fort **Construction** in particular. An underlying assumption of the study is that projects which receive careful pre-construction reviews have lower cost growth due to modifications than projects which are not reviewed. One of the stated objectives of the study is to evaluate the effectiveness of the District's review process. The purpose of this questionnaire is to obtain data to assist in making that evaluation.

1. Please identify your current job description, branch, and section.

Engr Division (29); Project Managers (10); Constr Reps (6); Res Off (5);

Have you ever received any training in how to conduct either technical or constructability reviews?

3. If yes, describe which, how and when you received that training.

Formal courses/seminars (19)); On-the-job training (6)

4. Circle the disciplines in which your training and/or experience gives you the expertise to conduct technical or constructability reviews ,more than one answer acceptable):

a .	civil (25)	
b.	Architectural (16)	I)
c.	Mechanical (16)	



31

 Circle the disciplines in which you normally conduct technical or constructability reviews (more than one answer acceptable);

a .	Civil (22)	\sim
Ъ.	Architectura	1 (16)
c.	Mechanical	(16)



Assume that the projects referred to in the next three questions are within your primary area or discipline of expertise.

6. How much time do you need to properly review the following type projects? (Consider only the time it takes to conduct the review.)

Small (less than \$500,000) projects - not complex

a .	less than 1 working	day (14)	4.	6 - 10 working days (Q
Ъ.	2 - 3 working days		8,	11 - 15 working days	¢
c.	4 - 5 working days	(5)	Ε.	over 15 working days	C

Medium (\$500,000 - \$1,000,000)	projects - madium complexity
a. less than 1 working day. (3)	d. 6 - 10 working days (5)
b. 2 - 3 working days (18)	e. 11 - 15 working days (0)
c. 4 - 5 working days (24)	f. over 15 working days (1)
Large (greater than \$1,000,000)) projects - very complex
a. less than 1 working day (0)	d. 6 - 10 working days (22)
b. 2 - 3 working days (7)	e. 11 - 15 working days (5)
c. 4 - 5 working days (15)	f. over 15 working days (2)
 Given (your) (the people under your sup	pervision's) normal daily work
requirements, how long prior to when com	ments are due do (you) (they) need
in order to perform a thorough technical	. or constructability review for:
Small (less than \$500,000) pro	jects - not complex
a. 1 - 5 working days (17)	d. 16 - 20 working days ((4))
b. 6 - 10 working days (19)	e. 21 - 25 working days ((6))
c. 11 - 15 working days (5)	f. over 25 working days (1))

Medium (\$500,000 - \$1,000,000) projects - medium complexity

a. 1 - 5 working days (5) b. 6 - 10 working days (20) c. 11 - 15 working days (15)	d. 16 - 20 working days e. 21 - 25 working days f. over 25 working days	(3) (6) (3)
Large (greater than \$1,000,000)	projects - very complex	
\sim		\mathbf{C}

a. 1 - 5 working days (1)	d. 16 - 20 working days (12)
b. 6 - 10 working days (7)	e. 21 - 25 working days 🚯
c. 11 - 15 working days (16)	f. over 25 working days (8)

 On the average, how long prior to when comments are due are (you) (the people under your supervision) given to do reviews for:

Small (less than \$500,000) projects - not complex

a. 1 - 5 working days (22)	d. 16 - 20 working days (3)
b. 6 - 10 working days (14)	e. 21 - 25 working days (1)
c. 11 - 15 working days (8)	f. over 25 working days (0)
Medium (\$500,000 - \$1,000,	000) projects - medium complexity

a. 1 - 5 working days	(11)	d. 16 - 20 working days (()	3)
b. 6 - 10 working days	723	e. 21 - 25 working days (()	1)
c. 11 - 15 working days	(01)	f. over 25 working days (()	2)

Large (greater than \$1,000,000) projects - very complex

a. 1 - 5 working days (7)	d. 16 - 20 working days 🕢
b. 6 - 10 working days (18)	e. 21 - 25 working days (3)
c. 11 - 15 working days (14)	f. over 25 working days (2)

9. What percentage of your time during a "typical week" is spent on each of the following activities (circle "0%" for those activities that do not apply to you): Technical/Constructability Reviews: a. 0% b. 1%-5% c. 6%-10% d. 11%-20% e. 21%-30% f. 31%-50% g. Over 50% Engineering Design: a. 0% b. 18-5% c. 68-10% d. 118-20% e. 218-30% f. 318-50% g. over 50% Contract/Design Modifications: a. 25 b. 13-55 c. 63-105 d. 115-205 s. 213-305 f. 315-505 g. over 505 (11) (18) (15) (6) (11) (5) (0) Field Inspections: a. 0% b. 18-5% c. 68-10% d. 118-20% e. 218-30% f. 318-50% g. Over 50% General Office Correspondence/Admin: a. 0% b. 19-5% c. 68-10% d. 118-20% e. 219-30% f. 318-50% g. Over 50% Meetings: a. 0% b. 18-5% c. 68-10% d. 118-20% e. 218-30% f. 318-50% g. over 50% Shop Drawing Review: a. 0% b. 18-5% c. 68-10% d. 118-20% e. 218-30% f. 318-50% g. Over 50% Coordination/Communication Between Corps, Contractor, DEH, A/E: a. 0% b. 1%-5% c. 6%-10% d. 11%-20% e. 21%-30% f. 31%-50% g. Over 50% Other (specify)_ a. 0% b. 18-5% c. 68-10% d 118-20% e. 218-30% f. 318-50% g. Over 50% 10. Prioritize the items shown in the previous question. Prioritize only those items that pertain to your work. You are asked to answer this question in two ways. First, show the priority that you feel SHOULD be assigned to each item. Second, show the priority that, based on your experience in the second District, you feel "the system" currently assigns to each item. Use "1" for the top priority with i... reasing numbers indicating items of lesser priority. Enter "0" or "N/A" for those items that do not apply to you:

Activity	REVIEWS 45:	Should Be	As Is	
a. Technical/Constructability Review b. Engineering Design c. Contract/Design Modifications	Top Third: Middle Third: Bottom Third:	<u>14</u> 20 · 10	<u>12</u> <u>19</u> <u>15</u>	
 a. Field Inspections e. General Office Correspondence/Adm f. Meetings g. Shop Drawing Review 	in :			·····
h. Coordination/Communication Betwee: Contractor, DEH, A/E Other (specify	n Corps,			

NOTE: Only time spent un modifications collated.

```
11. For what percentage of the reviews that (you) (the people under your
           supervision) have conducted did (you) (they) visit the site as part of
           the review process:
                 b. 0 (13)
                          25 c. 25 50 d. 50
(3) ((3)) ((4)
                                                      75% e. 75 - 100%
             (19)
           a.
                                                  ((4))
                                                                ((6))
                          (I) (the people under my supervision) do not visit sites as
      12. The reason (s)
          part of the design review process are (more than one answer possible):
          a. Insufficient funds budgeted
          b. Competing work requirements (32)
          c. I do not feel it is necessary ((5))
          d. Insufficient time due to short Suspense dates (28)
e. Other (specify) ((9))
          e. Other (specify)_
                                      District
      13. How familiar are you with Regulation 1110-1-1, "Review of Plans and
          Specifications"?
          a. Have never seen it (24)
          c. Have seen a copy but never read it (10)
c. Have read it, but not thoroughly familiar with contents (14)
          d. Thoroughly familiar with contents
                                                    ((7))
      14. Sufficient time is available to do thorough technical or constructability
          reviews:
                                               c. Sometimes d. Rarely
(27) (10)
                      b. Most of the time
(13)
          a. Alieys
                                                                               e. Neyar
               (11)
                                                                                   ((3))
     15. Identify the reason(s) for those occasions when sufficient time for reviews
          IS NOT available. (More than one answer possible):
          a. Insufficient funds budgeted (14)
b. Competing work requirements (37)
         c. Not enough information provided (9) (all from Engineering Division)
d. Incomplete plans/specifications (15)
         e. Short suspense dates
f. Other (specify) ((5))
                                      (36)
All answers :
Always=124
Hostly=32
               Large (over $1,000,000):
          a. Always b. Most of the time c. Sometimes d. Rarely e. Never
```



26. Thorough, detailed reviews save the government money:

a. Always b. yest of the time c. Sometimes d. Barely e. Mayer (16) (28) (11) (20) (0) 27. Given all of the daily tasks (I) (the people under my supervision) must accomplish, the emphasis placed by "the system" (the District, DEH, Customer, etc.) on reviews is: a. 200 much b. About right c. Not enough (6)) (29) 28. For designs involving renovation work (as opposed to new work), site visits are routinely conducted (by me) (the people under my supervision) as part of the review process: a. Livays b. yost of the time c. Sectimes d. Rawely (4) (13) (6) (16) e. Nesser (16)) (15)) 29. For designs involving new work (as opposed to renovation work), site visits are routinely conducted (by me) (the people under my supervision) as part of the review process: a. Always b. Most of the time c. Sometimes d. Rarely e. Nover ((3)) (9) (13) (13) (12) (18) 30. Comments from previous reviews are readily available to individuals conducting subsequent reviews: a. Always b. Most of the time c. Sometimes d. Barely (16)) (16)) (13)) (13)) ((8)) 19:00 T (′0:) 31. All variad review comments are incorporated into the final designs a. Alinays b. Most of the time c. Sometimes d. Realy (3)Never (22) ((6)) ((0)) (23) 32. Individuals conducting reviews and providing comments are given feedback as to which of their comments are included in design changes and which are not: a. Always b. Most of the time c. Sometimes d. Rarely e. Never (18)) (6) (17)) (11) (2)) 33. (I) (the people under my supervision) have checklists for use as reference when conducting reviews: e. Never

a. Always b. Most of the time c. Sometimes d. Barely e. Nowar (4) (8) (12) (8) (19)) Answer the following question on the basis of an absolute standard of quality. DO NOT qualify your answer.

34. The quality of the reviews (I) (the people under my supervision) conduct are as thorough and detailed as I feel the complexity of the project warrants:

. Always	b. Most of the time	c. Sometimes	d. Rarely	e. Newer
((5))	((25))	(20)	((0))	((0))
	\bigcirc	\bigcirc	\bigcirc	\bigcirc

a

1 . . .

- 35. How long have you worked for the Corps of Engineers (either in a District or DER position)?
 - a. less than 2 rrs (5) b. 2 - 5 yrs (5) c. 5 - 10 yrs (16) d. 10 - 20 yrs (20) e. more than 20 yrs (10)

36. How long have you worked for the District?

- 37. How much time have you spent as a design engineer (not limited to government work)?
 - a. 0 (14) b. less than two years (5) c. 2 - 5 yrs (8) d. 5 - 10 yrs (10) e. 10 - 20 yrs (8) f. more than 20 yes (11)
- 38. How much time have you spent in construction (not limited to government work)?
 - a. 0 (10) b. less than two years (21) c. 2 - 5 yrs (6) d. 5 - 10 yrs (3) e. 10 - 20 yrs (6) f. more than 20 yrs (10)
- 39. How much time have you spent in management/supervision of engineering or construction (not limited to government work)?
 - a. 0 (22) b. less than two years (4) c. 2 - 5 yrs (10) d. 5 - 10 yrs (7) e. 10 - 20 yrs (3) f. more than 20 yrs (4)

 \frown

- 40. Please identify the highest academic level and all degrees you have earned. If you hold a degree (or degrees), please circle the type of degree (i.e. associates, B.S., M.A.) and show the discipline in which it was received.
 - a. High School Diploma (4) b. College - no degree (2) c. Associates Degree (2) d. Bachelors Degree (B.S.) (B.A.) (41)) e. Masters Degree (M.S.) (M.A.) (9) f. Other (0)
- 41. Do you hold a certificate as an Engineer in Training (EIT)?

yes (7)'						no
(NOTE :	Does not	include	those	who	are	2150	
	licensed	profess	ional	engin	heer	s)	





42.	Are you a licensed Professional Engineer (F	E)? (NOTE: Includes all licensed pro-
	fessionals, such as architects, geo	logists, landscape architects, etz.)
	yes_(())	no <u>(25)</u>
43.	Are you a certified Engineer Technician?	
	yes	no <u>(55)</u>

BIBLIDGRAPHY

BIBLIOGRAPHY

BOOKS:

- Brown, Foster L.; Amos, Jimmy R.; and Mink, Oscar G. <u>Statistical</u> Concepts: A Basic Program. <u>2nd_ed.</u> <u>New York:</u> <u>Harper and Row</u>, 1975.
- Guilford, J.P. <u>Eundamental Statistics in Psychology and Education</u>. 4th ed. San Francisco: McGraw-Hill Book Company, 1965.
- Halpin, Daniel W. and Woodhead, Ronald W. <u>Construction Management</u>. New York: John Wiley and Sons, 1980.
- Rutman, Leonard. <u>Evaluation_Research_Methods</u>. Beverly Hills: Sage Publications, Inc., 1977.

Simon, Julian L. <u>Basic_Research_Methods_in_Social_Science</u>. New York: Random House, 1969.

GOVERNMENT_PUBLICATIONS:

- U.S. Army Engineer Pamphlet 415-1-2. <u>Modifications and Claims</u> Guide. <u>Washington</u>, D.C.: Office of the Chief of Engineers, 1 October 1976.
- U.S. Army Engineer Pamphlet 415-1-3. <u>Modification Impact Evaluation</u> Guide, <u>Washington</u>, D.C.: <u>Office of the Chief of Engineers</u> 2 July 1979.
- U.S. Army Engineer Regulation 415-1-11. <u>Biddability</u>. <u>Constructibility, and Operability</u>. Washington, D.C.: Office of the Chief of Engineers, 30 January 1981.
- U.S. Army Engineer Regulation 1110-1-8. <u>Reguired Visits to</u> <u>Construction Sites by Design Personnel</u>. Washington, D.C.: Office of the Chief of Engineers, 23 May 1980.
- U.S. Army Engineer Regulation 1110-2-1200. <u>Plans and</u> <u>Specifications</u>. Washington, D.C.: Office of the Chief of Engineers, 12 June 1972. With change 1, dated 18 September 1972.
- U.S. Army Engineer Regulation 1110-345-51. <u>Cost Targets (Military)</u>. Washington, D.C.: Office of the Chief of Engineers, 21 October 1980.

- U.S. Army Engineer Regulation 1110-345-100. <u>Design Policy for</u> <u>Military Construction</u>. Washington, D.C.: Office of the Chief of Engineers, 14 December 1973. With change 4, dated 28 June 1985.
- U.S. Army Engineer Regulation 1180-1-1. Engineer Contract Instructions. Section 26, "Contract Modifications." Washington D.C.: Office of the Chief of Engineers, 1 July 1980. With change 1, dated 1 June 1981.
- U.S. Army Engineer Regulation 1180-1-6. <u>Quality Management</u>. Washington, D.C.: Office of the Chief of Engineers, 24 April 1978. With change 1, dated 17 July 1978.
- District Regulation 415-2-1. <u>Resident Engineers Manual</u>. 1 August 1978. With change 4, dated 30 April 1982.
- District Regulation 1110-1-1. <u>Review of Plans and Specifications</u>. 1 November 1984.
- U.S. Department of the Army. <u>The Posture of the Army and Department</u> <u>of the Army Budget Estimate for Fiscal Year 1986</u>. Washington, D.C.: U.S. Government Printing Office, 1985.
- U.S. Department of the Army. <u>Construction Contract Negotiating</u> Guide.<u>Washington, D.C.: Office of the Chief of Engineers</u>. Fiscal Year 77 Edition.
- U.S. Department of Defense. <u>Annual Report to the Congress: Report</u> of the Secretary of Defense on the EY 1986 Budget. FY 1987 <u>Authorization Reguest and FY 1986-90 Defense Programs</u>. Washington, D.C.: U.S. Government Printing Office, 4 February 1985.

TECHNICAL REPORTS:

- Logistics Management Institute. <u>Military Versus Private Sector</u> <u>Construction Costs</u>. Report prepared by William B. Moore and Joseph S. Domin. Bethesda, Maryland: Logistics Management Institute, March 1985.
- Texas A&M Research Foundation. <u>Systems Analysis of Corps A/E</u> <u>Design Engineering</u>. Prepared under contract for the Directorate of Military Construction, Office of the Chief of Engineers. College Station, Texas: College of Engineering, Texas A&M University, June 1969.

- Texas A&M Research Foundation. <u>A_Systems_Approach_to_Design_and</u> <u>Construction_for_the_Corps_of_Engineers</u>. TR 68-041. Prepared under contract for the Directorate of Military Construction, Office of the Chief of Engineers. College Station, Texas: College of Engineering, Texas A&M University, May 1968.
- U.S. Army Construction Engineering Research Laboratory (CERL). <u>Nays</u> <u>to Improve Construction Contract Modification Processing: USAFEA</u> <u>Korea Case Study</u>. Technical Report P-85/11 and RAMP Information Paper No. 35. Champaign, Illinois: USA-CERL, May 1985.
- U.S. Army Construction Engineering Research Laboratory (CERL). <u>Automated Review Management System (ARMS)</u>. Preliminary draft prepared by Jeffrey Kirby and Don Hicks. Champaign, Illinois: USA-CERL, 3 May 1985.
- U.S. Army Construction Engineering Research Laboratory (CERL). <u>Modifications Processing Procedures: A Generalized Stochastic</u> <u>Network Model</u>. Technical Report P-82, prepared by M.J. O'Connor. Champaign, Illinois: USA-CERL, August 1977.
- U.S. Army Construction Engineering Research Laboratory (CERL). <u>Military Construction Contract Management</u>. Technical Report P-76, prepared by Ronald L. Foster. Champaign, Illinois: USA-CERL, November 1976.
- U.S. Army Corps of Engineers. <u>Report_of_the_Green_Ribbon_Panel_on</u> <u>U.S. Army_Corps_of_Engineers_Support_to_Army_Installation</u> <u>Commanders</u>. Washington, D.C.: Office of the Chief of Engineers, March 1985.
- U.S. Army Corps of Engineers. <u>Report of the Blue Ribbon Panel on</u> <u>Management of Construction Quality in the U.S. Army Corps of</u> <u>Engineers</u>. Washington, D.C.: Office of the Chief of Engineers, March 1983.

MAGAZINE/JOURNAL_ARTICLES:

- Diekmann, James E. and Nelson, Mark C. "Construction Claims: Frequency and Severity." <u>Journal of Construction Engineering</u> and Management. Vol. 111, No. 1. Proceedings of the American Society of Civil Engineers. March 1985, p. 74.
- Kagan, Harvey A. "How Designers Can Avoid Construction Claims." <u>Journal of Professional Issues in Engineering</u>. Vol. 111, No. 3. Proceedings of the American Society of Civil Engineers. July 1985, p. 100.

- [Moore, Walter P.] "Structural Safety: The Profession at a Crossroad." <u>Civil Engineering</u>. July 1985, p. 7. Excerpted from a keynote address to the 1984 annual meeting of the American Concrete-Institute.
- Nigro, William T. "Redicheck: A System of Interdisciplinary Coordination." <u>DPIC Communique</u>. Design Professionals Insurance Company. April 1984, page 1.
- Scott, Donald F. "Effective Contract Administration in Construction Management." Journal of the Construction Division. Vol. 100, No. C02. Proceedings of the American Society of Civil Engineers. June 1974, p. 117.
- "Design Changes: The Largest Cause of Overruns." Engineer News Record. 6 March 1975, p. 10.

MASTER'S THESES:

- Finley, Lloyd S. "Examination of the Constructibility Review in Government Contracting." Master's Thesis. School of Engineering, Purdue University. 3 August 1984.
- Rosmond, James R. "Analysis of Low Bidding and Change Order Rates for Navy Facilities Construction Contracts." Master's Thesis. Naval Postgraduate School, Monterey, California. June 1984.
- Rowland, Henry J. "The Causes and Effects of Change Orders on the Construction Process." Master's Thesis. School of Civil Engineering, Georgia Institute of Technology. November 1981.
- Turowski, Henry John. "Contractor Quality Control." Master's Thesis. Naval Postgraduate School, Monterey, California. December 1980.

OTHER SOURCES:

South Atlantic Division, Corps of Engineers. "Design Quality Forum." Forum Minutes. Atlanta, Beorgia. 3 September 1985. INITIAL DISTRIBUTION LIST

-

INITIAL DISTRIBUTION LIST

NOTE: One copy provided each addressee, unless otherwise indicated.

- Combined Arms Research Library U.S. Army Command and General Staff College Fort Leavenworth, Kansas 66027
- Construction Industry Institute ATTN: Dr. Richard Tucker ECJ 5.2 University of Texas, Austin Austin, TX 78712~1076

F

.

- 3. Defense Technical Information Center Cameron Station Alexandria, Virginia 22314
- Director of Engineering and Housing ATTN: LTC Robert Lowry Fort Leavenworth, Kansas 66027
- 5. LTC Willfred L. Dellva (2 Copies) Iowa State University 300 Carver Hall Ames, Iowa 50011
- MAJ Clayton H. Dethlefsen R.D. #4, Box 152-A Fort Riley Road Leavenworth, Kansas 66048
- MAJ Stanley J. Murphy DSRO U.S. Army Command and General Staff College Fort Leavenworth, Kansas 66027
- Office of the Chief of Engineers (5 Copies) ATTN: MG Mark J. Sisinyak (DAEN-ECZ) 20 Mass. Avenue, NW Washington, D.C. 20314-1000
- 9. U.S. Army Construction Engineering Research Laboratory P.O. Box 4005 Champaign, Illinois 61820-1305

 U.S. Army Engineer District, Alaska ATTN: District Engineer F.D. Box 898 Anchorage, Alaska 99506-0898

- 11. U.S. Army Engineer District, Baltimore ATTN: District Engineer P.O. Box 1715 Baltimore, MD 21203-1717
- 12. U.S. Army Engineer District, Far East ATTN: District Engineer APO San Francisco, CA 96301-0427
- 13. U.S. Army Engineer District, Fort Worth ATTN: District Engineer P.O. Box 17300 Fort Worth, TX 76102-0300
- 14. U.S. Army Engineer District, Japan ATTN: District Engineer APO San Francisco, CA 96301-0427
- 15. U.S. Army Engineer District, Kansas City (3 copies) ATTN: District Engineer 700 Federal Building Kansas City, MD 64106-2896
- 16. U.S. Army Engineer District, Los Angeles ATTN: District Engineer P.O. Box 2711 Los Angeles, CA 90053-2325
- 17. U.S. Army Engineer District, Louisville ATTN: District Engineer P.O. Box 59 Louisville, KY 40201-0059
- 18. U.S. Army Engineer District, Mobile ATTN: District Engineer P.O. Box 2288 Mobile, AL 36628-0001
- U.S. Army Engineer District, New York ATTN: District Engineer
 Federal Plaza
 New York, NY 10278-0090

- 20. U.S. Army Engineer District, Norfolk ATTN: District Engineer 803 Front Street Norfolk, VA 23510-1096
- 21. U.S. Army Engineer District, Omaha ATTN: District Engineer Rm. 6014 USPD and Courthouse Omaha, NE 63102-4978

- 22. U.S. Army Engineer District, Philadelphia (3 copies) ATTN: District Engineer U.S. Customs House 2nd and Chestnut Streets Philadelphia, PA 19106-2991
- 23. U.S. Army Engineer District, Sacramento ATTN: District Engineer 670 Capital Mall Sacramento, CA 95814-4794
- 24. U.S. Army Engineer District, Savannah ATTN: District Engineer P.D. Box B89 Savannah, 6A 31401-2708
- 25. U.S. Army Engineer District, Seattle ATTN: District Engineer P.O. Box C-3755 Seattle, WA 98124-2255
- 26. U.S. Army Engineer District, Tulsa ATTN: District Engineer P.O. Box 61 Tulsa, Ok 74121-0061
- 27. U.S. Army Engineer Division, Europe ATTN: Division Engineer APO New York 09757
- 28. U.S. Army Engineer Division, Huntsville ATTN: Division Engineer P.D. Box 1600 Huntsville, AL 35807-4301
- 29. U.S. Army Engineer Division, Missouri River ATTN: Division Engineer P.O. Box 103 Downtown Station Dmaha, NE 68101-0103

- 30. U.S. Army Engineer Division, North Atlantic ATTN: Division Engineer 90 Church Street New York, NY 10007-9998
- 31. U.S. Army Engineer Division, North Pacific ATTN: Division Engineer P.O. Box 2870 Fortland, OR 97208-2870
- 32. U.S. Army Engineer Division, Ohio River ATTN: Division Engineer P.O. Box 1159 Cincinnati, OH 45201-1159
- 33. U.S. Army Engineer Division, Pacific Ocean ATTN: Division Engineer Fort Schafter, HI 96858-5440
- 34. U.S. Army Engineer Division, South Atlantic ATTN: Division Engineer 510 Title Building 30 Pryor Street, S.W. Atlanta, GA 30335-6801
- 35. U.S. Army Engineer Division, South Pacific ATTN: Division Engineer 630 Sansome Street, Rm 720 San Francisco, CA 94111-2206
- 36. U.S. Army Engineer Division, Southwestern ATTN: Division Engineer 1114 Commerce St. Dallas, TX 75242-0216

